

The October board meeting location has changed to

Castro Valley USD, Boardroom 4400 Alma Avenue Castro Valley, CA 94546



GOVERNING BOARD MEETING AGENDA Thursday, October 6, 2016 5:30 pm

Location: 4400 Alma Avenue Castro Valley, CA 94546 Website: www.edenrop.org



Phone Numbers: (510) 293-2971 Fax (510) 293-8225

Governing Board Members

Gary Howard, President Lisa Brunner, Vice-President Janet Zamudio, Member Vince J. Rosato, Member

Castro Valley Unified School District Hayward Unified School District San Lorenzo Unified School District San Leandro Unified School District

Mission Statement

The mission of the Eden Area Regional Occupational Program is to prepare students for careers and further education as well as to instill workplace skills and ethics that enable them to compete successfully in the economy of today and the future.

Regular Meeting of the ROP Governing Board Castro Valley Unified School District, Board Room 4400 Alma Avenue., Castro Valley, CA 94546

Date: Thursday, October 6, 2016 Time: 5:30 p.m.

AGENDA

Welcome to the Eden Area Regional Occupational Program Governing Board Meeting. The purpose of the meeting is to consider matters of policy and business necessary for the operation of the Regional Occupational Program.

Visitors wishing to address the Governing Board are asked to complete a "Request to Address ROP Governing Board" card, available at the entrance to the meeting room, and submit it to the Governing Board's Administrative Secretary. Any member of the audience may speak on any agenda item by following this process, or upon recognition by the Chairperson by identifying him/herself and his/her organization affiliation prior to any action taken by the Governing Board. Such presentations may be limited. If there is a desire to address the Governing Board on a matter relating to the Eden Area ROP that does not appear on the agenda, this may be done during the "Other Business" section.

State law prohibits the ROP Governing Board from taking any action on or discussing items that are not on the posted agenda except to A) briefly respond to statements made or questions posed by the public in attendance; B) ask questions for clarification; C) provide a reference to a staff member or other resource for factual information in response to the inquiry; or D) ask a staff member to report back on the matter at the next meeting and/or place it on a future agenda. (Government Code Section 54954.2 (a))

This meeting is being recorded. These recordings are maintained by the Eden Area ROP for 30 days and are available for review to the public upon request.

- I. Call to Order
- II. Roll Call
- III. Pledge of Allegiance
- **IV. Mission Statement**
- V. Approval of Agenda

VI. CONSENT CALENDAR

Action by the ROP Governing Board of the Eden Area Regional Occupational Program means that all items listed under the Consent Calendar are adopted by one single motion, unless a member of the Governing Board, the Superintendent, or a member of the public requests that any such item be removed from the Consent Calendar and voted upon separately.

- A. Request the Governing Board to approve the Minutes of the Regular Governing Board Meeting of September 1, 2016 (pages 4-9)
- B. Request the Governing Board to approve the Bill Warrants (pages 10-15)
- C. Request the Governing Board to approve the Personnel Action Items (pages 16-17)
- D. Request the Governing Board to approve the Listed Donations-Vicente Gonzalez (page 18)
- E. Request the Governing Board to approve the Quarterly Report on Williams Act Complaints and Resolutions (pages 19-20)
- F. Request the Governing Board to approve the Revised Donation, Assignment and Assumption Agreement with Construction Craft Training Center (CCTC) (pages 21-92)
- G. Request the Governing Board to approve the Lease Agreement with L&M Investments for the Electrical Trainee Program-Turlock (pages 93-99)
- H. Request the Governing Board to approve the Sublease Agreement with American Home Inspectors Training (AHIT) (pages 100-102)

VII. INFORMATION ITEMS

- A. CDE Course Review-Photography (pages 103-104)
- B. Back to School Night (page 105)
- C. Principals' Breakfast (page 106)

VIII. ACTION ITEMS

A. Request the Governing Board to approve the MOU with Leadership Public Schools (pages 107-109)

Superintendent's Report

IX. Other Business/ Governing Board Reports

- A. Public
- B. ROP Governing Board

X. Recess to Closed Session

A. Public Employee Appointment/Employment (Pursuant Government Code section 54957): Superintendent's Contract

Page 3 – Agenda for the October 6, 2016 Regular Meeting of the ROP Governing Board

XI. Reconvene to Open Session and Report Action Taken in Closed Session

A. Public Employee Appointment/Employment (Pursuant Government Code section 54957): Superintendent's Contract

XII. Adjournment

Consent Calendar





Minutes of the Regular Meeting of the ROP Governing Board September 1, 2016

I. Call to Order

Trustee Gary Howard, President, called the meeting to order at 5:34 p.m. on Thursday, September 1, 2016 at the Eden Area Regional Occupational Program Board Room, 26316 Hesperian Blvd., Hayward, CA 94545.

II. Roll Call

Roll was called by Gabriela Juarez, Administrative Assistant.

Eden Area ROP Governing Board Present:			
Gary Howard, President Castro Valley USD			
Lisa Brunner, Vice-President Hayward USD			
Janet Zamudio, Member	San Lorenzo USD	arrived at 5:40 pm	
Evelyn Gonzalez, Alternate	San Leandro USD		
Evelyn Gonzalez, Alternate	San Leanuro USD		

Eden Area ROP Governing Board Absent: Vince Rosato, Member San Leandro USD

Superintendent: Linda Granger- present

ROP Administrators in Attendance:	
Craig Lang	Director
Sheila Lawrence	Assistant Director of Off-Site Programs
Marites Fermin	Business Manager
Evan Goldberg	Grant Coordinator
<u>ROP Staff in Attendance:</u> Erika Emery Gabriela Juarez	Careers in Education Program Instructor Superintendent's Administrative Assistant
Others in Attendance:	

Construction Craft Training Center (CCTC) Director

III. Pledge of Allegiance

Chris Floethe

Linda Granger led the Pledge of Allegiance.

IV. Mission Statement

Craig Lang read the Eden Area ROP Mission Statement.

V. Approval of Agenda

Trustee Evelyn Gonzalez moved to approve the agenda and Trustee Lisa Brunner, seconded the motion. By the following vote, the agenda was approved.

AYES:3 (Brunner, Gonzalez, Howard)NOES:0ABSTAIN:0ABSENT:2 (Rosato, Zamudio)

VI. Consent Calendar

Trustee Evelyn Gonzalez moved to approve the Consent Calendar with the exception of item E and I that was pulled for discussion and moved to be approved as separate items under consent:

- A. Approve the Minutes of the Regular Governing Board Meeting of August 4, 2016
- B. Approve the Bill Warrants Items
- C. Approve the Personnel Action Items
- D. Approve the Listed Donations-Jose A. Ortiz
- E. Approve the SkillsUSA Overnight Field Trips for the 2016-2017 Fiscal Year
- F. Approve the Public Disclosure of Eden Area ROP Employees' Agreement for the 2015-2016 Salary Increase
- G. Approve the MOU Addendum with Associated Builders and Contractors (ABC) for the 2016-2017 School Year
- H. Approve the Agreement with Alameda County Office of Education (ACOE) for Delivery Services for the 2016-2017 Fiscal Year
- I. Approve the Contract with Blackboard for Website Hosting for the 2016-2017 Fiscal Year
- J. Approve the Contract with Century Carpet, Inc. for Carpet Flooring Services for the 2016-2017 Fiscal Year
- K. Approve the Contract with Zamora & Sons for Painting Services for the 2016-2017 Fiscal year
- L. Approve the Agreement with Rhea Settles for Fall 2016
- M. Approve the contract with AT&T for Telephone Services
- N. Approve the Advisory Committee Meeting Minutes

Trustee Lisa Brunner seconded the motion.

AYES:3 (Brunner, Gonzalez, Howard)NOES:0ABSTAIN:0ABSENT:2 (Rosato, Zamudio)

E. Request the Governing Board to approve the SkillsUSA Overnight Field Trips for the 2016-2017 Fiscal Year

Trustee Lisa Brunner requested to pull the SkillsUSA Overnight Field Trips. Trustee Lisa Brunner just wanted other funding sources to be noted so that every student who qualifies can attend the competitions. Superintendent Granger highlighted that the ROP does do fundraisers throughout the year to raise money for the trips.

Upon review of and a motion by Trustee Evelyn Gonzalez and a second by Trustee Lisa Brunner, the Governing Board approved the SkillsUSA Overnight Field Trips.

Page 3 - Minutes of the Regular ROP Governing Board Meeting September 1, 2016

AYES:3 (Brunner, Gonzalez, Howard)NOES:0ABSTAIN:0ABSENT:2 (Rosato, Zamudio)

I. Request the Governing Board to approve the Contract with Blackboard for Website Hosting for the 2016-2017 Fiscal Year

Trustee Evelyn Gonzalez requested to pull the contract with Blackboard for website hosting. She had a question regarding the use of Blackboard. She wanted to know if blackboard was used for other services such as teleparent. Superintendent Granger shared that at this time it is only used for web hosting until we upgrade our phone system.

Upon review of and a motion by Trustee Evelyn Gonzalez and a second by Trustee Lisa Brunner, the Governing Board approved the Contract with Blackboard for website hosting for the 2016-2017 fiscal year.

AYES:	3 (Brunner, Gonzalez, Howard)
NOES:	0
ABSTAIN:	0
ABSENT:	2 (Rosato, Zamudio)

VII. Information Items

A. CDE Course Review-Careers in Education

Dr. Erika Emery, Instructor, shared that the Careers in Education program has returned as a full time at the ROP center. Students enrolled in this class qualify for a child development permit, meet the requirements for the U.C. g credit, earn community college credits through articulation agreements and job placement assistance. Dr. Emery also places students in internships throughout the community to build their skills. Dr. Emery concluded her report by sharing photos of the students in action.

B. Opening of School for the 2016-2017 School Year

Mr. Craig Lang, Director, reported on the opening of school. He shared that it had been a staggered start this year as the districts had multiple start dates. Students were here for the first day of ROP, regardless of their first day of school at the home high school. Strong collaboration has been established between school sites and the ROP to make sure students not only enroll in ROP, but into their desired class. Mr. Lang shared that our Career Counselor, Mrs. Audra Muñoz, has implemented using google docs for our enrollment. The google doc is a live document shared with the counselors at the high school sites. Using google docs allowed staff to fill classes faster and efficiently.

C. Dates for Sophomore Tours and the Student Award Ceremony

Sheila Lawrence, Assistant Director of Off-Site Programs, presented to the Governing Board the projected dates for the sophomore tours. She also shared the date of the Student Award Ceremony, it will be held on Wednesday, May 10, 2017 at 6:00 pm at the Castro Valley Center for the Arts. Ms. Lawrence then extended an invitation to the Governing Board to attend the

award ceremony. She also welcomed the Board members to attend their respective high schools sophomore tours dates.

D. Eden Area ROP Service Awards

Linda Granger, Superintendent, highlighted the employees that were recognized and honored for their commitment and service to the ROP. She shared that annually service awards are presented to staff for their years of experience and dedication to the organization. These awards were presented to staff at the August 15, 2016 Staff Development Day. Service awards were given to Heather Bystrom (10 years), Annette Senter (10 years), Jennifer Aguilar (15 years) and Francisco De Leon (20 years).

VIII. Action Items

A. Request the Governing Board to approve the 2015-2016 Unaudited Actual Income and Expenditure Report

Upon review of and a motion by Trustee Janet Zamudio and a second by Trustee Evelyn Gonzalez, the Governing Board approved the 2015-2016 unaudited actual income and expenditure report.

AYES:4 (Brunner, Gonzalez, Howard, Zamudio)NOES:0ABSTAIN:0ABSENT:1 (Rosato)

B. Request the Governing Board to approve the Revised Calendar of Governing Board Meetings for the 2016-2017 School Year

The Eden Area ROP 2016-2017 Governing Board Meeting Calendar was approved on June 2, 2016 at the Governing Board Meeting. After the June meeting it was announced that ACOE will be holding their Annual Teacher of the Year Awards Night on October 6, 2016 at 6:30 pm at the Castro Valley Center for the Arts at Castro Valley High School. This will conflict with the October meeting. ROP staff suggested changing the time of the meeting to start at 5:00 pm or changing the location to Castro Valley USD board meeting room located at 4400 Alma Avenue, Castro Valley. After some discussion the Board decided to move the location to CVUSD.

Upon review of and a motion by Trustee Lisa Brunner and a second by Trustee Janet Zamudio, the Governing Board approved the Revised Calendar of Governing Board Meetings for the 2016-2017 school year with a location change to the October meeting.

AYES:	4 (Brunner, Gonzalez, Howard, Zamudio)
NOES:	0
ABSTAIN:	0
ABSENT:	1 (Rosato)

C. Request the Governing Board to approve the Donation, Assignment and Assumption Agreement with Construction Craft Training Center

Page 5 - Minutes of the Regular ROP Governing Board Meeting September 1, 2016

Upon review of and a motion by Trustee Lisa Brunner and a second by Trustee Janet Zamudio, the Governing Board approved the Donation, Assignment and Assumption Agreement with Construction Craft Training Center.

AYES:	4 (Brunner, Gonzalez, Howard, Zamudio)
NOES:	0
ABSTAIN:	0
ABSENT:	1 (Rosato)

D. Request the Governing Board to approve the Contract for Consulting with Chris Floethe for the Electrical Trainee Program

Upon review of and a motion by Trustee Evelyn Gonzalez and a second by Trustee Lisa Brunner, the Governing Board approved the Contract for Consulting with Chris Floethe for the Electrical Trainee Program.

AYES:	4 (Brunner, Gonzalez, Howard, Zamudio)
NOES:	0
ABSTAIN:	0
ABSENT:	1 (Rosato)

E. Request the Governing Board to approve the Adult Program Salary Schedule

Upon review of and a motion by Trustee Janet Zamudio and a second by Trustee Lisa Brunner, the Governing Board approved the Adult Program Salary Schedule.

AYES:4 (Brunner, Gonzalez, Howard, Zamudio)NOES:0ABSTAIN:0ABSENT:1 (Rosato)

IX. Superintendent's Report

Linda Granger highlighted in her report the ROP's staff development day at the start of the year. The staff development day started with a guest speaker to discuss leading from where you are and that we are all leaders in our organization. This was a perfect message for our staff as there has been a lot of change in the last three years. The guest speaker also discussed changes for this generation of students. Studies show that this generation of students don't finish high school with work experience as they did before. He talked about what teachers can do to teach work ethics in their classroom because it's not happening outside the classroom. In keeping with the theme of change Superintendent Granger asked staff to share their input about the changes that have occurred over the last three years. She will use the data in her annual report.

Superintendent Granger also shared that ROP staff has signed up to several committees that will help with communication and progress of the ROP.

The Superintendent also highlighted that she started a weekly bulletin to the staff and that it was well received.

X. Other Business/Governing Board Reports

A. Public

None

B. Governing Board Reports

Janet Zamudio, San Lorenzo USD representative, shared that she brought back to the San Lorenzo Board the alignment of district calendars. The district plans to align with the first district to approve their calendar.

IV. Recess to Closed Session

The meeting was called into closed session at 6:28 pm.

A. Public Employee Performance Evaluation: Superintendent's Goals (Government Code Sec. 54957)

V. Reconvene to Open Session and Report any Action taken in Closed Session

The meeting resumed to open session at 6:35 p.m.

A. Public Employee Performance Evaluation: Superintendent's Goals (Government Code Sec. 54957)

Upon review of and a motion by Trustee Evelyn Gonzalezand a second by Trustee Janet Zamudio, the Governing Board approved the Superintendent's Goals for 2016-2017.

AYES:4 (Brunner, Gonzalez, Howard, Zamudio)NOES:0ABSTAIN:0ABSENT:1 (Rosato)

XI. Adjournment

The meeting was adjourned at 6:35 p.m.

Approved by the Eden Area ROP Governing Board ______.

Linda Granger, Superintendent Clerk to the ROP Governing Board

DATE:	October 6, 2016
TO:	ROP Governing Board
FROM:	Linda Granger, Superintendent
PREPARED BY:	Sabrina Ubhoff, Accounting Technician
SUBJECT:	Request the Governing Board to approve the Bill Warrants

CURRENT SITUATION

The bill warrants submitted for approval are for the period of August 17, 2016 through September 19, 2016 and include test warrant numbers and voided warrants.

DATE:	October 6, 2016
TO:	ROP Governing Board
FROM:	Linda Granger, Superintendent
PREPARED BY:	Mercedes Henderson, Personnel Coordinator
SUBJECT:	Request the Governing Board to approve the Personnel
	Action Items

CURRENT INFORMATION

The attached listing of personnel consent items are the Eden Area ROP Superintendent's recommendations for approval.

DATE:	October 6, 2016
TO:	ROP Governing Board
FROM:	Linda Granger, Superintendent
PREPARED BY:	Craig Lang, Director
SUBJECT:	Request the ROP Governing Board to approve the Listed
	Donations-Vincente Gonzalez

CURRENT SITUATION

Occasionally, gifts or monetary items are donated to the ROP. Vincente Gonzalez donated a 1997 Nissan 200SX to the Eden Area ROP Auto Technology program.

A letter of acceptance will be sent to all donors.

FISCAL IMPACT SOURCE

FUNDING

Not applicable

Not applicable

DATE:	October 6, 2016
TO:	ROP Governing Board
FROM:	Linda Granger, Superintendent
SUBJECT:	Request the Governing Board to approve the Quarterly
	Report on Williams Act Complaints and Resolutions

BACKGROUND

Education Code 35186 (d) requires the following:

A school district shall report summarized data on the nature and resolution of all complaints concerning deficiencies related to instructional materials, emergency or urgent facilities conditions and teacher vacancy or misassignment on a quarterly basis to the county superintendent of schools and the Governing Board of the school district. The summaries shall be publicly reported at a regularly scheduled meeting of the Governing Board of the school district. The report shall include the number of complaints with the number of resolved and unresolved complaints. The complaints and written responses shall be available as public records.

CURRENT SITUATION

Attached is a report for the complaints and resolutions through October 1, 2016 as specified by Education Code 35186 (d).



Quarterly Report on Williams Act Complaints

[Education Code 35186 (d)]

Report through: October 1, 2016

District: Person completing this form: Title: Eden Area Regional Occupational Program Gabriela Juarez Superintendent's Administrative Assistant

Quarterly Report Submission (check one)→ Date: October 1, 2016

	January 2016
	April 2016
	July 2016
\square	October 2016

Date for information to be reported publicly at the Governing Board meeting: October 6, 2016

Please check the box that applies:

No complaints were filed with any school in the district during the quarter indicated above.

Complaints were filed with schools in the district during the quarter indicated above. The following chart summarizes the nature and resolution of these complaints.

General Subject Area	Number of Complaints	Number of Resolved Complaints	Number of Unresolved Complaints
Textbooks and Instructional Materials	0	0	0
Teacher Vacancy or Mis-assignment	0	0	0
Facilities Conditions	0	0	0
CAHSEE Intensive Instruction and Services	0	0	0
TOTALS	0	0	0

Publicly reported at the Governing Board meeting on: October 6, 2016

DATE:	October 6, 2016
TO:	ROP Governing Board
FROM:	Linda Granger, Superintendent
SUBJECT:	Request the Governing Board to approve the Revised
	Donation, Assignment and Assumption Agreement with
	Construction Craft Training Center (CCTC)

BACKGROUND

The Eden Area ROP and Construction Craft Training Center (CCTC) have had a long partnership. When CCTC applied for approval to provide training services, we applied jointly with them and were assigned school #109 by the Department of Industrial Relations. CCTC has been providing this training for adults for 30 years.

CURRENT SITUATION

Construction Craft Training Center (CCTC) has approached the Eden Area ROP regarding assuming the operation of their adult electrical training program. In exchange for taking over the operation of this program, CCTC will be donating all of their existing furniture, equipment and materials owned by CCTC. The Donation, Assignment and Assumption Agreement provides the specifics of this transfer of property to the Eden Area ROP in exchange for the continued operation of the electrical trainee program.

Given the amount of time that it has taken for CCTC employees to complete the hiring process required by the State, we need to change the effective date on this agreement from 9/5/2016 to 10/1/2016. For this reason, we are bringing the agreement back to the Governing Board for approval with the effective date changed accordingly.

Fiscal Impact: funds generated from the adult programs will be used to cover the ongoing cost of providing the programs.

DONATION, ASSIGNMENT, AND ASSUMPTION AGREEMENT

This DONATION, ASSIGNMENT, AND ASSUMPTION AGREEMENT (the "<u>Agreement</u>") is made effective as of October 1, 2016 by and between Eden Area Regional Occupational Program, a California joint venture of Castro Valley, Hayward, San Leandro, and San Lorenzo Unified School Districts ("<u>Eden Area ROP</u>"), and Construction Craft Training Center, a California nonprofit public benefit corporation ("<u>CCTC</u>"), with reference to the following facts:

- A. Eden Area ROP is a governmental instrumentality and a qualified charitable organization under Section 170(c)(1) of the Internal Revenue Code of 1986, as amended (the "<u>Code</u>").
- B. CCTC is a qualified charitable organization under Section 501(c)(3) of the Code and a public charity under Section 509(a)(1) of the Code.
- C. CCTC desires to transfer the operation of its existing Electrician Training Program to Eden Area ROP by making a contribution to Eden Area ROP of all of its assets, and Eden Area ROP desires to accept the transfer of the operation of the Electrician Training Program and agrees to assume certain CCTC's liabilities as listed in Appendix VI, resulting in the net donation described in this Agreement.

NOW, THEREFORE, in consideration of the foregoing and mutual promises set forth below, the parties hereby agree as follows:

- 1. <u>Contribution</u>.
 - (a) <u>Assets</u>.

CCTC hereby irrevocably gives, delivers, transfers, and assigns to Eden Area ROP all right, title, and interest in and to the assets of CCTC more specifically described on in <u>Appendix I</u> (the "<u>Assets</u>").

- (b) <u>Leases</u>.
 - (1) <u>26200 Hayward Lease.</u>

CCTC hereby assigns all of its right, title, and interest in and to that certain Hayward Business Park Lease Agreement by and between Chris Floethe dba Construction Craft Training as tenant and Hayward Business Park, Inc. as landlord, for the premises located at 26200 Industrial Blvd., Hayward, CA 94545, a copy of which is attached to this Agreement as <u>Appendix II</u> (the "<u>26200</u> <u>Lease</u>") and the aggregate security deposit held by Hayward pursuant to the 26200 Lease.

(2) <u>26206 Hayward Lease.</u>

CCTC hereby assigns all of its right, title, and interest in and to that certain Hayward Business Park Lease Agreement by and between Chris Floethe dba Construction Craft Training as tenant and Hayward Business Park, Inc. as landlord, for the premises located at 26206 Industrial Blvd., Hayward, CA 94545, a copy of which is attached to this Agreement as <u>Appendix III</u> (the "<u>26206</u> <u>Lease</u>") and the aggregate security deposit held by Hayward pursuant to the 26206 Lease.

An assignment of lease from Chris Floethe dba Construction Craft Training to CCTC for the 26200 Lease and the 26206 Lease is attached to this Agreement as <u>Appendix IV</u> and incorporated herein by reference.

(3) <u>Turlock Lease.</u>

CCTC hereby assigns all of its right, title, and interest in and to that certain Multi-Tenant Lease – Gross by and between CCTC as tenant and L & M Investments as landlord, for the premises located at 2480 Acme Court, Turlock, CA 95380, a copy of which is attached to this Agreement as <u>Appendix V</u> (the "<u>Turlock Lease</u>") and the aggregate security deposit made by CCTC pursuant to the Turlock Lease.

2. <u>Assumption of Liabilities</u>.

Eden Area ROP hereby assumes only those CCTC liabilities as are set forth on <u>Appendix VI</u>, which include the liabilities under the 26200 Lease, the 26206 Lease, and the Turlock Lease. Other than the liabilities as are set forth on Appendix VI, Eden Area ROP does not assume any liabilities, known or unknown, of CCTC.

3. <u>Representations of CCTC and Chris Floethe.</u>

CCTC and Chris Floethe hereby represent and warrant to Eden Area ROP that:

- (a) CCTC has obtained all necessary consents to the transfer of the Assets, including the consents of Landlord as required under the 26200 Lease, the 26206 Lease, and the Turlock Lease.
- (b) The \$89,175 Long Term Note Payable as referenced on the August 30, 2015 CCTC Balance Sheet has been paid in full.
- (c) The \$10,800 Note Receivable as referenced on the August 30, 2015 CCTC Balance Sheet has been received and is no longer outstanding.
- (d) There are no liabilities of CCTC other than those listed on the most recent Balance Sheet.
- (e) The transfer of the Assets by CCTC will not violate any term of any agreement in place between CCTC and any third party.
- (f) CCTC owns the Assets free and clear from any lien or encumbrance, and has the power to transfer the Assets in accordance to the terms set forth in this Agreement.

4. <u>Indemnity.</u>

CCTC and Chris Floethe hereby, jointly and severally, agree to indemnify, defend and hold Eden Area ROP harmless for any damages incurred by Eden Area ROP based on CCTC and/or Chris Floethe's breach of this Agreement or any breach of the representations and warranties of CCTC and/or Chris Floethe as contained in this Agreement.

5. <u>Miscellaneous</u>.

(a) <u>Governing Law</u>.

This Agreement shall be governed by and construed according to the laws of the State of California that would apply if all parties were residents of California and this Agreement was made and performed in California.

(b) <u>Successors and Assigns</u>.

This Agreement and all its provisions shall be binding on and inure to the benefit of the successors and assigns of the parties.

(c) <u>Notices</u>.

Any notices given pursuant to this Agreement shall be in writing and shall be served either personally or delivered by United States mail, postage prepaid, registered or certified, return receipt requested, to the address set forth below for each party. Each party may change such address for notices by delivery of a written notice to the other party.

Eden Area ROP:	Eden Area Regional Occupational Program Attn: Linda Grainger, Superintendent 26316 Hesperian Blvd. Hayward, CA 94545
CCTC:	Construction Craft Training Center Attn: Chris Floethe, Executive Director 26200 Industrial Blvd. Hayward, CA 94545

(d) <u>Entire Agreement</u>.

This Agreement contains the entire understanding between the parties hereto with respect to the subject matter hereof, and supersedes all prior and contemporaneous agreements and understandings, inducements, or conditions, express or implied, oral or written, except as herein contained.

(e) <u>Amendment</u>.

This Agreement may not be modified or amended other than by an agreement in writing signed by an authorized representative of each party hereto.

IN WITNESS WHEREOF, the parties have entered into this Donation, Assignment, and Assumption Agreement effective as of the date set forth above.

Eden Area Regional Occupational Program

Construction Craft Training Center

By:

By: _____

Chris Floethe, Executive Director

Linda Granger, Superintendent

Chris Floethe

APPENDIX I TO DONATION AGREEMENT

Assets and Liabilities

1. All assets listed on attached Balance Sheet dated as of June 30, 2016, including but not limited to all accounts receivable of CCTC, all inventory, and all prepaid expenses.

Construction Craft Training Center Balance Sheet June 30,2016

	June 30,2016		
	Assets		
Current Assets: Cash Accounts Receivable Less: Reserve for Bad Debts Merchandise Inventory Prepaid Expenses Notes Receivable Total Current Assets	22580 \$51,875 2,594	\$22,580 49,281 4,300 11,512 0	\$87,673
<i>Fixed Assets:</i> Vehicles Less: Accumulated Depreciation	0 0	0	
Furniture and Fixtures Less: Accumulated Depreciation	22,500 2,250	20,250	
Equipment Less: Accumulated Depreciation	135,000 13,500	121,500	
Buildings Less: Accumulated Depreciation	0 0	0	
Land Total Fixed Assets		0	141,750
Other Assets: Goodwill Total Other Assets		0_	0
Total Assets			\$229,423

Liabilities and Capital

Current Liabilities:		
Accounts Payable	\$8,300	
2410 - Credit Card #8885	\$368	
2420- Credit Card #0339	\$11,401	
Payroll Liabilities	5,212	
Payroll wages	34,829	
Unearned Revenues	0	
Short-Term Notes Payable	0	
Short-Term Bank Loan Payable	0	
Total Current Liabilities		\$60,110

Construction Craft Training Center Balance Sheet June 30,2016

•	
0	
0	
	0
	60,110
0	
59,224	
	59,224
	\$119,334
	0 0

APPENDIX II TO DONATION AGREEMENT

26200 Hayward Lease

See attached.



May 28, 2015

Construction Craft Training 26200 Industrial Blvd Hayward, CA 94545

Dear Chris Floethe,

We have received your request to active the two year option term on your lease agreement for 26200 and 26206 Industrial Blvd. in Hayward, California. As further provided, the Option period shall be on the same terms and conditions as the lease agreement, with exception for the following terms:

The commencement date of the option begins June 1, 2015 and the option term ends May 31, 2017.

Per the option clause in the lease, Base Rent for 26200 Industrial Blvd will be increased by 2% in accordance with the Consumer Price Index clause to \$3055.00. It will be adjusted in accordance with the consumer price index on the commencement of the second year of the option term.

Base Rent for 26206 Industrial Blvd will be increased by 2% in accordance with the Consumer Price Index clause to \$2142.00. It will be adjusted in accordance with the consumer price index on the commencement of the second year of the option term.

Sincerely,

delsey Ressial

Kelsey Pressnall Administration Kelsey@friesproperties.com

Hayward Business Park Lease Agreement

Hayward Business Park Inc., herein called "Lessor" whose address is 26250 Industrial Boulevard, Hayward CA 94545 and Chris Floethe dba Construction Craft Training "Lessee" whose address is 26200 Industrial Blvd, Hayward, CA 94545, agree to the following terms and conditions relating to the lease of land and improvements within the Hayward Business Park.

1. GRANT AND TERM

1.1. LEASED PREMISES.

In consideration of the rents, covenants and agreements hereinafter reserved and contained on the part of Lessee to be observed and performed, the Lessor demises and leases to the Lessee and Lessee rents from Lessor, those certain premises, now in the Hayward Business Park (herein called the "Business Park"), commonly known as 26206 Industrial Bivd., in the City of Hayward, County of Alameda, California, premises consists of a store having approximately 2550 square feet herein called the "leased premises".

1.2. USE OF ADDITIONAL AREAS.

The use and occupation by the Lessee of the leased premises shall include the use in common with others entitled thereto of the common areas, employees' parking areas, service roads, loading facilities, sidewalks and customer car parking areas, and other facilities as may be designated from time to time by the Lessor, subject however to the terms and conditions of this agreement and to reasonable rules and regulations for the use thereof as prescribed from time to time by Lessor.

1.3. COMMENCEMENT OF TERM.

The term of this lease, and Lessee's obligation to pay rent, shall commence on June 1, 2014. In the event that the fixed date above does not occur on the first day of the month, Lessee shall pay rent for the fractional month on a per diem basis (calculated on the basis of a thirty day month) until the first day of the month when the term hereunder commences; and thereafter the minimum rent shall be paid in equal monthly installments on the first day of each and every month in advance.

1.4. ACCEPTANCE

Lessee shall, by entering into and occupying the demised premises be conclusively deemed to have accepted the same and to have acknowledged that said demised premises are in good order, condition and repair.

1.5. LENGTH OF TERM.

The term of this lease shall be for one (1) year and commencement of the term as provided in Section 1.3 hereof and ease shall expire on May 31, 2015.

1.6. EXCUSE OF LESSOR'S PERFORMANCE.

Anything in this agreement to the contrary notwithstanding, providing such cause is not due to the willful act or neglect of the Lessor, the Lessor shall not be deemed in default with respect to the performance of any of the terms, covenants and conditions of this lease if same shall be due to any strike, lockout, civil commotion, war-like operation, invasion, rebellion, hostilities, military or usurped power, sabotage, governmental regulations or controls, inability to obtain any material, service or financing rain or muddy conditions, through Act of God or other cause beyond the control of the Lessor.

2. RENT

2.1. MINIMUM RENT.

The fixed minimum rent during the term of this lease shall be payable by Lessee in monthly installments, or on or before the first day of each month in advance, at the office of the Lessor or at such other place designated by Lessor, without any prior demand therefore, and without any deduction or set-off whatsoever, and shall be as follows: Base Rent per month, is two thousand, one hundred (\$2100.00) dollars.

Initialed

1

- In the event the fixed minimum monthly rent is not paid to Lessor by Lessee within 10 (ten) days of the date on which it is due Lessee agrees to pay to Lessor as additional rent, a late-charge of 10% of rental amount due. Lessee further agrees to pay Lessor any costs incurred by Lessor in effecting the collection of such past due rent and late-charge including but not limited to reasonable fees of an attorney or collection agency. The parties hereto agree that all late charges shall be deemed additional rent and shall be due and payable as such within five (5) days of Lessee's receipt of a statement itemizing the same. Nothing herein contained shall limit any other remedy of Lessor.
- 2.2. REAL ESTATE TAXES

Lessee agrees to pay Lessee's pro rata share of all real property taxes and assessments which may be levied or assessed by any lawful authority against the land on which buildings are located and improvements thereon in the Business Park. Lessee shall pay said taxes monthly upon receipt from Lessor of a statement delineating Lessee share of said taxes and said share shall be paid within five (5) days after receipt of said statement. Lessee's pro rata share shall be apportioned according to the floor area of the demised premises as it relates to the total floor area of the building or buildings including the demised premises. All taxes for the year in which this lease commences shall be apportioned and adjusted.

As used herein, the term "real property tax" shall include any form of real estate tax or assessment, general, special, ordinary or extra ordinary, and license fee, commercial rental tax, improvement bond or bonds, levy or tax [other than inheritance, personal income or estate taxes] imposed upon the premises by any authority having the direct or indirect powers of tax, including any city, state or federal government, or any school, sanitary, fire, street, drainage or other improvement district thereof as against any legal or equitable interest of Lessor in the premises or in the real property of which the premises are a part, as against Lessor's business of leasing the premises. The term " real property tax" shall also include any tax, fee, levy, assessment or charge [1] in substitution of partially or totally, any tax, fee, levy, assessment or charge hereinabove included within the definition of real property tax or [2] the nature of which was hereinbefore included within the definition of real property tax or [3] which is imposed for a service or right not charged prior to June 1, 1978, or if previously charged has been increased since June 1, 1978.

2.3. ADDITIONAL RENT.

The Lessee shall pay as additional rent money required to be paid pursuant to sections 2.2, 8.1, 10.1, 11.2,11.4, and 12.1, and all other sums of money or charges required to be paid by Lessee under this lease, without any deductions or setoffs whatsoever, whether or not the same shall be designated "additional rent." If such amounts or charges are not paid at the time provided in this lease, the Lessee further agrees to pay to Lessor any cost incurred by Lessor in effectuating the collection of such past due amounts or charges including but not limited to fees of an attorney or a collection agency. The parties hereto agree that all of the hereinabove mentioned charges shall be deemed additional rent and shall be all due and payable as such within five (5) days of Lessee's receipt of a statement itemizing the same. Nothing herein contained shall limit any other remedy of Lessor.

- 2.4. ADJUSTMENTS TO MINIMUM RENT.
 - 2.4.1. Commencing with the end of the first lease year, the minimum rent may be adjusted and after the end of each subsequent lease year during the term of this lease in accordance with the Consumer Price Index, hereinafter called "CPI" as published by the Bureau of Labor Statistics (U.S. Cities Average for all Urban Consumers) between (i) the amount of that Index as last published prior to the beginning of the preceding 12-month period, and (ii) the amount of that Index as last published prior to the date on which the rental increase is to occur
 - 2.4.2. The CPI as of the first month shall be designated as the Beginning Index;
 - 2.4.3. Promptly after the end of the first Lease Year and each year thereafter, the annual rental shall be adjusted as follows. If the CPI has increased over the beginning index, the minimum monthly rent payable during the next lease year shall be set my multiplying the minimum rent set forth in paragraph 4.1 by a fraction, the numerator of which is the CPI and the denominator is the beginning index. As soon as the minimum monthly rent for the next lease year is set, the Lessor shall give Lessee notice of the amount of the new minimum monthly rent. If the index is changed so that the base year differs from that used as of the date most immediately preceding the date the first lease year commences, the CPI shall be converted in accordance with conversion factors

Initialed (

published by the United States Department of Labor Statistics. If the index is discontinued or revised during the term, such other government index with which it is replaced shall be used in order to obtain substantially the same result as would be obtained if the index were not discontinued. In no event shall the annual increase be less than 2% or more than 4%.

3. CONSTRUCTION OF LEASED PREMISES

3.1. CHANGES AND ADDITIONS TO BUILDINGS.

Lessor hereby reserves the right at any time to make alterations or additions to and to build additional stories on the building in which the premises are contained and to build adjoining the same. Lessor also reserves the right to construct other buildings or improvements in the Business Park from time to time and to make alterations thereof or additions thereto and to build additional stories on any such building or buildings and to build adjoining same. Easements for light and air are not included in the leasing of these premises to Lessee. Lessor further reserves the exclusive right to the roof except as provided in this lease agreement.

3.2. RIGHT TO RELOCATE.

Lessor reserves the right at any time to vary and adjust the size of the various buildings, co-Lessees, automobile parking areas, and other common areas as described in this lease, provided, however, that said parking area (including landscaped and common areas) shall at all times be in compliance with the minimum requirements of the City of Hayward, County of Alameda, State of California.

4. CONDUCT OF BUSINESS BY LESSEE

4.1. USE OF PREMISES.

Lessee shall use the leased premises for the purpose of conducting the business of: Office & Classroom. Lessee shall occupy the leased premises within thirty (30) days after the date of the notice provided for in Section 1.3 hereof, and shall conduct continuously in the leased premises the business above stated. Lessee will not use or permit, or suffer the use of the leased premises for any other business or purpose without Lessor's approval.

4.2. CONDUCT OF BUSINESS.

Lessee acknowledges that Lessor makes no representations as to the present or future condition of the demised premises, or to the fitness, desirability or zoning hereof for any particular purpose, and Lessor shall not be liable for any charges therein or additions thereto required by any public authority. Any permits or requirements of any kind pertaining to the operation of Lessee's business will be Lessee's responsibility and any construction or special equipment required by public authority to enable Lessee to conduct his business shall likewise be Lessee's sole cost and responsibility.

4.3. ACCEPTANCE OF PREMISES

Acceptance of demised premises except as may be otherwise herein provided, Lessees shall be entering into and occupying the demised premises shall be deemed to have accepted the demised premises "as is" and to have acknowledged that the same are as represented by Lessor and in good order, condition and repair. Lessee acknowledges that it has reviewed and examined the demised premises prior to the execution hereof to its complete satisfaction.

4.4. OPERATION OF BUSINESS.

Lessee shall operate one hundred percent (100%) of the leased premises during the entire term of this lease with due diligence and efficiency unless prevented from doing so by causes beyond Lessee's control. Subject to inability by reason of strikes or labor disputes, Lessee shall carry at all times in said premises a stock of merchandise of such size, character and quality as shall be reasonably designed to produce the maximum return to Lessee. Lessee shall conduct its business in the leased premises during the regular customary days and hours for such type of business in the City or trade area in which the Business Park is located.

Initialed Initialed

4.5. STORAGE

Lessee shall warehouse, store and/or stock in the leased premises only such goods, wares and merchandise as Lessee intends to offer for sale at retail at, in, from or upon the leased premises. No auction, fire or bankruptcy sales may be conducted in the leased premises without the previous written consent of Lessor.

5. GRANT OF CONCESSIONS

5.1. CONDITIONS TO GRANT.

The provision against subletting elsewhere contained in this lease shall not prohibit Lessee from granting concessions for the operation of one or more departments of the business which Lessee is permitted by Section 4.1 to conduct in or upon the leased premises; provided, however that [a] each such concession may be granted only upon receipt by Lessee of the written consent of the Lessor and shall be subject to all terms and provisions of this lease; [b] at least seventy-five percent (75%) of the sales floor area of the leased premises shall be at all times devoted to the business of and be operated by Lessee.

6. SECURITY DEPOSIT

6.1. AMOUNT OF DEPOSIT.

Lessee, contemporaneously with the execution of this lease, will deposit with Lessor the sum of two thousand and forty (\$2040.00). Said deposit shall be held by Lessor, without liability for interest, as security for the faithful performance by Lessee of all of the terms, covenants, and conditions of this lease by said Lessee to be kept and performed during the term hereof. If at any time during the term of this lease any of the rent herein reserved shall be overdue and unpaid, or any other sum payable by Lessee to Lessor hereunder shall be overdue and unpaid then Lessor may, at the option of Lessor (but Lessor shall not be required to), appropriate and apply any portion of said deposit to the payment of any such overdue rent or other sum. It is expressly understood that this sum does not apply toward rent.

6.2. USE AND RETURN OF DEPOSIT.

In the event of the failure of Lessee to keep and perform any of the terms, covenants and conditions of this lease to be kept and performed by Lessee, then the Lessor at its option may, after terminating this lease, appropriate and apply said entire deposit, or so much thereof as may be necessary, to compensate the Lessor for all loss or damage sustained or suffered by Lessor due to such breach on the part of Lessee. Should the entire deposit, or any portion thereof, be appropriated and applied by Lessor for the payment of overdue rent or other sums due and payable to Lessor by Lessee hereunder, then Lessee shall, upon the written demand of Lessor, forthwith remit to Lessor a sufficient amount in cash to restore said security to the original sum deposited and Lessee's failure to do so within five (5) days after receipt of such demand shall constitute a breach of this lease. Should Lessee comply with all of said terms, covenants and conditions and promptly pay all of the rental herein provided for as it fails due, and all other sums payable by Lessee to Lessor hereunder, the said deposit shall be returned in full to Lessee within two weeks from the end of the term of this lease, or upon the earlier termination of this lease.

6.3. TRANSFER OF DEPOSIT.

Lessor may deliver the funds deposited hereunder by Lessee to the purchaser of Lessor's interest in the leased premises, in the event that such interest is sold, and thereupon Lessor shall be discharged from any further liability with respect to such deposit.

7. PARKING AND COMMON USE AREAS AND FACILITIES

. 7.1. CONTROL OF COMMON AREAS BY LESSOR.

All automobile parking areas, driveways, entrances and exits thereto, and other facilities furnished by Lessor In or near the Business Park, including employee parking areas, the truck way or ways, loading docks, package pick-up stations, pedestrian sidewalks and ramps, landscaped areas, exterior stalrways, first-aid stations, comfort stations and other areas and improvements provided by Lessor for the general use, in common, of Lessees, their officers, agents

Instinler Initialed

4

employees and customers, shall at all times be subject to the exclusive control and management of Lessor, and Lessor shall have the right from time to time to establish, modify and enforce reasonable rules and regulations with respect to all facilities and areas mentioned in this Article. Lessor shall have the right to construct, maintain and operate lighting facilities on all said areas and improvements, to police the same, from time to time to change the area, level, location and arrangement of parking areas and other facilities hereinabove referred to; to restrict parking by Lessees, their officers, agents and employees to employee parking areas; to enforce parking charges (by operation of meters or otherwise), with appropriate provisions for free parking ticket validating by Lessees; to close all or any portion of said areas or facilities to such extent as may, in the opinion of Lessor's council, be legally sufficient to prevent a dedication thereof or the accrual of any rights to any person or the public therein; to close temporarily all or any portion of the parking areas or facilities; to discourage non-customer parking; and to do and perform such other acts in and to said areas and improvements as, in the use of good business judgment, the Lessor shall determine to be advisable with a view to the improvement of the convenience and use thereof by Lessees, their officers, agents, employees and customers. Lessor will operate and maintain the common facilities referred to above in such a manner as Lessor, in its sole discretion, shall determine from time to time. Without limiting the scope of such discretion, Lessor shall have the full right and authority to employ all personnel and to make all rules and regulations pertaining to and necessary for the proper operation and maintenance of the common areas and facilities.

7.2. LICENSE.

All common areas and facilities not within the leased premises, which Lessee may be permitted to use and occupy, are to be used and occupied under a revocable license, and if any such license be revoked, or if the amount of such areas be diminished, Lessor shall not be subject to any liability nor shall Lessee be entitled to any compensation or diminution or abatement of rent, nor shall such revocation or diminution of such areas be deemed constructive or actual eviction.

8, COST OF MAINTENANCE OF COMMON AREAS [NNN]

- 8.1. LESSEE TO BEAR PRO RATA SHARE OF EXPENSE.
 - 8.1.1. In each lease year, as defined in Section 2.1 hereof, Lessee will pay to Lessor, in addition to the rentals specified in Article II hereof, as further additional rent, subject to the limitation hereinafter set forth, a proportion of the Business Park's operating cost, hereinafter defined, based upon the ratio of the square feet of all the building space in the Business Park.
 - 8.1.2. For the purpose of this Section 8.1 the "Business Park's operating cost" means the total cost and expense incurred in operating and maintaining the common facilities, hereinafter defined, actually used or available for use by Lessee and the employees, agents, servants, customers and other invitees of Lessee, specifically including without limitation, gardening and landscaping, the cost of public liability and property damage insurance, fire and extended coverage insurance, joint utilities, painting, and repairs (not capital in nature), property management, employment of security guards, bookkeeping, real estate property taxes and assessments thereon, and the cost of personnel to implement such services, to direct parking, and to police the common facilities. "Common Facilities" means all areas, space, equipment, and special services provided by Lessor for the common or joint use and benefit of the occupants of the Business Park, their employees, agents, servants, customers and other invitees, including without limitation parking areas, access roads, driveways, retaining walls, landscaped areas, truck service ways or tunnels, loading docks, pedestrian malls, courts, stairs, ramps and sidewalks, comfort and first aid stations, washrooms, parcel pick-up stations and Business Park sign rental.
 - 8.1.3. The additional rent to be paid in this Section 8.1 and also Sections 2.2, 11.1 and 11.2 shall be paid monthly in advance. The monthly charge shall be estimated for a twelve (12) month period and adjusted annually upwards or downwards to reflect the actual cost incurred. Lessee will then be credited or billed accordingly at year end to reflect this adjustment. Lessee shall have the right to review the breakdown on the operating costs that are incurred for the premises once a year, should Lessee request.

Initialed

8.1.4. Changes in any particular floor area occurring during any monthly period shall be effective on the first day of the next succeeding monthly period, and the amount of any floor area in effect for the whole of any monthly period shall be the average of the total amounts in effect on the first day of each calendar month in such monthly period.

9. SIGNS, AWNINGS, CANOPIES, FIXTURES, ALTERATIONS

9.1. INSTALLATION BY LESSEE.

All fixtures installed by Lessee shall be new or completely reconditioned. Lessee shall not make or cause to be made any alterations, additions or improvements or install or cause to be installed any trade fixtures, exterior signs, floor covering, interior or exterior lighting, plumbing fixtures, shades or awnings or make any changes to the store front without first obtaining Lessor's written approval and consent. Lessee shall present to the Lessor plans and specifications for such work at the time approval is sought.

9.2. REMOVAL AND INSURANCE BY LESSEE.

All alterations, decorations, additions and improvements made by the Lessee, or made by the Lessor on the Lessee's behalf by agreement under this lease, shall remain the property of the Lessee for the term of the lease, or any extension or renewal thereof. The Lessee shall at all times maintain fire insurance with extended coverage in the name of the Lessor and the Lessee, in an amount adequate to cover the cost of replacement of all alterations, decorations, additions or improvements in the event of fire or extended coverage loss. Lessee shall deliver to the Lessor certificates of such fire insurance policies which shall contain a clause requiring the insurer to give the Lessor ten (10) days notice of cancellation of such policies. Such alterations, decorations, additions and improvements shall remove all such alterations, decorations, additions and improvements, and restore the leased premises as provided in Section 10.3 hereof. If the Lessee fails to remove such alterations, decorations, additions and improvements and restore the leased premises, then upon the expiration of this lease, or any renewal thereof, and upon the Lesse's removal from the premises, all such alterations, decorations, additions and improvements and restore the leased premises, all such alterations, decorations, additions and improvements and restore the leased premises, then upon the expiration of this lease, or any renewal thereof, and upon the Lesse's removal from the premises, all such alterations, decorations, additions and improvements and restore the leased premises, all such alterations, decorations, additions and improvements and restore the leased premises, all such alterations, decorations, additions and improvements and restore the leased premises, all such alterations, decorations, additions and improvements and restore the leased premises, all such alterations, decorations, additions and improvements and restore the leased premises, all such alterations, decorations, additions and improvements and restore the leased premis

9.3. LESSEE SHALL DISCHARGE ALL LIENS.

Lessee shall promptly pay all contractors and material men, if such were employed by Lessee or Lessee's agents for work or modifications in the leased premises, so as to minimize the possibility of a lien attaching to the leased premises, and should any such lien be made or filed, Lessee shall bond against or discharge the same within ten (10) days after written request by Lessor.

9.4. SIGNS, AWNINGS AND CANOPIES.

Lessee will not place or suffer to be placed or maintained on any exterior door, wall or window of the leased premises any sign, awning or canopy, or advertising matter or other thing of any kind, and will not place or maintain any decoration, lettering or advertising matter on the glass of any window or door of the leased premises without first obtaining Lessor's written approval and consent. Lessee further agrees to maintain such sign, awning, canopy, decoration, lettering, advertising matter or other thing as may be approved in good condition and repair at all times. Lessee agrees, at Lessee's sole cost, to obtain a sign in strict conformance with Lessor's sign criteria as to design, material, color, location, size, and letter style.

10. MAINTENANCE OF LEASED PREMISES

- 10.1. MAINTENANCE BY LESSEE.
 - 10.1.1.Lessee shall at all times keep the leased premises (including maintenance of exterior entrances, all glass and show window moldings) and all partitions, doors, door jams, door closers, door hardware, fixtures, equipment and appurtenances thereof (including electrical, lighting, heating and plumbing fixtures, and any air conditioning system, including leaks around ducts, pipes, vents, or other parts of the air conditioning, heating or plumbing systems which protrude through the roof) in good order, condition and repair including

Initialed Initialed

replacements (including reasonably periodic painting as determined by Lessor), damage by unavoidable casualty excepted, except for structural portions of the premises, which shall be maintained by Lessor, but if Lessor is required to make repairs to structural portions by reason of Lessee's negligent acts or omission to act, Lessor may add the cost of such repairs to the rent which shall thereafter become due.

- 10.1.2.Lessee shall comply with all legal requirements now or hereafter affecting or applying to the premises at Lessee's sole cost and expense.
- 10.1.3. Lessee shall keep the premises and all other parts of the Business Park free from any and all liens arising out of any work performed, materials furnished or obligations incurred by Lessee or for Lessee by Lessee's Agents, and agrees to bond against, discharge any mechanic's or material men's lien, or post security with Lessor, satisfactory to Lessor, within ten (10) days after the filing of any such lien. In no event said bond and security shall be in an amount less than twice the amount of such lien. In the event Lessee fails to do any of the following within said ten (10) day period, Lessor may pay the underlying claim thereof, or otherwise discharge said lien, at Lessee's sole cost and expense. Lessee shall reimburse Lessor therefore, together with interest at the interest rate of ten percent (10%) within ten (10) days after Lessor's demand therefore.
- 10.1.4.Lessee expressly waives Sections 1941 and 1933(i) of the California Civil Code, and all other rights to make repairs at the expense of Lessor or deduct any amounts from rent as provided in any statute or law in effect during the term of this Lease in any.
- 10.2. MAINTENANCE BY LESSOR.

If Lessee refuses or neglects to repair properly as required hereunder and to the reasonable satisfaction of Lessor as soon as reasonably possible after written demand, Lessor may make such repairs without liability to Lessee for any loss or damage that may accrue to Lessee's merchandise, fixtures, or other property or to Lessee's business by reason thereof, and upon completion thereof, Lessee shall pay Lessor's reasonable costs for making such repairs, upon presentation of bill thereof, as additional rent. Said bill shall include interest at ten (10%) percent on said cost beginning fifteen (15) days after the date of completion and written notification of such completion of repairs by Lessor to the Lessee.

10.3. SURRENDER OF PREMISES.

At the expiration of the tenancy hereby created, Lessee shall surrender the leased premises in the same condition as the leased premises were in upon delivery of possession thereto under this lease, reasonable wear and tear excepted, and damage by unavoidable casualty excepted to the extent that the same is covered by Lessor's fire insurance policy with extended coverage endorsement, and shall surrender all keys for the leased premises to Lessor at the place then fixed for the payment of rent and shall inform Lessor of all combinations on locks, safes and vaults, if any, in the leased premises. Lessee shall remove all its trade fixtures, and any alterations or improvements as provided in Section 9.2 hereof, before surrendering the premises as aforesaid and shall survive the expiration or other termination of the term of this lease.

- 10.4. RULES AND REGULATIONS.
 - 10.4.1.The Lessee agrees as follows:
 - 10.4.1.1. All loading and unloading of goods shall be done only at such times, in the areas, and through the entrances, designated for such purposes by Lessor.
 - 10.4.1.2. The delivery or shipping of merchandise, supplies and fixtures to and from the leased premises shall be subject to such rules and regulations as in the judgment of the Lessor are necessary for the proper operation of the leased premises or Business Park.
 - 10.4.1.3. All garbage and refuse shall be kept in the kind of container specified by Lessor, and shall placed outside of the premises prepared for collection in the manner and at the times and places specified by Lessor. If Lessor shall provide or designate a service for picking up refuse and garbage, Lessee shall use same at Lessee's cost. Lessee shall pay the cost of removal of any of Lessee's refuse or rubbish.

Initialed CAP Initialed

- 10.4.1.4. No aerial shall be erected on the roof or exterior walls of the premises, or on the grounds, without in each instance, the written consent of the Lessor any aerial so installed without such written consent shall be subject to removal without notice at any time.
- 10.4.1.5. No loud speakers, televisions, phonographs, radios or other devices shall be used in a manner so as to be heard or seen outside of the premises without the prior written consent of the Lessor.
- 10.4.1.6. The outside areas immediately adjoining the premises shall be kept clean and free from dirt and rubbish by the Lessee to the satisfaction of the Lessor and Lessee shall not place or permit any obstructions or merchandise in such areas.
- 10.4.1.7. Lessee and Lessee's employees shall park their cars only in those portions of the parking area designated for that purpose by Lessor. In the event the Lessee or its employees fail to park their cars in designated parking areas as aforesaid or parks illegally, then the Lessor at its option shall have the right to tow such vehicles, at vehicle Lessor expense, upon such infraction.
- 10.4.1.8. The plumbing facilities shall not be used for any other purpose than that for which they are constructed, and no foreign substance of any kind shall be thrown therein, and the expense of any breakage, stoppage, or damage resulting from a violation of this provision shall be borne by Lessee, who shall, or whose employees, agents or invitees shall have caused it.
- 10.4.1.9. Lessee shall not burn any trash or garbage of any kind in or about the leased premises, or the Business Park.
- 10.4.1.10. Lessee and Lessee's employees and agents shall not solicit business in the parking or other common areas, nor shall Lessee distribute any handbills or other advertising matter in automobiles parked in the parking area or in other common areas.
- 10.5. Lessor reserves the right from time to time to amend or supplement the foregoing rules and regulations, and to adopt and promulgate additional rules and regulations applicable to the leased premises. Notice of such rules and regulations and amendments and supplements thereto, if any, shall be given to the Lessee.
- 10.6. Lessee agrees to comply with all such rules and regulations upon notice to Lessee from Lessor, provided that such rules and regulations shall apply uniformly to all Lessees of the Business Park.

11. INSURANCE AND INDEMNITY

11.1. LIABILITY INSURANCE.

Lessee shall, during the entire term hereof, keep in full force and effect a policy of public liability and property damage insurance with respect to the leased premises, the sidewalks in front of the leased premises, and the business operated by Lessee and any sub-Lessees of Lessee in the leased premises in which the limits of public liability shall be not less than \$1,000,000.00 for each occurrence and in which the property damage liability shall be not less than \$500,000.00 for each occurrence. The policy shall name Lessor, any person, firms or corporations designated by Lessor, and Lessee as insured, and shall contain a clause that the insurer will not cancel or change the insurance without first giving the Lessor ten (10) days prior written notice. The Insurance shall be in the insurance company approved by Lessor and a copy of the policy or a certificate of insurance shall be delivered to Lessor.

11.2. FIRE INSURANCE PREMIUM.

Lessor shall, at its cost and expense, maintain fire and extended coverage insurance throughout the term of this lease in an amount equal to at least ninety percent (90%) of the replacement value (exclusive of foundation and excavation costs) of the leased premises and/or the building of which the leased premises are a part. However, Lessee agrees to reimburse Lessor for Lessee's prorate share of any premiums for said fire and extended coverage insurance that may be charged during the term of this lease on the amount of such insurance which may be carried by Lessor on said premises or the building of which they are a part. This reimbursement charge will be paid monthly in advance as specified in Section 8.1.

11.3. PLATE GLASS

Lessee shall be responsible for any and all plate and other glass damaged or broken from any cause whatsoever in and about the leased premises. Lessee may elect to be self-insured for plate glass breakage of damage

Initialed Initialed

8

11.4. INDEMNIFICATION OF LESSOR.

Lessee will indemnify Lessor and save it harmless from and against any and all claims, actions, damages, liability and expense in connection with loss of life, personal injury and/or damage to property arising from or out of any occurrence in, upon or at the leased premises, or the occupancy or use by Lessee of the leased premises or any part thereof, or occasioned wholly or in part by any act or omission of Lessee, its agents, contractors, employees, servants, lessees or concessionaires. In case Lessor shall, without fault on its part, be made a party to any litigation commenced solely by or against Lessee, then Lessee shall protect and hold Lessor harmless and shall pay all costs, expenses and reasonable attorney's fees that may be incurred or paid by Lessor in enforcing the covenants and agreements in this lease.

12. UTILITIES

12.1, UTILITY SERVICES.

It is the Lessee's responsibility to a range for gas, electrical, and any telecommunication service required.

12.2. UTILITY CHARGES,

Lessee shall be solely responsible for and promptly pay all charges for heat, water, gas electricity or any other utility used or consumed in the leased premises. Should Lessor elect to supply the water gas, heat, electricity or any other utility used or consumed in the leased premises, Lessee agrees to purchase and pay for the same as additional rent as apportioned by the Lessor. In no event shall Lessor be liable for an interruption or failure in supply of any such utilities to the leased premises

12.3. Lessor does NOT cover, include or guarantee any communication service or communication infrastructure, capacity, bandwidth or speed of said service including, but not limited to internet service, cable services, T-1 lines, or telephone services, and or service and distribution lines within the interior or exterior of the building.

13. OFFSET STATEMENT, ATTORNMENT SUBORDINATION

13.1. OFFSET STATEMENT

Estoppel certificate: Lessee shall at any time, and from time to time, upon not less than ten (10) days prior written request by Lessor execute, acknowledge, and deliver to Lessor a statement in writing certifying that this lease is unmodified and in full force and effect (or, that there shall have been modifications, that the lease is modified and in full force and effect as modified and stating the modifications) and the dates to which the fixed rent and any other charges or payments have been paid in advance. It is expressly understood and agreed that any such statement delivered pursuant to this paragraph may be relied upon by Lessor or any prospective purchaser, mortgagee, assignee of any mortgagee, or the trustee or beneficiary of any deed of trust placed upon the demised premises or the real property of which the demised premises are a part.

13.2. ATTORNMENT.

Lessee shall, in the event any proceedings are brought for the foreclosure of, or in the event of exercise of the power of sale under any mortgage made by the Lessor covering the leased premises, attorn to the purchaser upon any such foreclosure or sale and recognize such purchaser as the Lessor under this lease.

13.3. SUBORDINATION.

Upon request of the Lessor, Lessee will subordinate its rights hereunder to the lien of any mortgage or mortgages or the lien resulting from any other method of financing or refinancing, now or hereafter in force against the land and buildings of which the leased premises are a part or upon any buildings hereafter placed upon the land of which the lease premises are a part, and to all advances made or hereafter to be made upon the security thereof. This section shall be self-operative and no further instrument of subordination shall be required by any mortgagee.

13.4. ATTORNEY-IN-FACT.

The Lessee, upon request of any party in interest, shall execute promptly such instruments or certificates to carry out the intent of Sections 13.2 and 13.3 above as shall be requested by the Lessor. The Lessee hereby irrevocably appoints the Lessor as attorney-in-fact for the Lessee with full power and authority to execute and deliver in the

Initialed Initialed

name of the Lessee any such instruments or certificates. If fifteen (15) days after the date of a written request by Lessor to execute such instruments, the Lessee shall not have executed the same, the Lessor may, at its option, cancel this lease without incurring any liability on account thereof, and the term hereby granted is expressly limited accordingly.

14. ASSIGNMENT AND SUBLETTING

14.1. CONSENT REQUIRED.

Lessee will not assign this lease in whole or in part, nor sublet all or any part of the leased premises, without the prior written consent of Lessor in each instance. The consent by Lessor to any assignment or subletting shall not constitute a waiver of the necessity for such consent to any subsequent assignment or subletting. This prohibition against assigning or subletting shall be construed to include a prohibition against any assignment or subletting by operation of law. If this lease be assigned, or if the leased premises or any part thereof be underlet or occupied by anybody other than Lessee, Lessor may collect rent from the assignment, underletting, occupant, and apply the net amount collected to the rent herein reserved, but no such assignment, underletting, occupancy or collection shall be deemed a waiver of this covenant, or the acceptance of the assignee, under-Lessee or occupant as Lessee, or a release of Lessee from the further performance by Lessee of covenants on the part of Lessee herein contained. Notwithstanding any assignment or sublease, Lessee shall remain fully liable on this lease and shall not be released from performing any of the terms, covenants and conditions of this lease.

- 14.1.1. The acceptance of rent from any other person shall not be deemed to be a waiver of any of the provisions of this lease or consent to the assignment of the demised premises. In the event any of the following are not satisfied in Lessor's sole and absolute discretion and Lessor chooses to withhold consent to such requested assignment or subletting, such withholding shall be deemed to be reasonable:
 - 14.1.1.1. The intended use of the premises by such proposed assignee or sub-lessee:
 - 14.1.1.1.1. Shall not conflict with any lease then in effect with respect to or with any other use of any other premises by any other occupant in the Business Park;
 - 14.1.1.1.2. Shall conform with Lessor's desired "Lessee mix" within the Business Park;
 - 14.1.1.1.3. Shall be in keeping with the quality or character of the Business Park or not constitute a nuisance:
 - 14.1.1.1.4. Such proposed assignee or sub-lessee shall be of sound financial net worth; proposed assignee must be approved by Lessor prior to any closing of any transfer.
 - 14.1.1.1.5. Such proposed assignee or sub-lessee shall satisfy Lessor that it has had not less than three [3] years of prior experience operating a successful business of the type intended to be operated in the premises;
 - 14.1.1.1.6. The impact of such proposed assignee or sub-lessee would not have a disadvantageous impact on the Common Area or other occupants of the Business Park.
- 14.1.2. Any assignment or subletting to which Lessor has not consented (other than permitted above)shall be null and void, and at Lessor's option shall constitute a material default hereunder, permitting Lessor to terminate this Lease in addition to all other rights and remedies now or hereafter available to Lessor hereunder or at law or in equity.
- 14.1.3.Lessee shall forward to Lessor for Lessor's approval (1) financial statements of such proposed assignee or sublessee; (2) a history of the business experience of such proposed assignee or sub-lessee; and (3) a copy of the proposed assignment or sublease, pursuant to which such assignee or sub-lessee shall assume all obligations on Lessee's part to be performed hereunder and an agreement to observe all of the terms, provisions, covenants, conditions and agreements hereunder.

Initialed UAF

10

15. WASTE, GOVERNMENTAL REGULATIONS

15.1. WASTE OR NUISANCE.

Lessee shall not commit or suffer to be committed any waste upon the leased premises or any nuisance or other act or thing which may disturb the quiet enjoyment of any other Lessee in the building in which the leased premises may be located, or in the Business Park, or which may disturb the quiet enjoyment of any person within five hundred feet of the boundaries of the Business Park.

15.2. GOVERNMENTAL REGULATIONS.

Lessee shall, at Lessee's sole cost and expense, comply with all of the requirements of all city, county, municipal, state, federal and other applicable governmental authorities, now in force, or which may hereafter be in force, pertaining to the said premises, including the installation of additional facilities as required for the conduct and continuance of Lessee's business, and shall faithfully observe in the use of the premises all municipal and county ordinances and state and federal statutes now in force or which may hereafter be in force.

16. DESTRUCTION OF LEASED PREMISES

16.1. RECONSTRUCTION OF DAMAGED PREMISES.

In the event the premises shall be rendered partially or totally unleasable as a result of fire or other casualty covered by insurance carried by Lessor, then the premises shall be promptly repaired to the extent of the insurance proceed received by Lessor therefor, unless Lessor shall elect not to rebuild as hereinafter provided, the fixed minimum rental shall be abated in proportion to the amount of the leased premises rendered unleasable shall be pro ratably reduced until so repaired. Lessor shall be obligated to cause such repairs to be made Unless Lessor, at its sole option elect to cause Lessee to make such repairs, in which event Lessee shall promptly complete the same and Lessor will make available to Lessee for the sole purpose of reconstruction of the premises such portion of the insurance proceeds received by Lessor from its insurance carrier, allocated to the premises. Any amount expended by Lessee in excess of such insurance proceeds received by Lessor and made available to Lessee shall be the sole obligation of Lessee. In the event of reconstruction or repair by Lessor, any amount expended by Lessor in repairing the improvements to the premises in excess of the proceeds of insurance received by Lessor allocated to the premises shall be repayable by Lessee to Lessor within ten (10) days after receipt by Lessee from Lessor of a statement setting forth the amount of such excess, together with interest at the highest rate from the date Lessor incurred such amount to the date of payment by Lessee.

- 16.2. Lessor and Lessor's insurance carrier shall determine the amount of insurance proceeds attributable to the damage of such improvements which determination shall be binding upon Lessor and Lessee. The party required hereunder to repair the damage to the premises shall reconstruct such premises in accordance with the working drawings originally approved by Lessor. In no event shall Lessor be required to repair or replace Lessee's merchandise, trade fixtures, furnishings or equipment. If more than thirty-five percent (35%) of the Floor Area of the building in which the premises are located shall be damaged or destroyed by fire or other casualty, or if during the last three (3) years of the Term more than twenty-five percent (25%) of the premises or the Floor Area of the building in which the premises are located ("Building") shall be damaged or destroyed by fire or other casualty then Lessor may either elect that the Building and/or premises, as the case may be, be repaired or rebuilt or, at its election to so terminate. Such notice shall be given, if at all, within ninety (90) days after the occurrence of such damage or destruction. If Lessor is required or elects to repair or rebuild or requires the Lessee to repair or rebuild the premises as herein provided, Lessee shall repair or replace its merchandise, trade fixtures, furnishings and equipment in a manner and to at least a condition equal to that prior to its damage or destruction...
- 16.3. If the premises cannot be restored within 90 days of the loss, or the loss occurs during the last 3 months of the lease, this lease can be terminated by either party. During such 90 day period Lessee shall not be required to pay monthly lease/rent to the Lessor.

Initialed Initialed

11

17. EMINENT DOMAIN

17.1. TOTAL CONDEMNATION.

If the whole of the leased premises shall be acquired or condemned by eminent domain for any public or quasi-public use or purpose, then the term of this lease shall cease and terminate as of the date of title vesting in such proceeding and all rentals shall be paid up to that date and Lessee shall have no claim against Lessor for the value of any unexpired term of this lease.

17.2. TOTAL PARKING AREA.

If the whole of the common parking areas in the Business Park shall be acquired or condemned by eminent domain for any public or quasi-public use or purpose, then the term of this lease shall cease and terminate as of the date of title vesting in such proceeding unless Lessor shall take immediate steps to provide other parking facilities substantially equal to the previously existing ratio between the common parking areas and the leased premises, and such substantially equal parking facilities shall be provided by Lessor at its own expense within ninety (90) days from the date of acquisition. In the event that Lessor shall provide such other substantially equal parking facilities, then this lease shall continue in full force and unexpired term of this lease.

17.3. PARTIAL CONDEMNATION.

If any part of the leased premises shall be acquired or condemned by eminent domain for any public or quasi-public use or purpose, and in the event that such partial taking or condemnation shall render the leased premises unsuitable for the business of the Lessee, then the term of this lease shall cease and terminate as of the date of title vesting in such proceeding and Lessee shall have no claim against Lessor for the value of any unexpired term of this lease and the Lessee shall not have any payment obligations for the remainder term of the lease from the date of such Partial Condemnation. In the event of a partial taking or condemnation which is not extensive enough and is mutually agreed upon as suitable for the business of the Lessee, then Lessor shall promptly restore the leased premises to a condition comparable to its condition at the time of such condemnation less the portion lost in the taking, and this lease shall continue in full force and effect.

17.4. LESSOR'S DAMAGES.

In the event of any condemnation or taking as hereinabove provided, whether whole or partial, the Lessee shall not be entitled to any part of the award, as damages or otherwise, for such condemnation and Lessor is to receive the full amount of such award, the Lessee hereby expressly waiving any right or claim to any part thereof.

17.5. LESSEE'S DAMAGES.

Although all damages in the event of any condemnation are to belong to the Lessor whether such damages are awarded as compensation for diminution in value of the leasehold or to the fee of the leased premises, Lessee shall have the right to claim and recover from the condemning authority, but not from Lessor, such compensation as may be separately awarded or recoverable by Lessee in Lessee's own right on account of any and all damage to Lessee's business by reason of the condemnation and for or on account of any cost or loss to which Lessee might be put in removing Lessee's merchandise, furniture, fixtures, leasehold improvements and equipment.

18. DEFAULT OF LESSEE

18.1. RIGHT TO RE-ENTER.

In the event of any failure of Lessee to pay any rental due hereunder within ten (10) days after the same shall be due, or any failure to perform any other of the terms, conditions or covenants of this lease to be observed or performed by Lessee for more than THREE (3) days after written notice of such default shall have been received by Lessee, or if Lessee shall become bankrupt or insolvent, or file any debtor proceedings, or take or have taken against Lessee in any court pursuant to any statute either of the United States or of any State a petition in bankruptcy or insolvency or for reorganization or for the appointment of a receiver or trustee of all or a portion of Lessee's property, or if Lessee makes an assignment for the benefit of creditors, or petitions for or enters into an arrangement, or if Lessee shall abandon said premises, or suffer this lease to be taken under any writ of execution, then Lessor besides other rights

Initialed 44

or remedies it may have, shall have the immediate right of re-entry and may remove all persons and property from the leased premises and such property may be removed and stored in a public warehouse or elsewhere at the cost of, and for the account of Lessee, all without service of notice or resort to legal process and without being deemed guilty of trespass, or becoming liable for any loss or damage which may be occasioned thereby.

18.2. RIGHT TO RELET.

Pursuant to 18.1, Should Lessor elect to re-enter, as above provided, or should it take possession pursuant to legal proceedings or pursuant to any notice provided for by law, it may either terminate this lease or it may from time to time without terminating this lease, make such reasonable alterations and repairs as may be necessary in order to relet the premises, and relet said premises or any part thereof for such term or terms (which may be for a term extending beyond the term of this lease) and at such rental or rentals and upon such other terms and conditions as Lessor in its sole discretion may deem advisable; upon each such reletting all rentals received by the Lessor from such reletting shall be applied, first, to the payment of any indebtedness other than rent due hereunder from Lessee to Lessor; second, to the payment of any costs and expenses of such reletting, including brokerage fees and attorney's fees and of costs of such alterations and repairs; third, to the payment of rent due and unpaid hereunder, and the residue, if any, shall be held by Lessor and applied in payment of future rent as the same may become due and payable hereunder. If such rentals received from such reletting during any month be less than that to be paid during that month by Lessee hereunder, Lessee shall pay any such deficiency to Lessor. Such deficiency shall be calculated and paid monthly. No such re-entry or taking possession of said premises by Lessor shall be construed as an election on its part to terminate this lease unless a written notice of such intention is given to Lessee or unless the termination thereof is decreed by a court of competent jurisdiction. Notwithstanding any such reletting without termination, Lessor may at any time thereafter elect to terminate this lease for such previous breach. Should Lessor at any time terminate this lease for any breach, in addition to any other remedies it may have, it may recover from Lessee all damages it may incur by reason of such breach, including the cost of recovering the leased premises, reasonable attorney's fees, and including the worth at the time of such termination of the excess, if any, of the amount of rent and charges equivalent to rent reserved in this lease for the remainder of the stated term over the currently agreed rental value of the leased premises as described in section 2.3 for the remainder of the stated term, all of which amounts shall be immediately due and payable from Lessee to Lessor.

18.3. LEGAL EXPENSES.

In the event of any action or proceeding brought by either party against the other under this lease, each party shall bear their own costs and fees of its attorneys in such action or proceeding, including costs of appeal, if any.

18.4. WAIVER OF RIGHTS OF REDEMPTION.

Lessee hereby expressly waives any and all rights of redemption granted by or under any present or future laws in the event of Lessee being evicted or dispossessed for any cause, or in the event of Lessor obtaining possession of the leased premises, by reason of the violation by Lessee or any of the covenants or conditions of this lease, or otherwise.

19. ACCESS BY LESSOR

19.1. RIGHT OF ENTRY.

Given 24 hours notice, Lessor or Lessor's agents shall have the right to enter the leased premises during normal business hours and make reasonable efforts not to impact the business of the Lessee to examine the same, and to show them to prospective purchasers or lessees of the building, and to make such repairs, alterations improvements or additions as Lessor may deem necessary or desirable, and Lessor shall be allowed to take all material into and upon said premises that may be required therefore without the same constituting an eviction of Lessee in whole or in part and the rent reserved shall in no wise abate while said repairs, alterations, improvements, or additions are being made, in the event it is necessary to temporarily close the business office during such repairs. During the six months prior to the expiration of the term of this lease or any renewal term, Lessor may exhibit the premises to prospective Lessees or purchasers, and place upon the premises the usual notices "To Let" or "For Sale" which notices Lessee shall permit to remain thereon without molestation. Lessor will take all appropriate measures to attempt to make repairs during times that does not impede on the Lessee's Business.

Initialed

20. LESSEE'S PROPERTY

20.1. TAXES ON LEASEHOLD.

Lessee shall be responsible for and shall pay before delinquency all municipal, county or state taxes assessed during the term of this lease against any leasehold interest or lessee's personal property of any kind, owned by or placed in, upon or about the leased premises by the Lessee.

20.2. LOSS AND DAMAGE.

Lessor shall not be liable for any damage to property of Lessee or of others located on the leased premises, nor for the loss of or damage to any property of Lessee or of others by theft or otherwise. Lessor shall not be liable for any injury or damage to persons or property resulting from fire, explosion, falling plaster, steam, gas, electricity, water, rain or leaks from any part of the leased premises or from the pipes, appliances or plumbing works or from the roof, street or sub-surface or from any other place or by dampness or by any other cause of whatsoever nature. Lessor shall not be liable for any such damage caused by other Lessees or persons in the leased premises, occupants of adjacent property, of the Business Park, or the public, or caused by operations in construction of any private, public or quasi-public work. All property of Lessee kept or stored on the leased premises shall be so kept or stored at the risk of Lessee only and Lessee shall hold Lessor harmless from any claims arising out of damage to the same, including subrogation claims by Lessee's insurance carriers, unless such damage shall be caused by the willful act or gross neglect of Lessor.

20.3. NOTICE BY LESSEE.

Lessee shall give immediate notice to Lessor in case of fire or accidents in the leased premises or in the building of which the premises are a part or of defects therein or in any fixtures or equipment.

21. HOLDING OVER, SUCCESSORS

21.1. HOLDING OVER.

Any holding over after the expiration of the term hereof, with the consent of the Lessor, shall be construed to be a tenancy from month to month at the rents herein specified (prorated on a monthly basis) and shall otherwise be on the terms and conditions herein specified, so far as applicable.

21.2. SUCCESSORS.

All rights and liabilities herein given to, or imposed upon, the respective parties hereto shall extend to and bind the several respective heirs, executors, administrators, successors, and assigns of the said parties; and if there shall be more than one Lessee, they shall all be bound jointly and severally by the terms, covenants and agreements herein. No rights, however, shall inure to the benefit of any assignee of Lessee unless the assignment to such assignee has been approved by Lessor in writing as provided in Section 16.1 hereof.

22. QUIET ENJOYMENT

22.1. LESSOR'S COVENANT.

Upon payment by the Lessee of the rents herein provided, and upon the observance and performance of all the covenants, terms and conditions on Lessee's part to be observed and performed, Lessee shall peaceably and quietly hold and enjoy the leased premises for the term hereby demised without hindrance or interruption by Lessor or any other person or persons lawfully or equitably claiming by, through or under the Lessor, subject, nevertheless, to the terms and conditions of this lease.

23. MISCELLANEOUS

23.1. WAIVER.

The waiver by Lessor of any breach of any term, covenant or condition herein contained shall not be deemed to be a waiver of such term, covenant or condition or any subsequent breach of the same or any other term, covenant or condition herein contained. The subsequent acceptance of rent hereunder by Lessor shall not be deemed to be a waiver of any preceding breach by Lessee of any term, covenant or condition of this lease, other than the failure of

Initialed Up Initialed

Lessee to pay the particular rental so accepted, regardless of Lessor's knowledge of such preceding breach at the time of acceptance of such rent. No covenant, term or condition of this lease shall be deemed to have been waived by Lessor, unless such waiver is in writing by Lessor.

23.2. ACCORD AND SATISFACTION.

No payment by Lessee or receipt by Lessor of a lesser amount than the monthly rent herein stipulated shall be deemed to be other than on account (of the earliest stipulated rent, nor shall any endorsement or statement on any check or any letter accompanying any check or payment as rent be deemed an accord and satisfaction, and Lessor may accept such check or payment without prejudice to Lessor's right to recover the balance of such rent or pursue any other remedy in this lease period.

23.3, ENTIRE AGREEMENT.

This lease and the Exhibits, and Rider, if any, attached hereto and forming a part hereof, set forth all the covenants, promises, agreements, conditions or understandings, either oral, or written, between them other than are herein set forth. Except as herein otherwise provided, no subsequent alteration, amendment, change or addition to this lease shall be binding upon Lessor or Lessee unless reduced to writing and signed by them.

23.4. NO PARTNERSHIP.

Lessor does not, in any way or for any purpose, become a partner of Lessee in the conduct of its business, or otherwise, or joint adventurer or a member of a joint enterprise with Lessee. The provisions of this lease relating to the percentage rent payable hereunder are included solely for the purpose of providing a method whereby the rent is to be measured and ascertained.

23.5. FORCE MAJEURE.

In the event that either party hereto shall be delayed or hindered in or prevented from the performance of any act required hereunder by reason of strikes, lock-outs, labor troubles, inability to procure materials, failure of power, restrictive governmental laws or regulations, riots, insurrection, war or other reason of a like nature not the fault of the party delayed in performing work or doing acts required under the terms of this lease, then performance of such act shall be excused for the period of the delay and the period for the performance of any such act shall be extended for a period equivalent to the period of such delay. The provisions of this Section 24.5 shall not operate to excuse Lessee from the prompt payment of rent, percentage rent, additional rent or any other payments required by the terms of this lease.

23.6. NOTICES.

(a) Any notice by Lessee to Lessor must be served by certified or registered mail, postage prepaid, addressed to Lessor at the address first hereinabove given or at such other address as Lessor may designate by written notice.

(b) Any notice by Lessor to Lessee must be served by Certified or registered mail, postage prepaid, addressed to Lessee at the leased premises or at such other address as Lessee shall designate by written notice.

23.7. CAPTIONS AND SECTION NUMBERS.

The captions, section numbers, article numbers, and index appearing in this lease are inserted only as a matter of convenience and in no way define, limit, construe, or describe the scope or intent of such sections or articles of this lease nor in any way affect this lease.

23.8. LESSEE DEFINED, USE OF PRONOUN

The word "Lessee" shall be deemed and taken to mean each and every person or party mentioned as a Lessee herein, be the same one or more; and if there shall be more than one Lessee, any notice required or permitted by the terms of this lease may be given by or to any one thereof, and shall have the same force and effect as if given by or to all thereof. The use of the neuter singular pronoun to refer to Lessor or Lessee shall be deemed a proper reference even though Lessor or Lessee may be an individual, a partnership, a corporation, or a group of two or more individuals or corporations. The necessary grammatical changes required to make the provisions of this lease apply in the plural sense where there is more than one Lessor or Lessee and to either corporations, partnerships, or individuals, males or females, shall in all instances be assumed as though in each case fully expressed.

23.9. PARTIAL INVALIDITY, If any term, covenant or condition of this lease or the application thereof to any person or circumstance shall, to any extent, be invalid or unenforceable, the remainder of this lease, or the application of such

Initialed CHP

term, covenant or condition to persons or circumstances other than those as to which it is held invalid or unenforceable, shall not be affected thereby and each term, covenant or condition of this lease shall be valid and be enforced to the fullest extent permitted by law.

24. RECORDING

24.1. Lessee shall not record this lease without the written consent of Lessor.

25. OPTION

- 25.1. If all terms and conditions have been carried out successfully, Lessor hereby grants Lessee the right, option and privilege of extending and renewing the term of the Lease for One (1) additional period(s) of Two (2) year(s) beginning on the termination date of the lease. Lessee shall notify Lessor no later than 90 days prior to expiration of this lease to inform Lessor of Lesse's intent to exercise the option. If notice is not received within the time specified, it will be assumed that Tenant will not exercise the option and the option shall be forfeited.
- 25.2. If assignee timely exercises the option to renew the term, the term shall thereafter automatically be extended and renewed for an additional two (2) year(s), which the extended and renewed term shall be on all of the same covenants, terms, and conditions as are contained in the Lease, except for the Base Rent which at the beginning of the first full calendar month in the option term, and if applicable, at one year intervals thereafter, the monthly rental shall increase, but not decrease, in proportion to any percentage increase in the Consumer Price Index as published by the Bureau of Labor Statistics (U.S. Cities Average for all Urban Consumers) between (i) the amount of that Index as last published prior to the beginning of the preceding 12-month period, and (ii) the amount of that Index as last published prior to the date on which the rental increase is to occur. In any event, the rent for any lease year shall be increased by not less than two (2%) per annum and shall not be increased by more than four (4%) per annum. In no event shall the monthly rental, as adjusted hereunder, be less than the applicable minimum rental set forth in section 2.01 above.
- 25.3. In the event that Lessee and Lessor cannot arrive at an agreement within 30 days after the expiration of this lease, all terms shall be considered on a month to month tenancy and the option shall be forfeited.

Daniel Fries

Lessor, Lessor's Agent Fries Properties Inc. 39678 Mission Blvd. Fremont, Ca 94539

Chris Floethe Lessee Construction Craft Training 26200 Industrial Blvd Hayward, CA 94545

18 2014

None

Date

Co-Lessee

Initialed

16

APPENDIX III TO DONATION AGREEMENT

26206 Hayward Lease

See attached.



39678 Mission Blvd. Fremont CA 94539 Phone (510) 793-1100 Fax (510) 793-1455 friesproperties.com

May 28, 2015

Construction Craft Training 26200 Industrial Blvd Hayward, CA 94545

Dear Chris Floethe,

We have received your request to active the two year option term on your lease agreement for 26200 and 26206 Industrial Blvd. in Hayward, California. As further provided, the Option period shall be on the same terms and conditions as the lease agreement, with exception for the following terms:

The commencement date of the option begins June 1, 2015 and the option term ends May 31, 2017.

Per the option clause in the lease, Base Rent for 26200 Industrial Blvd will be increased by 2% in accordance with the Consumer Price Index clause to \$3055.00. It will be adjusted in accordance with the consumer price index on the commencement of the second year of the option term.

Base Rent for 26206 Industrial Blvd will be increased by 2% in accordance with the Consumer Price Index clause to \$2142.00. It will be adjusted in accordance with the consumer price index on the commencement of the second year of the option term.

Sincerely,

delsey Ressnall

Kelsey Pressnall Administration Kelsey@friesproperties.com

Hayward Business Park Lease Agreement

Hayward Business Park Inc., herein called "Lessor" whose address is 26250 Industrial Boulevard, Hayward CA 94545 and Chris Floethe dba Construction Craft Training "Lessee" whose address is 26200 Industrial Blvd, Hayward, CA 94545, agree to the following terms and conditions relating to the lease of land and improvements within the Hayward Business Park.

1. GRANT AND TERM

1.1. LEASED PREMISES.

In consideration of the rents, covenants and agreements hereinafter reserved and contained on the part of Lessee to be observed and performed, the Lessor demises and leases to the Lessee and Lessee rents from Lessor, those certain premises, now in the Hayward Business Park (herein called the "Business Park"), commonly known as 26200 Industrial Blvd., in the City of Hayward, County of Alameda, California, premises consists of a store having approximately 3650 square feet herein called the "leased premises".

1.2. USE OF ADDITIONAL AREAS.

The use and occupation by the Lessee of the leased premises shall include the use in common with others entitled thereto of the common areas, employees' parking areas, service roads, loading facilities, sidewalks and customer car parking areas, and other facilities as may be designated from time to time by the Lessor, subject however to the terms and conditions of this agreement and to reasonable rules and regulations for the use thereof as prescribed from time to time by Lessor.

1.3. COMMENCEMENT OF TERM.

The term of this lease, and Lessee's obligation to pay rent, shall commence on June 1, 2014. In the event that the fixed date above does not occur on the first day of the month, Lessee shall pay rent for the fractional month on a per diem basis (calculated on the basis of a thirty day month) until the first day of the month when the term hereunder commences; and thereafter the minimum rent shall be paid in equal monthly installments on the first day of each and every month in advance.

1.4. ACCEPTANCE

Lessee shall, by entering into and occupying the demised premises be conclusively deemed to have accepted the same and to have acknowledged that said demised premises are in good order, condition and repair.

1.5. LENGTH OF TERM.

The term of this lease shall be for one (1) year and commencement of the term as provided in Section 1.3 hereof and ease shall expire on May 31, 2015.

1.6. EXCUSE OF LESSOR'S PERFORMANCE.

Anything in this agreement to the contrary notwithstanding, providing such cause is not due to the willful act or neglect of the Lessor, the Lessor shall not be deemed in default with respect to the performance of any of the terms, covenants and conditions of this lease if same shall be due to any strike, lockout, civil commotion, war-like operation, invasion, rebellion, hostilities, military or usurped power, sabotage, governmental regulations or controls, inability to obtain any material, service or financing rain or muddy conditions, through Act of God or other cause beyond the control of the Lessor.

2. RENT

2.1. MINIMUM RENT.

The fixed minimum rent during the term of this lease shall be payable by Lessee in monthly installments, or on or before the first day of each month in advance, at the office of the Lessor or at such other place designated by Lessor, without any prior demand therefore, and without any deduction or set-off whatsoever, and shall be as follows: Base Rent per month, is two thousand, nine hundred and ninety-five (\$2995.00) dollars.

Initialed

In the event the fixed minimum monthly rent is not paid to Lessor by Lessee within 10 (ten) days of the date on which it is due Lessee agrees to pay to Lessor as additional rent, a late-charge of 10% of rental amount due. Lessee further agrees to pay Lessor any costs incurred by Lessor in effecting the collection of such past due rent and late-charge including but not limited to reasonable fees of an attorney or collection agency. The parties hereto agree that all late charges shall be deemed additional rent and shall be due and payable as such within five (5) days of Lessee's receipt of a statement itemizing the same. Nothing herein contained shall limit any other remedy of Lessor.

2.2. REAL ESTATE TAXES

Lessee agrees to pay Lessee's pro rata share of all real property taxes and assessments which may be levied or assessed by any lawful authority against the land on which buildings are located and improvements thereon in the Business Park. Lessee shall pay said taxes monthly upon receipt from Lessor of a statement delineating Lessee share of said taxes and said share shall be paid within five (5) days after receipt of said statement. Lessee's pro rata share shall be apportioned according to the floor area of the demised premises as it relates to the total floor area of the building or buildings including the demised premises. All taxes for the year in which this lease commences shall be apportioned and adjusted.

As used herein, the term "real property tax" shall include any form of real estate tax or assessment, general, special, ordinary or extra ordinary, and license fee, commercial rental tax, improvement bond or bonds, levy or tax [other than inheritance, personal income or estate taxes] imposed upon the premises by any authority having the direct or indirect powers of tax, including any city, state or federal government, or any school, sanitary, fire, street, drainage or other improvement district thereof as against any legal or equitable interest of Lessor in the premises or in the real property of which the premises are a part, as against Lessor's business of leasing the premises. The term " real property tax" shall also include any tax, fee, levy, assessment or charge [1] in substitution of partially or totally, any tax, fee, levy, assessment or charge hereinabove included within the definition of real property tax or [3] which is imposed for a service or right not charged prior to June 1, 1978, or if previously charged has been increased since June 1, 1978.

2.3. ADDITIONAL RENT.

The Lessee shall pay as additional rent money required to be paid pursuant to sections 2.2, 8.1, 10.1, 11.2,11.4, and 12.1, and all other sums of money or charges required to be paid by Lessee under this lease, without any deductions or setoffs whatsoever, whether or not the same shall be designated "additional rent." If such amounts or charges are not paid at the time provided in this lease, the Lessee further agrees to pay to Lessor any cost incurred by Lessor in effectuating the collection of such past due amounts or charges including but not limited to fees of an attorney or a collection agency. The parties hereto agree that all of the hereinabove mentioned charges shall be deemed additional rent and shall be all due and payable as such within five (5) days of Lessee's receipt of a statement itemizing the same. Nothing herein contained shall limit any other remedy of Lessor.

- 2.4. ADJUSTMENTS TO MINIMUM RENT.
 - 2.4.1. Commencing with the end of the first lease year, the minimum rent may be adjusted and after the end of each subsequent lease year during the term of this lease in accordance with the Consumer Price Index, hereinafter called "CPI" as published by the Bureau of Labor Statistics (U.S. Cities Average for all Urban Consumers) between (i) the amount of that Index as last published prior to the beginning of the preceding 12-month period, and (ii) the amount of that Index as last published prior to the date on which the rental increase is to occur
 - 2.4.2. The CPI as of the first month shall be designated as the Beginning Index;
 - 2.4.3. Promptly after the end of the first Lease Year and each year thereafter, the annual rental shall be adjusted as follows. If the CPI has increased over the beginning index, the minimum monthly rent payable during the next lease year shall be set my multiplying the minimum rent set forth in paragraph 4.1 by a fraction, the numerator of which is the CPI and the denominator is the beginning index. As soon as the minimum monthly rent for the next lease year is set, the Lessor shall give Lessee notice of the amount of the new minimum monthly rent. If the index is changed so that the base year differs from that used as of the date most immediately preceding the date the first lease year commences, the CPI shall be converted in accordance with conversion factors.

Initialed Initialed

published by the United States Department of Labor Statistics. If the index is discontinued or revised during the term, such other government index with which it is replaced shall be used in order to obtain substantially the same result as would be obtained if the index were not discontinued. In no event shall the annual increase be less than 2% or more than 4%.

3. CONSTRUCTION OF LEASED PREMISES

3.1. CHANGES AND ADDITIONS TO BUILDINGS.

Lessor hereby reserves the right at any time to make alterations or additions to and to build additional stories on the building in which the premises are contained and to build adjoining the same. Lessor also reserves the right to construct other buildings or improvements in the Business Park from time to time and to make alterations thereof or additions thereto and to build additional stories on any such building or buildings and to build adjoining same. Easements for light and air are not included in the leasing of these premises to Lessee. Lessor further reserves the exclusive right to the roof except as provided in this lease agreement.

3.2. RIGHT TO RELOCATE.

Lessor reserves the right at any time to vary and adjust the size of the various buildings, co-Lessees, automobile parking areas, and other common areas as described in this lease, provided, however, that said parking area (including landscaped and common areas) shall at all times be in compliance with the minimum requirements of the City of Hayward, County of Alameda, State of California.

4. CONDUCT OF BUSINESS BY LESSEE

4.1. USE OF PREMISES.

Lessee shall use the leased premises for the purpose of conducting the business of: Office & Classroom. Lessee shall occupy the leased premises within thirty (30) days after the date of the notice provided for in Section 1.3 hereof, and shall conduct continuously in the leased premises the business above stated. Lessee will not use or permit, or suffer the use of the leased premises for any other business or purpose without Lessor's approval.

4.2. CONDUCT OF BUSINESS.

Lessee acknowledges that Lessor makes no representations as to the present or future condition of the demised premises, or to the fitness, desirability or zoning hereof for any particular purpose, and Lessor shall not be liable for any charges therein or additions thereto required by any public authority. Any permits or requirements of any kind pertaining to the operation of Lessee's business will be Lessee's responsibility and any construction or special equipment required by public authority to enable Lessee to conduct his business shall likewise be Lessee's sole cost and responsibility.

4.3. ACCEPTANCE OF PREMISES

Acceptance of demised premises except as may be otherwise herein provided, Lessees shall be entering into and occupying the demised premises shall be deemed to have accepted the demised premises "as is" and to have acknowledged that the same are as represented by Lessor and in good order, condition and repair. Lessee acknowledges that it has reviewed and examined the demised premises prior to the execution hereof to its complete satisfaction.

4.4. OPERATION OF BUSINESS.

Lessee shall operate one hundred percent (100%) of the leased premises during the entire term of this lease with due diligence and efficiency unless prevented from doing so by causes beyond Lessee's control. Subject to inability by reason of strikes or labor disputes, Lessee shall carry at all times in said premises a stock of merchandise of such size, character and quality as shall be reasonably designed to produce the maximum return to Lessee. Lessee shall conduct its business in the leased premises during the regular customary days and hours for such type of business in the City or trade area in which the Business Park is located.

Initialed Initialed

4.5. STORAGE

Lessee shall warehouse, store and/or stock in the leased premises only such goods, wares and merchandise as Lessee intends to offer for sale at retail at, in, from or upon the leased premises. No auction, fire or bankruptcy sales may be conducted in the leased premises without the previous written consent of Lessor.

5. GRANT OF CONCESSIONS

5.1. CONDITIONS TO GRANT.

The provision against subletting elsewhere contained in this lease shall not prohibit Lessee from granting concessions for the operation of one or more departments of the business which Lessee is permitted by Section 4.1 to conduct in or upon the leased premises; provided, however that [a] each such concession may be granted only upon receipt by Lessee of the written consent of the Lessor and shall be subject to all terms and provisions of this lease; [b] at least seventy-five percent (75%) of the sales floor area of the leased premises shall be at all times devoted to the business of and be operated by Lessee.

6. SECURITY DEPOSIT

6.1. AMOUNT OF DEPOSIT.

Lessee, contemporaneously with the execution of this lease, will deposit with Lessor the sum of seven hundred and seventy-seven dollars and fifty cents (\$777.50). Said deposit shall be held by Lessor, without liability for interest, as security for the faithful performance by Lessee of all of the terms, covenants, and conditions of this lease by said Lessee to be kept and performed during the term hereof. If at any time during the term of this lease any of the rent herein reserved shall be overdue and unpaid, or any other sum payable by Lessee to Lessor hereunder shall be overdue and unpaid then Lessor may, at the option of Lessor (but Lessor shall not be required to), appropriate and apply any portion of said deposit to the payment of any such overdue rent or other sum. It is expressly understood that this sum does not apply toward rent.

6.2. USE AND RETURN OF DEPOSIT.

In the event of the failure of Lessee to keep and perform any of the terms, covenants and conditions of this lease to be kept and performed by Lessee, then the Lessor at its option may, after terminating this lease, appropriate and apply said entire deposit, or so much thereof as may be necessary, to compensate the Lessor for all loss or damage sustained or suffered by Lessor due to such breach on the part of Lessee. Should the entire deposit, or any portion thereof, be appropriated and applied by Lessor for the payment of overdue rent or other sums due and payable to Lessor by Lessee hereunder, then Lessee shall, upon the written demand of Lessor, forthwith remit to Lessor a sufficient amount in cash to restore said security to the original sum deposited and Lessee's failure to do so within five (5) days after receipt of such demand shall constitute a breach of this lease. Should Lessee comply with all of said terms, covenants and conditions and promptly pay all of the rental herein provided for as it falls due, and all other sums payable by Lessee to Lessor hereunder, the said deposit shall be returned in full to Lessee within two weeks from the end of the term of this lease, or upon the earlier termination of this lease.

6.3. TRANSFER OF DEPOSIT.

Lessor may deliver the funds deposited hereunder by Lessee to the purchaser of Lessor's interest in the leased premises, in the event that such interest is sold, and thereupon Lessor shall be discharged from any further liability with respect to such deposit.

7. PARKING AND COMMON USE AREAS AND FACILITIES

7.1. CONTROL OF COMMON AREAS BY LESSOR.

All automobile parking areas, driveways, entrances and exits thereto, and other facilities furnished by Lessor In or near the Business Park, including employee parking areas, the truck way or ways, loading docks, package pick-up stations, pedestrian sidewalks and ramps, landscaped areas, exterior stairways, first-aid stations, comfort stations and other areas and improvements provided by Lessor for the general use, in common, of Lessees, their officers, agents,

Initialed O Initialed

employees and customers, shall at all times be subject to the exclusive control and management of Lessor, and Lessor shall have the right from time to time to establish, modify and enforce reasonable rules and regulations with respect to all facilities and areas mentioned in this Article. Lessor shall have the right to construct, maintain and operate lighting facilities on all said areas and improvements, to police the same, from time to time to change the area, level, location and arrangement of parking areas and other facilities hereinabove referred to; to restrict parking by Lessees, their officers, agents and employees to employee parking areas; to enforce parking charges (by operation of meters or otherwise), with appropriate provisions for free parking ticket validating by Lessees; to close all or any portion of said areas or facilities to such extent as may, in the opinion of Lessor's council, be legally sufficient to prevent a dedication thereof or the accrual of any rights to any person or the public therein; to close temporarily all or any portion of the parking areas or facilities; to discourage non-customer parking; and to do and perform such other acts in and to said areas and improvements as, in the use of good business judgment, the Lessor shall determine to be advisable with a view to the improvement of the convenience and use thereof by Lessees, their officers, agents, employees and customers. Lessor will operate and maintain the common facilities referred to above in such a manner as Lessor, in its sole discretion, shall determine from time to time. Without limiting the scope of such discretion, Lessor shall have the full right and authority to employ all personnel and to make all rules and regulations pertaining to and necessary for the proper operation and maintenance of the common areas and facilities.

7.2. LICENSE.

All common areas and facilities not within the leased premises, which Lessee may be permitted to use and occupy, are to be used and occupied under a revocable license, and if any such license be revoked, or if the amount of such areas be diminished, Lessor shall not be subject to any liability nor shall Lessee be entitled to any compensation or diminution or abatement of rent, nor shall such revocation or diminution of such areas be deemed constructive or actual eviction.

8. COST OF MAINTENANCE OF COMMON AREAS [NNN]

- 8.1. LESSEE TO BEAR PRO RATA SHARE OF EXPENSE.
 - 8.1.1. In each lease year, as defined in Section 2.1 hereof, Lessee will pay to Lessor, in addition to the rentals specified in Article II hereof, as further additional rent, subject to the limitation hereinafter set forth, a proportion of the Business Park's operating cost, hereinafter defined, based upon the ratio of the square feet of all the building space in the Business Park.
 - 8.1.2. For the purpose of this Section 8.1 the "Business Park's operating cost" means the total cost and expense incurred in operating and maintaining the common facilities, hereinafter defined, actually used or available for use by Lessee and the employees, agents, servants, customers and other invitees of Lessee, specifically including without limitation, gardening and landscaping, the cost of public liability and property damage insurance, fire and extended coverage insurance, joint utilities, painting, and repairs (not capital in nature), property management, employment of security guards, bookkeeping, real estate property taxes and assessments thereon, and the cost of personnel to implement such services, to direct parking, and to police the common facilities. "Common Facilities" means all areas, space, equipment, and special services provided by Lessor for the common or joint use and benefit of the occupants of the Business Park, their employees, agents, servants, customers and other invitees, including without limitation parking areas, access roads, driveways, retaining walls, landscaped areas, truck service ways or tunnels, loading docks, pedestrian malls, courts, stairs, ramps and sidewalks, comfort and first aid stations, washrooms, parcel pick-up stations and Business Park sign rental.
 - 8.1.3. The additional rent to be paid in this Section 8.1 and also Sections 2.2, 11.1 and 11.2 shall be paid monthly in advance. The monthly charge shall be estimated for a twelve (12) month period and adjusted annually upwards or downwards to reflect the actual cost incurred. Lessee will then be credited or billed accordingly at year end to reflect this adjustment. Lessee shall have the right to review the breakdown on the operating costs that are incurred for the premises once a year, should Lessee request.

Initialed

8.1.4. Changes in any particular floor area occurring during any monthly period shall be effective on the first day of the next succeeding monthly period, and the amount of any floor area in effect for the whole of any monthly period shall be the average of the total amounts in effect on the first day of each calendar month in such monthly period.

9. SIGNS, AWNINGS, CANOPIES, FIXTURES, ALTERATIONS

9.1. INSTALLATION BY LESSEE.

All fixtures installed by Lessee shall be new or completely reconditioned. Lessee shall not make or cause to be made any alterations, additions or improvements or install or cause to be installed any trade fixtures, exterior signs, floor covering, interior or exterior lighting, plumbing fixtures, shades or awnings or make any changes to the store front without first obtaining Lessor's written approval and consent. Lessee shall present to the Lessor plans and specifications for such work at the time approval is sought.

9.2. REMOVAL AND INSURANCE BY LESSEE.

All alterations, decorations, additions and improvements made by the Lessee, or made by the Lessor on the Lessee's behalf by agreement under this lease, shall remain the property of the Lessee for the term of the lease, or any extension or renewal thereof. The Lessee shall at all times maintain fire insurance with extended coverage in the name of the Lessor and the Lessee, in an amount adequate to cover the cost of replacement of all alterations, decorations, additions or improvements in the event of fire or extended coverage loss. Lessee shall deliver to the Lessor certificates of such fire insurance policies which shall contain a clause requiring the insurer to give the Lessor ten (10) days notice of cancellation of such policies. Such alterations, decorations, additions and improvements shall not be removed from the premises without prior consent in writing from the Lessor. Upon expiration of this lease, or any renewal term thereof, the Lessee shall remove all such alterations, decorations, additions and improvements, and restore the leased premises as provided in Section 10.3 hereof. If the Lessee fails to remove such alterations, decorations, additions and improvements and restore the leased premises, then upon the expiration of this lease, or any renewal thereof, and upon the Lessee's removal from the premises, all such alterations, decorations, additions and improvements and restore the leased premises, then upon the expiration of this lease, or any renewal thereof, and upon the Lesse's removal from the premises, all such alterations, decorations, additions and improvements and restore the leased premises, all such alterations, decorations, additions and improvements and restore the leased premises, all such alterations, decorations, additions and improvements and restore the leased premises, all such alterations, decorations, additions and improvements and restore the leased premises, all such alterations, decorations, additions and improvements and restore the leased premises, all such alterations, dec

9.3. LESSEE SHALL DISCHARGE ALL LIENS.

Lessee shall promptly pay all contractors and material men, if such were employed by Lessee or Lessee's agents for work or modifications in the leased premises, so as to minimize the possibility of a lien attaching to the leased premises, and should any such lien be made or filed, Lessee shall bond against or discharge the same within ten (10) days after written request by Lessor.

9.4. SIGNS, AWNINGS AND CANOPIES.

Lessee will not place or suffer to be placed or maintained on any exterior door, wall or window of the leased premises any sign, awning or canopy, or advertising matter or other thing of any kind, and will not place or maintain any decoration, lettering or advertising matter on the glass of any window or door of the leased premises without first obtaining Lessor's written approval and consent. Lessee further agrees to maintain such sign, awning, canopy, decoration, lettering, advertising matter or other thing as may be approved in good condition and repair at all times. Lessee agrees, at Lessee's sole cost, to obtain a sign in strict conformance with Lessor's sign criteria as to design, material, color, location, size, and letter style.

10. MAINTENANCE OF LEASED PREMISES

10.1. MAINTENANCE BY LESSEE.

10.1.1.Lessee shall at all times keep the leased premises (including maintenance of exterior entrances, all glass and show window moldings) and all partitions, doors, door jams, door closers, door hardware, fixtures, equipment and appurtenances thereof (including electrical, lighting, heating and plumbing fixtures, and any air conditioning system, including leaks around ducts, pipes, vents, or other parts of the air conditioning, heating or plumbing systems which protrude through the roof) in good order, condition and repair including

Initialed

replacements (including reasonably periodic painting as determined by Lessor), damage by unavoidable casualty excepted, except for structural portions of the premises, which shall be maintained by Lessor, but if Lessor is required to make repairs to structural portions by reason of Lessee's negligent acts or omission to act, Lessor may add the cost of such repairs to the rent which shall thereafter become due.

- 10.1.2.Lessee shall comply with all legal requirements now or hereafter affecting or applying to the premises at Lessee's sole cost and expense.
- 10.1.3.Lessee shall keep the premises and all other parts of the Business Park free from any and all liens arising out of any work performed, materials furnished or obligations incurred by Lessee or for Lessee by Lessee's Agents, and agrees to bond against, discharge any mechanic's or material men's lien, or post security with Lessor, satisfactory to Lessor, within ten (10) days after the filing of any such lien. In no event said bond and security shall be in an amount less than twice the amount of such lien. In the event Lessee fails to do any of the following within said ten (10) day period, Lessor may pay the underlying claim thereof, or otherwise discharge said lien, at Lessee's sole cost and expense. Lessee shall reimburse Lessor therefore, together with interest at the interest rate of ten percent (10%) within ten (10) days after Lessor's demand therefore.
- 10.1.4.Lessee expressly waives Sections 1941 and 1933(i) of the California Civil Code, and all other rights to make repairs at the expense of Lessor or deduct any amounts from rent as provided in any statute or law in effect during the term of this Lease in any.
- 10.2. MAINTENANCE BY LESSOR.

If Lessee refuses or neglects to repair properly as required hereunder and to the reasonable satisfaction of Lessor as soon as reasonably possible after written demand, Lessor may make such repairs without liability to Lessee for any loss or damage that may accrue to Lessee's merchandise, fixtures, or other property or to Lessee's business by reason thereof, and upon completion thereof, Lessee shall pay Lessor's reasonable costs for making such repairs, upon presentation of bill thereof, as additional rent. Said bill shall include interest at ten (10%) percent on said cost beginning fifteen (15) days after the date of completion and written notification of such completion of repairs by Lessor to the Lessee.

10.3. SURRENDER OF PREMISES.

At the expiration of the tenancy hereby created, Lessee shall surrender the leased premises in the same condition as the leased premises were in upon delivery of possession thereto under this lease, reasonable wear and tear excepted, and damage by unavoidable casualty excepted to the extent that the same is covered by Lessor's fire insurance policy with extended coverage endorsement, and shall surrender all keys for the leased premises to Lessor at the place then fixed for the payment of rent and shall inform Lessor of all combinations on locks, safes and vaults, if any, in the leased premises. Lessee shall remove all its trade fixtures, and any alterations or improvements as provided in Section 9.2 hereof, before surrendering the premises as aforesaid and shall repair any damage to the leased premises caused thereby. Lessee's obligation to observe or perform this covenant shall survive the expiration or other termination of the term of this lease.

- 10.4. RULES AND REGULATIONS.
 - 10.4.1.The Lessee agrees as follows:
 - 10.4.1.1. All loading and unloading of goods shall be done only at such times, in the areas, and through the entrances, designated for such purposes by Lessor.
 - 10.4.1.2. The delivery or shipping of merchandise, supplies and fixtures to and from the leased premises shall be subject to such rules and regulations as in the judgment of the Lessor are necessary for the proper operation of the leased premises or Business Park.
 - 10.4.1.3. All garbage and refuse shall be kept in the kind of container specified by Lessor, and shall placed outside of the premises prepared for collection in the manner and at the times and places specified by Lessor. If Lessor shall provide or designate a service for picking up refuse and garbage, Lessee shall use same at Lessee's cost. Lessee shall pay the cost of removal of any of Lessee's refuse or rubbish.

Initialed

- 10.4.1.4. No aerial shall be erected on the roof or exterior walls of the premises, or on the grounds, without in each instance, the written consent of the Lessor any aerial so installed without such written consent shall be subject to removal without notice at any time.
- 10.4.1.5. No loud speakers, televisions, phonographs, radios or other devices shall be used in a manner so as to be heard or seen outside of the premises without the prior written consent of the Lessor.
- 10.4.1.6. The outside areas immediately adjoining the premises shall be kept clean and free from dirt and rubbish by the Lessee to the satisfaction of the Lessor and Lessee shall not place or permit any obstructions or merchandise in such areas.
- 10.4.1.7. Lessee and Lessee's employees shall park their cars only in those portions of the parking area designated for that purpose by Lessor. In the event the Lessee or its employees fail to park their cars in designated parking areas as aforesaid or parks illegally, then the Lessor at its option shall have the right to tow such vehicles, at vehicle Lessor expense, upon such infraction.
- 10.4.1.8. The plumbing facilities shall not be used for any other purpose than that for which they are constructed, and no foreign substance of any kind shall be thrown therein, and the expense of any breakage, stoppage, or damage resulting from a violation of this provision shall be borne by Lessee, who shall, or whose employees, agents or invitees shall have caused it.
- 10.4.1.9. Lessee shall not burn any trash or garbage of any kind in or about the leased premises, or the Business Park.
- 10.4.1.10. Lessee and Lessee's employees and agents shall not solicit business in the parking or other common areas, nor shall Lessee distribute any handbills or other advertising matter in automobiles parked in the parking area or in other common areas.
- 10.5. Lessor reserves the right from time to time to amend or supplement the foregoing rules and regulations, and to adopt and promulgate additional rules and regulations applicable to the leased premises. Notice of such rules and regulations and amendments and supplements thereto, if any, shall be given to the Lessee.
- 10.6. Lessee agrees to comply with all such rules and regulations upon notice to Lessee from Lessor, provided that such rules and regulations shall apply uniformly to all Lessees of the Business Park.

11. INSURANCE AND INDEMNITY

11.1. LIABILITY INSURANCE.

Lessee shall, during the entire term hereof, keep in full force and effect a policy of public liability and property damage insurance with respect to the leased premises, the sidewalks in front of the leased premises, and the business operated by Lessee and any sub-Lessees of Lessee in the leased premises in which the limits of public liability shall be not less than \$1,000,000.00 for each occurrence and in which the property damage liability shall be not less than \$500,000.00 for each occurrence. The policy shall name Lessor, any person, firms or corporations designated by Lessor, and Lessee as insured, and shall contain a clause that the insurer will not cancel or change the insurance without first giving the Lessor ten (10) days prior written notice. The insurance shall be in the insurance company approved by Lessor and a copy of the policy or a certificate of insurance shall be delivered to Lessor.

11.2. FIRE INSURANCE PREMIUM.

Lessor shall, at its cost and expense, maintain fire and extended coverage insurance throughout the term of this lease in an amount equal to at least ninety percent (90%) of the replacement value (exclusive of foundation and excavation costs) of the leased premises and/or the building of which the leased premises are a part. However, Lessee agrees to reimburse Lessor for Lessee's prorata share of any premiums for said fire and extended coverage insurance that may be charged during the term of this lease on the amount of such insurance which may be carried by Lessor on said premises or the building of which they are a part. This reimbursement charge will be paid monthly in advance as specified in Section 8.1.

11.3. PLATE GLASS

Lessee shall be responsible for any and all plate and other glass damaged or broken from any cause whatsoever in and about the leased premises. Lessee may elect to be self-insured for plate glass breakage of damage

Initialed Initialed

11.4. INDEMNIFICATION OF LESSOR.

Lessee will indemnify Lessor and save it harmless from and against any and all claims, actions, damages, liability and expense in connection with loss of life, personal injury and/or damage to property arising from or out of any occurrence in, upon or at the leased premises, or the occupancy or use by Lessee of the leased premises or any part thereof, or occasioned wholly or in part by any act or omission of Lessee, its agents, contractors, employees, servants, lessees or concessionaires. In case Lessor shall, without fault on its part, be made a party to any litigation commenced solely by or against Lessee, then Lessee shall protect and hold Lessor harmless and shall pay all costs, expenses and reasonable attorney's fees that may be incurred or paid by Lessor in enforcing the covenants and agreements in this lease.

12. UTILITIES

12.1. UTILITY SERVICES.

It is the Lessee's responsibility to arrange for gas, electrical, and any telecommunication service required.

12.2. UTILITY CHARGES.

Lessee shall be solely responsible for and promptly pay all charges for heat, water, gas electricity or any other utility used or consumed in the leased premises. Should Lessor elect to supply the water gas, heat, electricity or any other utility used or consumed in the leased premises, Lessee agrees to purchase and pay for the same as additional rent as apportioned by the Lessor. In no event shall Lessor be liable for an interruption or failure in supply of any such utilities to the leased premises.

12.3. Lessor does NOT cover, include or guarantee any communication service or communication infrastructure, capacity, bandwidth or speed of said service including, but not limited to internet service, cable services, T-1 lines, or telephone services, and or service and distribution lines within the interior or exterior of the building.

13. OFFSET STATEMENT, ATTORNMENT SUBORDINATION

13.1. OFFSET STATEMENT

Estoppel certificate: Lessee shall at any time, and from time to time, upon not less than ten (10) days prior written request by Lessor execute, acknowledge, and deliver to Lessor a statement in writing certifying that this lease is unmodified and in full force and effect (or, that there shall have been modifications, that the lease is modified and in full force and effect as modified and stating the modifications) and the dates to which the fixed rent and any other charges or payments have been paid in advance. It is expressly understood and agreed that any such statement delivered pursuant to this paragraph may be relied upon by Lessor or any prospective purchaser, mortgagee, assignee of any mortgagee, or the trustee or beneficiary of any deed of trust placed upon the demised premises or the real property of which the demised premises are a part.

13.2. ATTORNMENT.

Lessee shall, in the event any proceedings are brought for the foreclosure of, or in the event of exercise of the power of sale under any mortgage made by the Lessor covering the leased premises, attorn to the purchaser upon any such foreclosure or sale and recognize such purchaser as the Lessor under this lease.

13.3. SUBORDINATION.

Upon request of the Lessor, Lessee will subordinate its rights hereunder to the lien of any mortgage or mortgages or the lien resulting from any other method of financing or refinancing, now or hereafter in force against the land and buildings of which the leased premises are a part or upon any buildings hereafter placed upon the land of which the lease premises are a part, and to all advances made or hereafter to be made upon the security thereof. This section shall be self-operative and no further instrument of subordination shall be required by any mortgagee.

13.4. ATTORNEY-IN-FACT.

The Lessee, upon request of any party in interest, shall execute promptly such instruments or certificates to carry out the intent of Sections 13.2 and 13.3 above as shall be requested by the Lessor. The Lessee hereby irrevocably appoints the Lessor as attorney-in-fact for the Lessee with full power and authority to execute and deliver in the

Initialed Initialed

name of the Lessee any such instruments or certificates. If fifteen (15) days after the date of a written request by Lessor to execute such instruments, the Lessee shall not have executed the same, the Lessor may, at its option, cancel this lease without incurring any liability on account thereof, and the term hereby granted is expressly limited accordingly.

14. ASSIGNMENT AND SUBLETTING

14.1. CONSENT REQUIRED.

Lessee will not assign this lease in whole or in part, nor sublet all or any part of the leased premises, without the prior written consent of Lessor in each instance. The consent by Lessor to any assignment or subletting shall not constitute a waiver of the necessity for such consent to any subsequent assignment or subletting. This prohibition against assigning or subletting shall be construed to include a prohibition against any assignment or subletting by operation of law. If this lease be assigned, or if the leased premises or any part thereof be underlet or occupied by anybody other than Lessee, Lessor may collect rent from the assignee, under-Lessee or occupant, and apply the net amount collected to the rent herein reserved, but no such assignment, underletting, occupancy or collection shall be deemed a waiver of this covenant, or the acceptance of the assignee, under-Lessee or occupant as Lessee, or a release of Lessee from the further performance by Lessee of covenants on the part of Lessee herein contained. Notwithstanding any assignment or sublease, Lessee shall remain fully liable on this lease and shall not be released from performing any of the terms, covenants and conditions of this lease.

- 14.1.1. The acceptance of rent from any other person shall not be deemed to be a waiver of any of the provisions of this lease or consent to the assignment of the demised premises. In the event any of the following are not satisfied in Lessor's sole and absolute discretion and Lessor chooses to withhold consent to such requested assignment or subletting, such withholding shall be deemed to be reasonable:
 - 14.1.1.1. The intended use of the premises by such proposed assignee or sub-lessee:
 - 14.1.1.1. Shall not conflict with any lease then in effect with respect to or with any other use of any other premises by any other occupant in the Business Park;
 - 14.1.1.1.2. Shall conform with Lessor's desired "Lessee mix" within the Business Park;
 - 14.1.1.3.Shall be in keeping with the quality or character of the Business Park or not constitute a nuisance;
 - 14.1.1.1.4. Such proposed assignee or sub-lessee shall be of sound financial net worth; proposed assignee must be approved by Lessor prior to any closing of any transfer.
 - 14.1.1.5. Such proposed assignee or sub-lessee shall satisfy Lessor that it has had not less than three [3] years of prior experience operating a successful business of the type intended to be operated in the premises;
 - 14.1.1.1.6. The impact of such proposed assignee or sub-lessee would not have a disadvantageous impact on the Common Area or other occupants of the Business Park.
- 14.1.2. Any assignment or subletting to which Lessor has not consented (other than permitted above)shall be null and void, and at Lessor's option shall constitute a material default hereunder, permitting Lessor to terminate this Lease in addition to all other rights and remedies now or hereafter available to Lessor hereunder or at law or in equity.
- 14.1.3.Lessee shall forward to Lessor for Lessor's approval (1) financial statements of such proposed assignee or sublessee; (2) a history of the business experience of such proposed assignee or sub-lessee; and (3) a copy of the proposed assignment or sublease, pursuant to which such assignee or sub-lessee shall assume all obligations on Lessee's part to be performed hereunder and an agreement to observe all of the terms, provisions, covenants, conditions and agreements hereunder.

Initialed

15. WASTE, GOVERNMENTAL REGULATIONS

15.1. WASTE OR NUISANCE.

Lessee shall not commit or suffer to be committed any waste upon the leased premises or any nuisance or other act or thing which may disturb the quiet enjoyment of any other Lessee in the building in which the leased premises may be located, or in the Business Park, or which may disturb the quiet enjoyment of any person within five hundred feet of the boundaries of the Business Park.

15.2. GOVERNMENTAL REGULATIONS.

Lessee shall, at Lessee's sole cost and expense, comply with all of the requirements of all city, county, municipal, state, federal and other applicable governmental authorities, now in force, or which may hereafter be in force, pertaining to the said premises, including the installation of additional facilities as required for the conduct and continuance of Lessee's business, and shall faithfully observe in the use of the premises all municipal and county ordinances and state and federal statutes now in force or which may hereafter be in force.

16. DESTRUCTION OF LEASED PREMISES

16.1. RECONSTRUCTION OF DAMAGED PREMISES.

In the event the premises shall be rendered partially or totally unleasable as a result of fire or other casualty covered by insurance carried by Lessor, then the premises shall be promptly repaired to the extent of the insurance proceed received by Lessor therefor, unless Lessor shall elect not to rebuild as hereinafter provided, the fixed minimum rental shall be abated in proportion to the amount of the leased premises rendered unleasable shall be pro ratably reduced until so repaired. Lessor shall be obligated to cause such repairs to be made Unless Lessor, at its sole option elect to cause Lessee to make such repairs, in which event Lessee shall promptly complete the same and Lessor will make available to Lessee for the sole purpose of reconstruction of the premises such portion of the insurance proceeds received by Lessor from its insurance carrier, allocated to the premises. Any amount expended by Lessee in excess of such insurance proceeds received by Lessor and made available to Lessee shall be the sole obligation of Lessee. In the event of reconstruction or repair by Lessor, any amount expended by Lessor in repairing the improvements to the premises in excess of the proceeds of insurance received by Lessor allocated to the premises shall be repayable by Lessee to Lessor within ten (10) days after receipt by Lessee from Lessor of a statement setting forth the amount of such excess, together with interest at the highest rate from the date Lessor incurred such amount to the date of payment by Lessee.

- 16.2. Lessor and Lessor's insurance carrier shall determine the amount of insurance proceeds attributable to the damage of such improvements which determination shall be binding upon Lessor and Lessee. The party required hereunder to repair the damage to the premises shall reconstruct such premises in accordance with the working drawings originally approved by Lessor. In no event shall Lessor be required to repair or replace Lessee's merchandise, trade fixtures, furnishings or equipment. If more than thirty-five percent (35%) of the Floor Area of the building in which the premises are located shall be damaged or destroyed by fire or other casualty, or if during the last three (3) years of the Term more than twenty-five percent (25%) of the premises or the Floor Area of the building in which the premises are located ("Building") shall be damaged or destroyed by fire or other casualty then Lessor may either elect that the Building and/or premises, as the case may be, be repaired or rebuilt or, at its election to so terminate. Such notice shall be given, if at all, within ninety (90) days after the occurrence of such damage or destruction. If Lessor is required or elects to repair or rebuild or requires the Lessee to repair or rebuild the premises as herein provided, Lessee shall repair or replace its merchandise, trade fixtures, furnishings and equipment in a manner and to at least a condition equal to that prior to its damage or destruction...
- 16.3. If the premises cannot be restored within 90 days of the loss, or the loss occurs during the last 3 months of the lease, this lease can be terminated by either party. During such 90 day period Lessee shall not be required to pay monthly lease/ rent to the Lessor.

Initialed

17. EMINENT DOMAIN

17.1. TOTAL CONDEMNATION.

If the whole of the leased premises shall be acquired or condemned by eminent domain for any public or quasi-public use or purpose, then the term of this lease shall cease and terminate as of the date of title vesting in such proceeding and all rentals shall be paid up to that date and Lessee shall have no claim against Lessor for the value of any unexpired term of this lease.

17.2. TOTAL PARKING AREA.

If the whole of the common parking areas in the Business Park shall be acquired or condemned by eminent domain for any public or quasi-public use or purpose, then the term of this lease shall cease and terminate as of the date of title vesting in such proceeding unless Lessor shall take immediate steps to provide other parking facilities substantially equal to the previously existing ratio between the common parking areas and the leased premises, and such substantially equal parking facilities shall be provided by Lessor at its own expense within ninety (90) days from the date of acquisition. In the event that Lessor shall provide such other substantially equal parking facilities, then this lease shall continue in full force and effect. In any event, Lessee shall have no claim against Lessor for the value of any unexpired term of this lease.

17.3. PARTIAL CONDEMNATION.

If any part of the leased premises shall be acquired or condemned by eminent domain for any public or quasi-public use or purpose, and in the event that such partial taking or condemnation shall render the leased premises unsuitable for the business of the Lessee, then the term of this lease shall cease and terminate as of the date of title vesting in such proceeding and Lessee shall have no claim against Lessor for the value of any unexpired term of this lease and the Lessee shall not have any payment obligations for the remainder term of the lease from the date of such Partial Condemnation. In the event of a partial taking or condemnation which is not extensive enough and is mutually agreed upon as suitable for the business of the Lessee, then Lessor shall promptly restore the leased premises to a condition comparable to its condition at the time of such condemnation less the portion lost in the taking, and this lease shall continue in full force and effect.

17.4. LESSOR'S DAMAGES.

In the event of any condemnation or taking as hereinabove provided, whether whole or partial, the Lessee shall not be entitled to any part of the award, as damages or otherwise, for such condemnation and Lessor is to receive the full amount of such award, the Lessee hereby expressly waiving any right or claim to any part thereof.

17.5. LESSEE'S DAMAGES.

Although all damages in the event of any condemnation are to belong to the Lessor whether such damages are awarded as compensation for diminution in value of the leasehold or to the fee of the leased premises, Lessee shall have the right to claim and recover from the condemning authority, but not from Lessor, such compensation as may be separately awarded or recoverable by Lessee in Lessee's own right on account of any and all damage to Lessee's business by reason of the condemnation and for or on account of any cost or loss to which Lessee might be put in removing Lessee's merchandise, furniture, fixtures, leasehold improvements and equipment.

18. DEFAULT OF LESSEE

18.1. RIGHT TO RE-ENTER.

In the event of any failure of Lessee to pay any rental due hereunder within ten (10) days after the same shall be due, or any failure to perform any other of the terms, conditions or covenants of this lease to be observed or performed by Lessee for more than THREE (3) days after written notice of such default shall have been received by Lessee, or if Lessee shall become bankrupt or insolvent, or file any debtor proceedings, or take or have taken against Lessee in any court pursuant to any statute either of the United States or of any State a petition in bankruptcy or insolvency or for reorganization or for the appointment of a receiver or trustee of all or a portion of Lessee's property, or if Lessee makes an assignment for the benefit of creditors, or petitions for or enters into an arrangement, or if Lessee shall abandon said premises, or suffer this lease to be taken under any writ of execution, then Lessor besides other rights

Initialed

or remedies it may have, shall have the immediate right of re-entry and may remove all persons and property from the leased premises and such property may be removed and stored in a public warehouse or elsewhere at the cost of, and for the account of Lessee, all without service of notice or resort to legal process and without being deemed guilty of trespass, or becoming liable for any loss or damage which may be occasioned thereby.

18.2. RIGHT TO RELET.

Pursuant to 18.1, Should Lessor elect to re-enter, as above provided, or should it take possession pursuant to legal proceedings or pursuant to any notice provided for by law, it may either terminate this lease or it may from time to time without terminating this lease, make such reasonable alterations and repairs as may be necessary in order to relet the premises, and relet said premises or any part thereof for such term or terms (which may be for a term extending beyond the term of this lease) and at such rental or rentals and upon such other terms and conditions as Lessor in its sole discretion may deem advisable; upon each such reletting all rentals received by the Lessor from such reletting shall be applied, first, to the payment of any indebtedness other than rent due hereunder from Lessee to Lessor; second, to the payment of any costs and expenses of such reletting, including brokerage fees and attorney's fees and of costs of such alterations and repairs; third, to the payment of rent due and unpaid hereunder, and the residue, if any, shall be held by Lessor and applied in payment of future rent as the same may become due and payable hereunder. If such rentals received from such reletting during any month be less than that to be paid during that month by Lessee hereunder, Lessee shall pay any such deficiency to Lessor. Such deficiency shall be calculated and paid monthly. No such re-entry or taking possession of said premises by Lessor shall be construed as an election on its part to terminate this lease unless a written notice of such intention is given to Lessee or unless the termination thereof is decreed by a court of competent jurisdiction. Notwithstanding any such reletting without termination, Lessor may at any time thereafter elect to terminate this lease for such previous breach. Should Lessor at any time terminate this lease for any breach, in addition to any other remedies it may have, it may recover from Lessee all damages it may incur by reason of such breach, including the cost of recovering the leased premises, reasonable attorney's fees, and including the worth at the time of such termination of the excess, if any, of the amount of rent and charges equivalent to rent reserved in this lease for the remainder of the stated term over the currently agreed rental value of the leased premises as described in section 2.3 for the remainder of the stated term, all of which amounts shall be immediately due and payable from Lessee to Lessor.

18.3. LEGAL EXPENSES.

In the event of any action or proceeding brought by either party against the other under this lease, each party shall bear their own costs and fees of its attorneys in such action or proceeding, including costs of appeal, if any.

18.4. WAIVER OF RIGHTS OF REDEMPTION.

Lessee hereby expressly waives any and all rights of redemption granted by or under any present or future laws in the event of Lessee being evicted or dispossessed for any cause, or in the event of Lessor obtaining possession of the leased premises, by reason of the violation by Lessee or any of the covenants or conditions of this lease, or otherwise.

19. ACCESS BY LESSOR

19.1. RIGHT OF ENTRY.

Given 24 hours notice, Lessor or Lessor's agents shall have the right to enter the leased premises during normal business hours and make reasonable efforts not to impact the business of the Lessee to examine the same, and to show them to prospective purchasers or lessees of the building, and to make such repairs, alterations improvements or additions as Lessor may deem necessary or desirable, and Lessor shall be allowed to take all material into and upon said premises that may be required therefore without the same constituting an eviction of Lessee in whole or in part and the rent reserved shall in no wise abate while said repairs, alterations, improvements, or additions are being made, in the event it is necessary to temporarily close the business office during such repairs. During the six months prior to the expiration of the term of this lease or any renewal term, Lessor may exhibit the premises to prospective Lessees or purchasers, and place upon the premises the usual notices "To Let" or "For Sale" which notices Lessee shall permit to remain thereon without molestation. Lessor will take all appropriate measures to attempt to make repairs during times that does not impede on the Lessee's Business.

Initialed

20. LESSEE'S PROPERTY

20.1. TAXES ON LEASEHOLD.

Lessee shall be responsible for and shall pay before delinguency all municipal, county or state taxes assessed during the term of this lease against any leasehold interest or lessee's personal property of any kind, owned by or placed in, upon or about the leased premises by the Lessee.

20.2. LOSS AND DAMAGE.

Lessor shall not be liable for any damage to property of Lessee or of others located on the leased premises, nor for the loss of or damage to any property of Lessee or of others by theft or otherwise. Lessor shall not be liable for any injury or damage to persons or property resulting from fire, explosion, falling plaster, steam, gas, electricity, water, rain or leaks from any part of the leased premises or from the pipes, appliances or plumbing works or from the roof. street or sub-surface or from any other place or by dampness or by any other cause of whatsoever nature. Lessor shall not be liable for any such damage caused by other Lessees or persons in the leased premises, occupants of adjacent property, of the Business Park, or the public, or caused by operations in construction of any private, public or quasi-public work. All property of Lessee kept or stored on the leased premises shall be so kept or stored at the risk of Lessee only and Lessee shall hold Lessor harmless from any claims arising out of damage to the same, Including subrogation claims by Lessee's Insurance carriers, unless such damage shall be caused by the willful act or gross neglect of Lessor.

20.3. NOTICE BY LESSEE.

Lessee shall give immediate notice to Lessor in case of fire or accidents in the leased premises or in the building of which the premises are a part or of defects therein or in any fixtures or equipment.

21. HOLDING OVER, SUCCESSORS

21.1. HOLDING OVER.

Any holding over after the expiration of the term hereof, with the consent of the Lessor, shall be construed to be a tenancy from month to month at the rents herein specified (prorated on a monthly basis) and shall otherwise be on the terms and conditions herein specified, so far as applicable.

21.2. SUCCESSORS.

All rights and liabilities herein given to, or imposed upon, the respective parties hereto shall extend to and bind the several respective heirs, executors, administrators, successors, and assigns of the said parties; and if there shall be more than one Lessee, they shall all be bound jointly and severally by the terms, covenants and agreements herein. No rights, however, shall inure to the benefit of any assignee of Lessee unless the assignment to such assignee has been approved by Lessor in writing as provided in Section 16.1 hereof.

22. QUIET ENJOYMENT

22.1. LESSOR'S COVENANT.

Upon payment by the Lessee of the rents herein provided, and upon the observance and performance of all the covenants, terms and conditions on Lessee's part to be observed and performed, Lessee shall peaceably and quietly hold and enjoy the leased premises for the term hereby demised without hindrance or interruption by Lessor or any other person or persons lawfully or equitably claiming by, through or under the Lessor, subject, nevertheless, to the terms and conditions of this lease.

23. MISCELLANEOUS

23.1. WAIVER.

The waiver by Lessor of any breach of any term, covenant or condition herein contained shall not be deemed to be a waiver of such term, covenant or condition or any subsequent breach of the same or any other term, covenant or condition herein contained. The subsequent acceptance of rent hereunder by Lessor shall not be deemed to be a waiver of any preceding breach by Lessee of any term, covenant or condition of this lease, other than the failure of

Initialed Initialed

Lessee to pay the particular rental so accepted, regardless of Lessor's knowledge of such preceding breach at the time of acceptance of such rent. No covenant, term or condition of this lease shall be deemed to have been waived by Lessor, unless such waiver is in writing by Lessor.

23.2. ACCORD AND SATISFACTION.

No payment by Lessee or receipt by Lessor of a lesser amount than the monthly rent herein stipulated shall be deemed to be other than on account (of the earliest stipulated rent, nor shall any endorsement or statement on any check or any letter accompanying any check or payment as rent be deemed an accord and satisfaction, and Lessor may accept such check or payment without prejudice to Lessor's right to recover the balance of such rent or pursue any other remedy in this lease period.

23.3. ENTIRE AGREEMENT.

This lease and the Exhibits, and Rider, if any, attached hereto and forming a part hereof, set forth all the covenants, promises, agreements, conditions or understandings, either oral, or written, between them other than are herein set forth. Except as herein otherwise provided, no subsequent alteration, amendment, change or addition to this lease shall be binding upon Lessor or Lessee unless reduced to writing and signed by them.

23.4. NO PARTNERSHIP.

Lessor does not, in any way or for any purpose, become a partner of Lessee in the conduct of its business, or otherwise, or joint adventurer or a member of a joint enterprise with Lessee. The provisions of this lease relating to the percentage rent payable hereunder are included solely for the purpose of providing a method whereby the rent is to be measured and ascertained.

23.5. FORCE MAJEURE.

In the event that either party hereto shall be delayed or hindered in or prevented from the performance of any act required hereunder by reason of strikes, lock-outs, labor troubles, inability to procure materials, failure of power, restrictive governmental laws or regulations, riots, insurrection, war or other reason of a like nature not the fault of the party delayed in performing work or doing acts required under the terms of this lease, then performance of such act shall be excused for the period of the delay and the period for the performance of any such act shall be extended for a period equivalent to the period of such delay. The provisions of this Section 24.5 shall not operate to excuse Lessee from the prompt payment of rent, percentage rent, additional rent or any other payments required by the terms of this lease.

23.6. NOTICES.

(a) Any notice by Lessee to Lessor must be served by certified or registered mail, postage prepaid, addressed to Lessor at the address first hereinabove given or at such other address as Lessor may designate by written notice.

(b) Any notice by Lessor to Lessee must be served by Certified or registered mail, postage prepaid, addressed to Lessee at the leased premises or at such other address as Lessee shall designate by written notice.

23.7. CAPTIONS AND SECTION NUMBERS.

The captions, section numbers, article numbers, and index appearing in this lease are inserted only as a matter of convenience and in no way define, limit, construe, or describe the scope or intent of such sections or articles of this lease nor in any way affect this lease.

23.8. LESSEE DEFINED, USE OF PRONOUN.

The word "Lessee" shall be deemed and taken to mean each and every person or party mentioned as a Lessee herein, be the same one or more; and if there shall be more than one Lessee, any notice required or permitted by the terms of this lease may be given by or to any one thereof, and shall have the same force and effect as if given by or to all thereof. The use of the neuter singular pronoun to refer to Lessor or Lessee shall be deemed a proper reference even though Lessor or Lessee may be an individual, a partnership, a corporation, or a group of two or more individuals or corporations. The necessary grammatical changes required to make the provisions of this lease apply in the plural sense where there is more than one Lessor or Lessee and to either corporations, associations, partnerships, or individuals, males or females, shall in all instances be assumed as though in each case fully expressed.

23.9. PARTIAL INVALIDITY, If any term, covenant or condition of this lease or the application thereof to any person or circumstance shall, to any extent, be invalid or unenforceable, the remainder of this lease, or the application of such

Initialed Initialed

term, covenant or condition to persons or circumstances other than those as to which it is held invalid or unenforceable, shall not be affected thereby and each term, covenant or condition of this lease shall be valid and be enforced to the fullest extent permitted by law.

24. RECORDING

24.1. Lessee shall not record this lease without the written consent of Lessor.

25. OPTION

- 25.1. If all terms and conditions have been carried out successfully, Lessor hereby grants Lessee the right, option and privilege of extending and renewing the term of the Lease for One (1) additional period(s) of Two (2) year(s) beginning on the termination date of the lease. Lessee shall notify Lessor no later than 90 days prior to expiration of this lease to inform Lessor of Lessee's intent to exercise the option. If notice is not received within the time specified, it will be assumed that Tenant will not exercise the option and the option shall be forfeited.
- 25.2. If assignee timely exercises the option to renew the term, the term shall thereafter automatically be extended and renewed for an additional two (2) year(s), which the extended and renewed term shall be on all of the same covenants, terms, and conditions as are contained in the Lease, except for the Base Rent which at the beginning of the first full calendar month in the option term, and if applicable, at one year intervals thereafter, the monthly rental shall increase, but not decrease, in proportion to any percentage increase in the Consumer Price Index as published by the Bureau of Labor Statistics (U.S. Cities Average for all Urban Consumers) between (i) the amount of that Index as last published prior to the beginning of the preceding 12-month period, and (ii) the amount of that Index as last published prior to the date on which the rental increase is to occur. In any event, the rent for any lease year shall be increased by not less than two (2%) per annum and shall not be increased by more than four (4%) per annum. In no event shall the monthly rental, as adjusted hereunder, be less than the applicable minimum rental set forth in section 2.01 above.
- 25.3. In the event that Lessee and Lessor cannot arrive at an agreement within 30 days after the expiration of this lease, all terms shall be considered on a month to month tenancy and the option shall be forfeited.

Daniel Fries Lessor, Lessor's Agent Fries Properties Inc. 39678 Mission Blvd. Fremont, Ca 94539

27 Date

Chris Floethe

Lessee Construction Craft Training 26200 Industrial Blvd Hayward, CA 94545

MA47 1,2014 Date

None

Date

Co-Lessee

Initialed Initialed

APPENDIX IV TO DONATION AGREEMENT

Assignment of Leases

See attached.

ASSIGNMENT OF REAL PROPERTY LEASE

This ASSIGNMENT OF REAL PROPERTY LEASE (the "<u>Assignment</u>") is made as of _________, 2016 by and between Chris Floethe dba Construction Craft Training ("<u>Assignor</u>") and Construction Craft Training Center, a California nonprofit public benefit corporation ("<u>Assignee</u>"). Assignor and Assignee are referred to collectively as the "<u>Parties</u>".

RECITALS

- A. Assignor and Hayward Business Park Inc. ("<u>Lessor</u>") entered into that certain Hayward Business Park Lease Agreement commenced June 1, 2014, with respect to the premises located at 26200 Industrial Blvd., Hayward, California 94545 (the "<u>26200 Lease</u>").
- B. Assignor and Lessor entered into that certain Hayward Business Park Lease Agreement commenced June 1, 2014, with respect to the premises located at 26206 Industrial Blvd., Hayward, California 94545 (the "<u>26206 Lease</u>" and together with the 26200 Lease, the "<u>Leases</u>").
- C. Assignee intends to transfer the operation of its existing Electrician Training Program to Eden Area Regional Occupational Program, a California joint venture of Castro Valley, Hayward, San Leandro, and San Lorenzo Unified School Districts ("<u>Eden Area ROP</u>") by making a contribution to Eden Area ROP of all of its assets, including any rights under real property leases for the premises at which Assignee operates the Electrician Training Program, pursuant to the terms of a Donation Agreement (the "<u>Donation</u>").
- D. In connection with the Donation, Assignor and Assignee mutually desire (a) that Assignor assign all of his right, title, and interest in, under, and to the Leases to Assignee and (b) that Assignee assume all of Assignor's obligations under the Leases.

AGREEMENT

In consideration of the agreements contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties agree as follows:

1. <u>Effective Date</u>.

For all purposes under this Assignment, the term "<u>Effective Date</u>" shall mean the effective date of the Donation.

2. <u>Assignment and Assumption</u>.

(a) Effective as of the Effective Date, Assignor hereby assigns, transfers, and sets over unto Assignee all of Assignor's right, title, and interest in, under, and to the Leases.

(b) Assignee hereby accepts the foregoing assignment and assumes and agrees to perform when due the obligations of Assignor pursuant to the Leases in accordance with the terms thereof.

3. <u>Governing Law.</u>

This Assignment shall be governed by and construed in accordance with the laws of the State of California.

4. <u>Counterparts.</u>

This Assignment may be executed in any number of counterparts, each of which shall be an original, but all of which together shall constitute one and the same instrument. Any counterpart may be delivered by facsimile or electronic transmission, each of which shall be deemed an original.

5. <u>Successors and Assigns; Amendments</u>.

This Assignment shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns. No amendment of any provision of this Assignment shall be valid unless the same shall be in writing and signed by Assignor and Assignee.

- Signature Page Follows This Page -

IN WITNESS WHEREOF, the parties hereto have executed this Assignment of Real Property Lease as of the date first written above.

ASSIGNOR:

ASSIGNEE:

Construction Craft Training Center

Chris Floethe dba Construction Craft Training By:

Chris Floethe, Executive Director

August ___, 2016

L & M Investments P.O. Box 766 Turlock, CA 95381

Re: Request for Consent to Assignment of Lease

To Whom It May Concern:

Reference is made to that certain Standard Industrial/Commercial Multi-Tenant Lease - Gross commenced April 1, 2001 (the "Lease") by and between Construction Craft Training Center, a California nonprofit public benefit corporation ("<u>CCTC</u>"), as tenant, and L & M Investments ("<u>L&M</u>"), as landlord, with respect to the premises located at 2480 Acme Ct., Building B (north half), Turlock, California 95380 (the "<u>Premises</u>"). CCTC is providing this letter to request L&M's consent to an assignment of the Lease as described further below.

CCTC is currently planning a contribution of all of its assets to Eden Area Regional Occupational Program, a California joint venture of Castro Valley, Hayward, San Leandro, and San Lorenzo Unified School Districts ("<u>Eden Area ROP</u>") net the assumption of certain liabilities of CCTC by Eden Area ROP.

The purpose of this letter, therefore, is to advise L&M of the pending contribution and assignment, and, to the extent required under Section 12 of the Lease, obtain L&M's consent to such assignment of the Lease.

By executing this letter as provided below, L&M hereby:

- 1. consents to and approves the assignment for all purposes of the Lease;
- 2. acknowledges that the Lease is in full force and effect and is not aware of any existing default under the Lease or of any event or circumstance that with the giving of notice or passage of time, or both, could result in a default under the Lease;
- 3. to L&M's knowledge, has neither given notice to CCTC nor received notice from CCTC of a currently existing dispute regarding the Lease;
- 4. confirms that CCTC has paid all rent and other amounts due and owing as of the date of this letter; and
- 5. confirms that attached hereto is a true, correct, and complete copy of the Lease and that the Lease has not been amended or modified, except as attached hereto.

If L&M is agreeable to the contents of this letter, please execute this letter where indicated below and email an executed copy to the undersigned at chris@cctc.edu. Please also return one executed original by mail to the following address: 26200 Industrial Blvd., Hayward, CA 94545.

If you have any questions, please feel free to contact me at 510-506-1532.

Sincerely,

Chris Floethe Executive Director Construction Craft Training Center

Acknowledged and Agreed:

L & M Investments

By:_____

Name:_____

Title:

Date: _____

August ___, 2016

Fries Properties Inc. 39678 Mission Blvd. Fremont, CA 94539

Re: Request for Consent to Assignment of Leases

To Whom It May Concern:

Reference is made to (i) that certain Hayward Business Park Lease Agreement commenced June 1, 2014 by and between Chris Floethe dba Construction Craft Training ("Floethe"), as tenant, and Lessor Business Park Inc. ("Lessor"), as landlord, with respect to the premises located at 26200 Industrial Blvd., Lessor, California 94545 (the "26200 Lease"), and (ii) that certain Lessor Business Park Lease Agreement commenced June 1, 2014 by and between Floethe, as tenant, and Lessor, as landlord, with respect to the premises located at 26206 Industrial Blvd., Lessor, California 94545 (the "26206 Lease" and together with the 26200 Lease, the "Leases"). Floethe is providing this letter to request Lessor's consent to an assignment of the Leases as described further below.

Construction Craft Training Center, a California nonprofit public benefit corporation ("<u>CCTC</u>"), is currently planning a contribution of all of its assets to Eden Area Regional Occupational Program, a California joint venture of Castro Valley, Hayward, San Leandro, and San Lorenzo Unified School Districts ("<u>Eden Area ROP</u>") net the assumption of certain liabilities of CCTC by Eden Area ROP. In conjunction with such contribution, Floethe will assign his rights under the Leases to CCTC, which will in turn assign such rights to Eden Area ROP.

The purpose of this letter, therefore, is to advise Lessor of the pending contribution and assignment, and, to the extent required under Section 14 of the Leases, obtain Lessor's consent to such assignment of the Leases.

By executing this letter as provided below, Lessor hereby:

- 1. consents to and approves the assignment for all purposes of both the 26200 Lease and the 26206 Lease;
- 2. acknowledges that each of the 26200 Lease and 26206 Lease is in full force and effect and that Lessor is not aware of any existing default under either the 26200 Lease or the 26206 Lease or of any event or circumstance that with the giving of notice or passage of time, or both, could result in a default under either the 26200 Lease or the 26206 Lease;
- 3. to Lessor's knowledge, has neither given notice to Floethe nor received notice from Floethe of a currently existing dispute regarding either the 26200 Lease or the 26206 Lease;

- 4. confirms that Floethe has paid all rent and other amounts due and owing under the Leases as of the date of this letter; and
- 5. confirms that attached hereto is a true, correct, and complete copy of the Leases and that neither the 26200 Lease nor the 26206 Lease has been amended or modified, except as attached hereto.

If Lessor is agreeable to the contents of this letter, please execute this letter where indicated below and email an executed copy to the undersigned at chris@cctc.edu. Please also return one executed original by mail to the following address: 26200 Industrial Blvd., Hayward, CA 94545.

If you have any questions, please feel free to contact me at 510-506-1532.

Sincerely,

Chris Floethe

Acknowledged and Agreed:

Lessor Business Park Inc.

By:_____

Title: _____

Date: _____

APPENDIX V TO DONATION AGREEMENT

Turlock Lease

See attached.

Addendum "A" to lease Agreement Dated February 23, 2001

By and between:

(Lessor) (Lessee) L & M Investments Construction Craft Training Ctr.

Property Address:

2480 Acme Court Turlock, CA

Lessee agrees to extend the afore-mentioned lease for an additional year from October 1, 2015 through September 30, 2016 with 5% rent increase to \$1,405.69/month.

All terms and conditions of the afore-mentioned lease will apply to this lease addendum as well.

Lessor and Lessee have hereunto subscribed their names below in agreement with the above terms.

L & M Investments

Construction Craft Training Ctr.

7-28-1 Date

/20/S

,

. - -



1. Basic 1.1	Provisions (Parties:	("Basic Provi This Lease ("	isions"). "Lease"), dated fo	r reference purpo	ses only	Februar	y 23, 2001	, is made by and
between <u>L & I</u>	i Invest	ments						
								("Lessor")
			Indan Grata					(
and Constru	etion Ci	ait Tra	ining Cente	<u>r</u>			<u>, , , , , , , , , , , , , , , , , , , </u>	
						("Lesse"), (collec	tively the "Parties", o	r Individually a "Party").
1.2(a)	Premise	s: That cert	ain portion of the l	Project (as defin	ad below), inclu	ding all improvement	ts therein or to be pro	wided by Lessor under the
terms of this Lea	ise, commont	y known by th	e street address o	2480 P	cme Ct.,	Building B	(north half)	, located in the City of
of	<u>^</u>	ν <u>α</u> .		with zip code	95380 .	as outlined on EXNI	DK arractie	o ticiero fi Licerando haura
generally descrit	bed as (desc	xibe briefly ti	ne nature of the F	remises): One	half of	a 6000 sf k	building, dee	med to be
3000 sq. 1	Et.							
Paragraph 2.7 b ("Building") or buildings and im 1.2(b)	elow) as her to any other provements to Parking	einsiter speci buildings in f hereon, are fa : Common	fled, but shail not the Project. The F amin collisctively re area U	have any rights remises, the Bu ferred to as the " nreserved vehicl	to the roor, ex liding, the Com Project." (See e parking space	mon Areas, the land also Peredraph 2.)	upon which they are	mmon Areas (as defined in 19 containing the Premises located, along with all other 0
reserved vehicle	parking spac	es ("Reserve	d Parking Space	s'''). (See also Pi	aragraph 2.6.)			
1.3	Term:	<u>3</u> y	ears and	months ("O	riginal Term")	commencing <u>Apri</u>	11 1, 2001	L 4 \
("Commencern	ent Date") ar	nd ending	March	31, 2004		("Expiration Date"	"). (See also Paragrap	n o.) Demographic 3.2 and 3.3 \
1.4	Early Po	ossession:	by	agreement		_ ("Early Possesal	Sin Date"). (See 660) firet	Paragraphs 3.2 and 3.3.) day of each month
1.5	Base R	ant: \$	1,110	per mont	A ("Base Kom	r), payable on sie _	17790	day of each month
commancing <u>2</u>				Mar Barber Plank	المعقمة بالأسحر بمبتا المة			
□) if this box is 1.6	checked, the	re are provision in Share of C	ons in this Lease fo ommon Area Opt	rating Expense	s:		percent (%) ("Lesses's Share").
1.0	Base R	ant and Othe	r Monies Paid Lin	on Execution:				
	(a)					<u>:11 1, 2001 -</u>	- April 30, 2	001
	(b)	Common	Area Operating E	xpenses: \$		for the period	·	·
	(¢)	Security D	leposit: \$	1,110	_ ("Security D	leposit"). (See also	Paragraph 5.)	· · · · · · · · · · · · · · · · · · ·
	(d)					.00		······································
	(e)						·	
1,8	Agreed	Use: Cons	truction Tr	aining ce	ILEL		·	
		-1			<u> </u>			
								(See also Paragraph 6.)
1.9 1.10			sor is the "Insurin :: (See also Parag The following rea			8.)		ransaction (check applicable
boxes):					re	presents Lessor exci	lusively ("Lessor's Bro	oker");
					re	presenta Lessee exc	lusively ("Lessee's Br	oker"); or
Cold	well Ba	nker Com	mercial End	Halev & As	SD. (8	oresents both Lesso	r and liessee ("Duar A	gency).
	(b) Paul	ment to Stol	ers: Upon execut	ion and delivery o	of this Lease by	both Parties, Lessor	shall pay to the Broke	5
the brokenage f Base Rent for	the brokerage	services ran	dered by the Broke	ns).				_ ~
1.11	Guarar	tor. The obl	igations of the Les	see under this Le	ase are to be g	uaranteed by	and the second s	a size Descents (27.)
								e also Paragraph 37.) through 51 and
1.12 Exhibits	Adden throu			eto is an Addeni Institute a part o		" crystating of Para	graphs <u>50</u>	unordiu <u>ot</u> mu
2.1 terms, covenant in calculating R size is more or	its and condit tent, is an ap lass.	lions set forth proximation w	in this Lease. Uni high the Parties as	ess otherwise pr rea is resconsbi	ovided herein, a e and any payn lices contained	any statement of size hents based thereon within the Bulking ("	are not subject to revis "Unit") to Lessee brook	he rental, and upon all of th , or that may have been use fon whether or not the actus m clean and free of debris o much described in Paraman
7.1(b) below a heating, vantila	ment Date or re obtained b rting and air o d operating co	the Early Pos y Lessee and conditioning s andition on sa	esession Date, whi I in effect within the ystems ("HVAC"), id date and that the	chever first occul inty days followin loading doors, it structural eleme	rs ("Start Date of the Start Date of any, and all of onts of the roof,	"), and, so long as in te, warrants that the her such elements it bearing walls and fou ne or elements shou	edsting electrical, plut in the Unit, other than t undation of the Unit shi Int mailfunction or fail w	macts described in Paragraphing, fire sprinkler, lighting hose constructed by Lesse all be free of material defect thin the appropriate warrant receipt of written notice fro

period, Lesson shall, as Lesson's sole obligation with respect to such matter, except as otherwise provided in this Lesse, promptly after receipt of written house from Lesses setting forth with specificity the nature and extent of such non-compliance matturction or failure, recitity series at Lesson's expense. The warranty periods shall be as follows: (i) 6 months as to the HVAC systems, and (ii) 30 days as to the required notice within the appropriate warranty period, correction of any such non-compliance, matfunction or failure shall be the obligation of Lessee is Lessee is sole cost and expense (except for the repeirs to the fire sprinkler systems, noof, foundations, and/or bearing wells - see Paragraph 7).

initiais

Page 1 of 12

.

07/27/2016 15:09 ACME ELECTRIC

-

(FAX)209 667 1251

P.005/019

• 2.3 Compliance sor warrants that the improvements on the Premises and the Common Av imply with the building codes that were in affect at the time that each such improvement, or portion thereof, was constructed, and also with all applicable laws, covenants or restrictions of record, regulations, and ordinances in effect on the Start Date ("Applicable Requirements"). Said verrenty does not apply to the use to which tessee will put the Premises or to any Atterations or Utility Installations (as defined in Paragraph 7.3(a).) made or to be made by Lessee. NOTE: Lessee is responsible for determining whether or not the zoning is appropriate for Lessee's Intended use, and acknowledges that past uses of the Premises may no longer be allowed. If the Premises do not the comply with adjustment of Lessee's abal, except as otherwise provided, promoting that needs to be remisses may no longar be allowed. If the remisses do not extent of such non-compliance, rectify the same at Lesso's expanse. If Lessee does not give Lessor written notice of a non-compliance with this warranty within 6 months following the Start Date, correction of that non-compliance shall be the obligation of Lessee at Lessee's cole crist and expanse. If the Applicable Requirements are hereafter changed so as to require during the term of this Lease the construction of an addition to or an attention of the Unit, Premises and/or Building, the remediation of any Hazardous Substance, or the reinforcement or other physical modification of the Unit, Premises and/or Building ("Capital Expenditure"), Lesser and Lessee shall allocate the cost of such work as follows:

(e) Subject to Paragraph 2.3(c) below, if such Capital Expenditures are required as a result of the specific and unique use of the Premises by Lessee as compared with uses by tenants in general, Lessee shall be fully responsible for the cost thereof, provided, however, that if such Capital Expenditure is required during the last 2 years of this Lesse and the cost thereof exceeds 6 months' Base Rent, Lessee may instead terminate this Lesse unless Lessor notifies required during the test 2 years of this Lease and the cost thereof exceeds 6 monthly Base Rent, Lease may instead terminate this Lease unless tessor notifies Leases, in writing, within 10 days after receipt of Leases's termination notice that Leasor has elected to pay the difference between the actual cost thereof and the amount equal to 6 months' Base Rent. If Leases elects termination cases shall immediately cases the use of the Premises which requires such Capital Expenditure and deliver to Leasor written notice specifying a termination date at least 90 days thereafter. Such termination date shall, however, in no event be earlier than the last day that Leases could legally utilize the Premises without commencing such Capital Expenditure. (b) if such Capital Expenditure is not the result of the specific and unique use of the Premises (such as, governmentally mendated seismic modifications), then Leasor and Leases the obligation to pay for the portion of such costs reasonably attributable to the Premises pursuant to the formula set out in Paragraph 7.1(d); provided, however, that if such Capital Expenditure is required during the last 2 years of this Lease or if Leasor reasonably determines that its its in the start frank the actual to eavier the specific back on the terminate the last of the premises of the specific read unique use of the Premises of the lease or of Leasor reasonably

the formula set out in Paragraph 7.1(d); provided, however, that if such Capital Expenditure is required during the fast 2 years of this Lease or if Leasor reasonably determines that it is not economically feasible to pay its share thereof, Leasor shall have the option to terminate this Lease upon 90 days prior written notice to Leasor unless Leasee notifies Leasor, in writing, within 10 days alier receipt of Leasor's termination notice that Leasee will pay for such Capital Expenditure. If Leasor that not elect to terminate, and fails to tender ha share of any such Capital Expenditure, Lessee way advance such funde and deduct same, with interest, from Rent until Lessor's share of such costs have been fully paid. If Lessee is unable to finance Lessor's share, or if the balance of the Rent due and payable for the remainder of this Lease is not sufficient to fully reimburse Lessee on an offset basis, Lessee shall have the right to terminate this Lease upon 30 days written notice to Lessor. (c) Notwithstanding the above, the provisions concerning Capital Expenditures intended to apply only to non-voluntary, unexpecied, and new Applicable Requirements. If the Capital Expenditures are instead triggered by Lessee as a result of an actual or proposed change in use, change in intensity of use, or modification to the Premises then, and in that event, Lessee shall be fully responsible for the cost thereof, and Lessee shall not have any right to terminate this Lease.

2.4 Acknowledgements. Lessae acknowledges that: (a) It has been advised by Lessor and/or Brokers to satisfy itself with respect to the condition of the Premises (including but not limited to the electrical, HVAC and the sprinkler systems, security, environmental aspects, and compliance with Applicable Requirements and the Americans with Disabilities Act), and their suitability for Lessee's intended use, (b) Lessee has made such investigation as it Acknowledgements. L deems necessary with reference to such metters and assumes all responsibility therefor as the same relate to its occupancy of the Premises, and (o) neither Lessor, Lessor's agents, nor Brokers have made any oral or written representations or warranties with respect to said matters other than as set forth in this Lease. In

Lessor's agents, nor Brokers have made any oral or written representations or warranties with respect to said matters other than as set forth in this Lesse. In addition, Lessor acknowledges that: (i) Brokers have made no representations, promises or warranties concerning Lesse's ability to honor the Lesse or suitability to occupy the Premises, and (ii) it is Lessor's sole responsibility to invastigate the financial capability and/or suitability of all proposed tenants. 2.5 Lesses as Prior Owner/Occupant. The warranties made by Lessor in Paragraph 2 shall be of no force or effect if immediately prior to the Start Date Lessee was the owner or occupant of the Premises. In such event, Lessee shall be responsible for any necessary corrective work. 2.6 Vehicle Parking. Lessee shall be efficient to use the number of Unreserved Parking Spaces and Reserved Parking Spaces shall be used for parking by vehicles no larger than full-size passenger automobiles or pick-up trucks, herain called "Permitted Size Vehicles," Lessee shall be parked in the Common Areas designated from written promises of a sport solution of the Common Areas designated for written to time by Lessor for parking. Lessee shall be added and unicating of vehicles by adopting Rules and Regulations as provided in Paragraph 2.9. No vehicles other than Permitted Size Vehicles may be parked in the Common Areas designated for written premission of Lessor. (a) Lessee shall not permit or allow any vehicles that belong to or are controlled by Lesser for such activities. (b) Lessee shall not service or store any vehicles that belong to or are controlled by Lesser for such activities. (c) If Lessee are shall be as other vehicles in the Common Areas.

(c) If Lessee permits or allows any of the prohibited activities described in this Paragraph 2.6, then Lessor shall have the right, without notice, in addition to such other rights and remedias that it may have, to remove or tow away the vehicle involved and charge the cost to Lessee, which cost shall be tely payable upon demand by Lessor.

immediately payable upon demand by Lessor. 2.7 Common Areas - Definition. The term "Common Areas" is defined as all areas and facilities outside the Premises and within the exterior boundary line of the Project and Interior utility raceways and installations within the Unit that are provided and designated by the Lessor from time to time for the general non-exclusive use of Lessor, Lessee and other tenants of the Project and their respective employees, suppliers, shippers, customers, contractors and invitees, including parking areas, loading and unloading areas, trash areas, roadways, walkways, driveways and landscaped areas. 2.8 Common Areas - Lessee"s Rights. Lessor grants to Lessee, for the benefit of Lessee and its employees, suppliers, shippers, contractors, customers and invitees, during the term of this Lesse, the non-exclusive right to use, in common with others entitled to such use, the Common Areas as they exist from time to time, subject to any rights, powers, and privileges reserved by Lessor under the terms hereof or under the terms of any rules and regulations or restrictions entities the set of the subject to any rights, powers, and privileges reserved by Lessor under the terms hereof or under the terms of any rules and regulations or participations of the use of the subject to any rights, powers, and privileges reserved by Lessor under the terms hereof or under the terms of any rules and regulations or participations of the use of the subject to any rights, powers, and privileges reserved by Lessor under the terms hereof or under the dependent instructions the dependent of the restrictions the restrictions that and the terms hereof or under the dependent of instructions the restrictions of the restrictions the rule to the restrictions the rule of the final dept to the terms hereof or under the dependent of instructions the rule of t

non area to arrise, subject to any rights, powers, and privates reserved by Lesson under the terms hereor of under the terms or any rules that regulations or restrictions governing the use of the Project. Under no circumstances shall the right herein granted to use the Common Areas be deemed to include the right to store any property, temporarity or permanently, in the Common Areas. Any such slorage shall be permitted only by the prior written consent of Lessor or Lessor's designated agant, which consent may be revoked at any time. In the event that any unauthorized storage shall occur, then Lessor shall have the right, without notice, in addition to such other rights and remedies that it may have, to remove the property and charge the cost to Lessee, which cost shell be immediately payable upon demand by Lessor.

2.9 Common Areas - Rules and Regulations. Lessor or such other person(s) as Lessor may appoint shall have the exclusive control and management of the Common Areas and shall have the right, from time to time, to establish, modify, amend and enforce reasonable rules and regulations ("Rules and Regulations") for the management, safely, care, and cleanliness of the grounds, the parking and unloading of vehicles and the preservation of good order, as well as for the convenience of other occupants or terrants of the Building and the Project and their invites. Lessee agrees to able by and conform to all such Rules well as for the convenience of other occupants or temants of the Building and the Project and their invitees. Lesses agrees to ablde by and conform to all such Rules and Regulations, and to cause its employees, suppliers, shippers, customers, contractors and invitees to as abide and conform. Lesser of the non-compliance will easily and Regulations by other temants of the Project.
 2.10 Common Areas - Changes, Lessor shall have the right, in Lessor's cole discretion, from time to time:

 (a) To make changes to the Common Areas, including, without limitation, changes in the location, size, shape and number of driveways, entrances, parking areas, loading and unloading areas, including, without limitation, changes in the location, size, shape and number of driveways, entrances, parking areas, loading and unloading areas, including, without limitation, changes in the location, size, shape and number of driveways, entrances, parking areas, loading and unloading areas, including, without limitation, changes in the location, size, shape and number of driveways, entrances, (b) To close temporarily any of the Common Areas for maintenance purposes so long at reasonable access to the Premises remains available;
 (c) To designate other land outside the boundaries of the Project be a part of the Common Areas;
 (d) To designate other land outside the boundaries to the Oreanen Areas;

(b) (c) (d) (e)

To add additional buildings and improvements to the Common Areas; To use the Common Areas while engaged in making additional improvements, repairs or alterations to the Project, or any portion thereof; and To do and perform such other acts and make such other changes in, to or with respect to the Common Areas and Project as Lessor may, in the (ff) ss judgment, deem to be appropriate Vercise of sound I

З.

initiaia

Term.

3.1 Term. The Commencement Date, Expiration Date and Original Term of this Lease are as specified in Paragraph 1.3. 3.2 Early Possession. If Lesse totally or partially occupies the Premises prior to the Commencement Date, the obligation to pay Base Rent shall be abated for the period of such early possession. All other terms of this Lease (including but not limited to the obligations to pay Lesses's Share of Common Area Operating Expenses, Real Property Taxes and insurance premiums and to maintain the Premises) shall, however, be in effect during such period. Any such early on shall not affect the Expiration Date.

Delay in Possession. Lessor agrees to use its best commercially reasonable efforts to deliver possession of the Premises to Last 3.3 Commencement Date. If, despite said efforts, Lessor is unable to deliver possession as agread, Lessor shall not be subject to sny liability therefor, nor shall such failure affect the validity of this Lesse. Lesses shall not, however, be obligated to pay Rent or perform its other obligations until it receives possession of the Premises. If possession is not delivered within 60 days after the commancement Date, Lessee may, at its option, by notice in writing within 10 days after the end of Premates. In possession is not benefated within to dark sittle fate Commandament back, Lasseer may, at its option, by house in which within to dark sittle fate commandament back, Lasseer may, at its option, by house in which we to toolvid by Lasser within solid 50 by period, caused this Lesse, in which event the Paties shall be discharged from all obligations hereunder. If such within nolice is not toolvid by Lasser within solid 10 day period, Lassee's right to cancel shall terminate. Except as otherwise provided, if possession is not tendered to Lessee by the Start Date and Lessee does not terminate this Lesse, as aforecald, any period of rent abatement that Lessee would otherwise have enjoyed shall run from the date of delivery of possession and continue for a pariod equal to what Lessee have alloyed under the terms hereof, but minus any advance date of delivery of possessions of Lessee. If possession of the Premises is not delivered within 4 months after the Commencement Date, this Lease shall terminate unless other

omissions of Lessee. If possession of the Premises is not device of many transformation of the Premises to Lessee until Lessee complies with its obligation to 3.4 Lessee Compliance. Lessor shall not be required to tender possession of the Premises to Lessee until Lessee complies with its obligation to provide evidence of insurance (Paragraph 8.5). Pending delivery of such evidence, Lessee shall be required to perform all of its obligations under this Lesse from and after the Start Date, including the payment of Rent, notwithstanding Lessor's election to withhold possession pending receipt of such evidence of insurance.

(vi)

. '

P.006/019

Further, if beases is required to perform any other conditions prior to or concurrent with the Start Date, the Start Date occur but Lessor may elect to withhold possession until such conditions are satisfied.

4 12mmt Rent Defined. All monetary obligations of Lessee to Lessor under the terms of this Lease (except for the Security Deposit) are deemed to be 4.1. rent ("Rent").

4.2 Common Area Operating Expenses. Lessee shall pay to Lessor during the term hereof, in addition to the Base Rent, Lesses's Share (as ad in Paragraph 1.8.) of all Common Area Operating Expenses, as hereinafter defined, during each calendar year of the term of this Lesse, in accordance with the following provisions:

"Common Area Operating Expenses" are defined, for purposes of this Lease, as all costs incurred by Lessor relating to the ownership and (a) operation of the Project, including, but not limited to, the follows

ig, put not imited to, the tokowing: The operation, repair and maintenance, in neat, clean, good order and condition, but not the replacement (see subparagraph (s)), of (î) the following:

The Common Areas and Common Area improvements, including parking areas, loading and unloading areas, trach areas, ys, landscaped areas, bumpers, irrigation systems, Common Area lighting facilities, fences and gates, elevators, roofs, and (100) roadways, parkways, walko roof drainage systems.

- Exterior signs and any tenant directories. (bb)
- Any fire sprinkler systems. The cost of water, gas, electricity and telephone to service the Common Areas and any utilities not separately metered. (00)
- sposal, pest control services, property management, security services, and the costs of any environmental inspections. Reserves set askie (or maintenance and repair of Common Areas. (II) (III) Tresh dk (₩) (₩) Any increase above the Base Real Property Taxies (as defined in Paragraph 10). Any "incurance Cost increase" (as defined in Paragraph 8).

Any deductible portion of an insured loss concerning the Building or the Common Areas. The cost of any Capital Expenditure to the Building or the Project not covered under the provisions of Paragraph 2.3 or shall allocate the cost of any such Capital Expenditure over a 12 year period and Lessee shall not be required to pay more than (vii) (viii) provided; however, that Le

Lesses's Share of 1/144th of the cost of such Capital Expenditure in any given month. Any other services to be provided by Lessor that are stated elsewhere in this Lesse to be a Common Area Operating (bc)

(b) Any Common Area Operating Expanses and Real Property Taxes that are specifically attributable to the Unit, the Building or to any other building in the Project or to the operation, repair and maintenance thereor, shall be afforable to the Building or other building. However, any Common Area Operating Expanses and Real Property Taxes that are not specifically attributable to the Building or other building. However, any Common Area Operating Expanses and Real Property Taxes that are not specifically attributable to the Building or to any other building. However, any Common Area Operating Expanses and Real Property Taxes that are not specifically attributable to the Building or to any other building or to the operation, repair and maintenance thereof, shall be equitably allocated by Lessor to all buildings in the Project.

(c) The inclusion of the improvements, facilities and services set forth in Subparagraph 4.2(a) shall not be deemed to impose an obligation upon (c) The inclusion of the improvements, facilities and services set forth in Subparagraph 4.2(a) shall not be deemed to impose an obligation upon or to either have said improvements of facilities or to provide those services unless the Project stready has the same, Lessor stready provides the services, or

Lessor to either have said improvements or facilities or to provide those services unless the Project already has the same, Lessor already provides the services, or Lessor has agreed elsewhere in this Lesse to provide the same or some of them. (d) Lesse's Share of Common Area Operating Expenses shall be payable by Lessee within 10 days after a reasonably detailed statement of actual expenses is presented to Lessee. At Lessor's option, however, an amount may be estimated by Lesser from time to time of Lessee's Share of annual Common Area Operating Expenses and the same shall be payable monthly or quarterly, as Lessor's hall designate, during each 12 month period of the Lesse term, on the same day Operating Expenses and the same shall be payable monthly or quarterly, as Lessor's hall designate, during each 12 month period of the Lesse term, on the same day Operating Expenses and the same shall be payable monthly or quarterly, as Lessor's hall designate, during each 12 month period of the Lesse term, on the same day operating Expenses and the same shall be payable monthly or quarterly, as Lessor's hall designate, during each 12 month period of the Lesse term, on the same day operating Expenses and the same shall be preceding year. If Lessee's payments under this Pangraph 4.2(d) during the preceding year exceed Lesses's Share of Dessiting Expenses inclured during the preceding year exceed Lesses's Share is indicated on such statement, Lessor shall be credited the amount of such over-payment against Lessee's Share of as indicated on such statement, Lesse within 10 days after delivery by Lessor to Lessee's Share is fundaese's Share of the cases shall pay to Lesser's the amount of the deficiency within 10 days after delivery by Lessor to Lessee of the statement. (c) When a capital component such as the parking lot paving, elevatore, fonced, etc. requires replacement, rether than repair or maintenance, Lessor shall, at Lessor's expense, be responsible for such replacement. Such represense, as radio costs are not Commo

expenses and/or costs are not Common Area Operating Expenses. 4.3 Payment. Lessee shall cause payment of Rent to be received by Lessor in lawful money of the United States, without offset or deduction (except as specifically permitted in this Lesse), on or before the day on which it is due. Rent for any period during the term hereof which is for less than one full (except as specifically permitted in this Lesse), on or before the day on which it is due. (except as approximately permitted in units Lease), on or before the day on which it is due. Kent for any period during the term hereof which is for lease than one full calendar month shall be prorated based upon the actual number of days of said month. Payment of Rent shall be made to Leasor at its address stated herein or to such other persons or piece as Leasor may from time to time designate in writing. Acceptance of a payment which is less than the amount then due shall not be a waiver of Leasor's rights to the balance of such Rent, regardless of Leasor's endorsement of any check so stating. In the event that any check, draft, or other instrument of payment given by Lessee to Leasor is dishonored for any reason, Lessee agrees to pay to Lessor the sum of \$25.

5. Security Deposit. Lessee shall deposit with Lessor upon execution hereof the Security Deposit as security for Lessee's faithful performance of its obligations under this Lesse. If Lessee fails to pay Rent, or otherwise Defaults under this Lesse, Lessor may use, apply or retain all or any portion of said Security Deposit for the payment of any amount due Lessor or to reimburse or compensate Lessor for any fability, expense, loss or damage which Lessor may suffer or incur Deposit for the payment of any amount due Lessor or to reimburse or compensate Lessor for any fability, expense, loss or damage which Lessor may suffer or incur Deposit for the payment of any amount due Lessor or to reimburse or compensate Lessor for any fability, expense, loss or damage which Lessor may suffer or incur Deposit. If Lessor uses or applese all or any portion of the Security Deposit, Lesses as hall within 10 days after written request therefor deposit monles with Lessor so that the table amount of the Security Deposit at all times bear the same proportion to upon written request from Lessor, deposit additional monies with Lessor shall have the right to increase due see a manifed to accommodate a sublessee or assignee, Lessor shall have the right to increase the Security Deposit to the submit necessary, in Lessor's builting and the request that the Premises may suffer as a result interefor. If a change in other deposit monies with Lessor shall have the right to increase the Security Deposit to the submit and all the security Deposit. Lesser and Security Deposit to the security Deposit, to account for any increased wear and tear that the Premises may suffer as a result interefor. If a change in control of Lessee is, in Lessor's suffy any increase the security Deposit to be at a commercially reasorable isseif based on such change in financial additional monies with Lessor as shall be sufficient to cause the Security Deposit to be at a commercially reasorable isseif does and on the sease of the security Deposit or termination of

8. Use. 5.1 Use. Lesses shall use and occupy the Premises only for the Agreed Use, or any other legal use which is reasonably comparable thereto, and for no other purpose. Lesses shall not use or permit the use of the Premises in a manner that is uniawful, creates demage, waste or a nuisance, or that disturbe occupants of or causes damage to neighboring pramises or properties. Lessor shall not unreasonably withhold or delay its consent to any written request for a modification of the Agreed Use, so long as the same will not impair the structural integrity of the improvements on the Premises or the mechanical or electrical systems therein, and/or is not significantly more burdensome to the Premises. If Lessor's elects to withhold consent, Lessor shall within 7 days after such request give written notification of same, which notice shall include an explanation of Lessor's objections to the change in the Agreed Use.

give written notification or same, which notice shall include an explanation of Lesson's objections to the change in the Agreed Lise.

 Azardous Substances.
 (a) Reportable Uses Require
 Consent. The term "Hazardous Substance" as used in this Lease shall mean any product, or waste whose presence, use, manufacture, disposal, transportation, or release, either by itself or in combination with other materials expected to be on the Premises, is either: (i) potentielly injurious to the public health, safety or welfare, the environment or the Premises, (ii) regulated or monitored by any governmental substances in all include, but not be limited to, hydrocarbors, petroleum, gasoline, and/or crude oil or any products, by-products or fractions thereof. Lessee shall mean on the Premises which constitutes a Reportable Use of Hazardous Substances without the express prior written consent of Lesson and timely compliance (at Lessee's expense) with all Applicable Requirements. Reportable Use's shall mean (i) the installation or use of any above or below and timely compliance (at Lessee's expense) with all Applicable Requirements. Reportable Use's shall mean (i) the installation or use of any above or below and timely compliance (at Lessee's expense) with all Applicable Requirements. Reportable Use's shall mean (i) the installation or use of any above or below and timely compliance (at Lessee's expense) with all Applicable Requirements and customary materials reasonship required to be used in the normal course of neighboring properties. Notwithstanding the foregoing, Lessee any ordinary and customary materials reasonship required to be used in the normal course of neighboring properties. Notwithstanding the foregoing, Lessee are any use any ordinary and customary materials reasonship required to be used in the normal course of neighboring properties. Notwithstanding the toregoing, Lessee may use any ordinary and customary materials reasonship required to be us

Page 3 of 12

(FAX)209 667 1251

The solution recommended, when it is not contain or required, for the cleaned or restricting to any containination of, and for the maintenance, security and/or monitoring of the Premises or neighboring properties, that was caused or materially contributed to by Lesses, or pertaining to or involving any Hazardous Substance brought onto the Premises during the term of this Lesse, by or for Lesses, or any third party.
(d) Lesses indemnification. Lesses shall indemnify, defend and hold Lessor, its agents, employees, lenders and ground lessor, if any, harmiase from and against any and all loss of rents and/or demages, liabilities, judgments, claims, expenses, penalties, and attorneys' and consultants' fees arising out of or involving any Hazardous Substance brought onto the Premises by or for Lesses with respect to underground migration of any Hazardous Substance under the Premises from areas outside of the Project). Lesses's obligations shall be but not he affect of any effectively to the the affect of any enabled and used to the premises by or for Lesses, or any third party (provided, however, that Lesses with respect to underground migration of any Hazardous Substance under the Premises from areas outside of the Project). Lesses's obligations shall be but not he affect of any effectively to any the standed but with the affect of any enderground migration. shall include, but not be limited to, the effects of any contamination or injury to person, property or the environment created or suffered by Lessee, and the cost of investigation, removal, remediation, restoration and/or adatement, and shall survive the expiration or termination of this Lesse. No termination, cancellation or release agreement interred into by Lessor and Lessee shall release lessee from its obligations under this Lease with respect to Hazardous Substances, unless specifically so agreed by Lessor in writing at the time of such agreement. (e) Lessor Indemnification. Lessor and its successors and assigns shall indemnify, defend, reimburse and hold Lessee, its

employees and lenders, harmless from and against any and all environmental damages, including the cost of remediation, which wolds de as a result of Hazardous Substances on the Premises prior to the Start Date or which are caused by the gross negligence or willful misconduct of Lessor, its agents or employees. Lessor's obligations, as and when required by the Applicable Requirements, shall include, but not be limited to, the cost of investigation, namoval, remediation, restoration and/or abstement, and shall survive the expination or termination of this Lesse.

investigations and Remediations. Lessor shall retain the responsibility and pay for any investigations or neme (f)

(f) Investigations and Reinediations. Lessor shall retain the responsibility and pay for any investigations or remediation measures required by governmental entities having jurisdiction with respect to the existence of Hazardous Substances on the Premises prior to the Start Date, unless such remediation measure is required as a result of Lessor's juse (including "Alterations", as defined in paragraph 7.3(a) below) of the Premises prior to the Start Date, unless such remediation measure is required as a result of Lessor's juse (including "Alterations", as defined in paragraph 7.3(a) below) of the Premises in which event Lessor's egents to have reasonable access to the Premises at reasonable times in order to carry out Lessor's investigative and remediat responsibilities. Lessor Termination Option. If a Hazardous Substance Condition (see Paragraph 9.1(e)) occurs during the term of this Lesses, unless Lessee is legally responsibile therefor (in which case Lessee shall make the investigation and remediation thereof required by the Applicable Requirements and this Lesse shall continue in full force and effect, but subject to Lessor's rights under Paragraph 6.2(d) and Paragraph 13), Lessor may, at Lessor's option, either (i) investigate and remediate such Hazardous Substance Condition, if required, as soon as reasonably possible at Lessor's expense, in which event this Lessor's expense, greater, give written notice to Lessor (10) if the estimated dos to remediate such condition exceeds 12 times the thermontity Base Rent of \$100,000, whichever is greater, give written notice to Lessor 6 Lessor's floy Lessor or knowledge of the socurrence of such Hazardous Substance Condition, of Lessor's expense estimation notice, Lessore and field of also following the date of such notice. In the event Lessor elects to give a termination notice, Lessore and y, within 10 days thereafter, dive written notice to Lessor of Lessor's commitment to pay the sinount by which the cost of the remediation of such Hazardous Substance days thereafter, give written notice to Lessor of Lesser's commitment to pay the amount by which the cost of the remediation of such Hazardous Substance Condition exceeds an amount equal to 12 times the then monthly Base Rent or \$100,000, whichever is greater. Lesse shall provide Lessor with said funds or satisfactory assurance thereof within 30 days following such commitment. In such event, this Lesse shall continue in full force and effect, and Lessor shall proceed to make such remediation as soon as reasonably possible after the required funds are available. If Lessee does not give such notice and provide the required funds or assurance thereof within the time provided, this Lesse shall terminate as of the date specified in Lessor's notice of termination,

or assurance thereof within the time provided, this Lease shall terministe as of the date specified in Lessor's notice of termination. 6.3 Lesser's Compliance with Applicable Requirements. Except as otherwise provided in this Lease, Lessee shall, at Lesse's sole expanse, fully, diligently and in a timely manner, materially comply with all Applicable Requirements, the requirements of any applicable fire insurance underwriter or rating burceu, and the recommendiations of Lessor's engineers and/or consultants which relate in any manner to the Premises, without regard to whather said requirements are now in effect or become effective after the Start Date. Lessee shall, within 10 days after receipt of Lessor's written request, provide Lessor with copies of all permits and other documents, and other information evidencing Lessee's compliance with any Applicable Requirements specified by Lessor, and shall immediately upon receipt, notify Lessor in writing (with copies of any documents involved) of any threatened or actual claim, notice, citation, warning, compliant or report pertaining to or involving the failure of Lessees or the Premises to comply with any Applicable Requirements. 8.4 Inspection; Compliance. Lessori and Lessor's "Lender" (es defined in Paragreph 30) and consultants shall have the right to enter into Premises with this Lease. The cost of any such inspections shall be paid by Lessor, unless a violation of Applicable Requirements, or a contamination is found to exist or be limininent, or the inspection is requested or ordered by a governmental authority. In such case, Lessee shall upon request reimburse Lessor for the cost of such inspection, as long as such inspection is reasonably related to the violation or contamination.

7. Maintenance: Repairs: Utility Installations: Trade Fixtures and Alterations ee's Obligations.

7.1 Lessee's Obligations. (a) in General. Subject to the provisions of Paragraph 2.2 (Condition), 2.3 (Compliance), 8.3 (Lessee's Compliance with Applicable Requirements), 7.2 (Lessor's Obligatione), 9 (Damage or Destruction), and 14 (Condemnation), Lessee shall, at Lessee's sole expense, keep the Premises equilibrily installations (intended for Lessee's exclusive use, no matter where located), and Attrations in good order, condition and repair (whether or not the potion of the Premises equiring repairs, or the means of repairing the same, are reasonably or readily accessible to Lessee, and whether or not the need for such repairs occurs as a result of Lessee's use, any prior use, the elements or the age of such portion of the Premises provide to the elements or the age of such portion of the Premises eleves, and whether or not the need for such repairs occurs as a result of Lessee's use, any prior use, the elements or the age of such portion of the Premises in good order, condition and repair, which excluding any items (which are the responsibility of Lessoer pursuant to Paragraph 7.2. Lessee, in keeping the Premises in good order, condition and repair, shall exercise and perform good maintenance practices, specifically including the procurement and maintenance of the service contracts required by Paragraph 7.1(b) below. Lessee's obligations shall include restorations, replacements or renewals when necessary to keep the Premises and all improvements theremo are and thereoid in pood ender, conditions and repair, thereoid is not deter, condition and repair is thereoid in pood ender, conditions and repair, thereoid is not deter, conditions and sections, replacements or renewals when necessary to keep the Premises and all improvements theremo read in pood ender, condition and repair (there or a part thereoid in pood ender, conditions, and sections). improvements thereon or a part thereof in good order, condition and state of repair.

(b) Service Contracts. Lesses solution and state of repair.
(c) Service Contracts. Network and the service of the following equipment and improvements, if any, if and when installed on the Prenises.
(i) HVAC equipment, (ii) bother and pressure vessels, (iii) clarifiers, and (v) any other equipment, if reasonably required by Lesser. However, Lessor, and the realistic of the following equipment and improvements, if any, if and when installed on the Prenises.

demand, for the cost thereof. (c) Failure to Perform. If Lessee fails to perform Lessee's obligations under this Paragraph 7.1, Lessor may enter upon the Premises after 10 days' prior written notice to Lessee (except in the case of an emergency, in which case on notice shall be required), perform such obligations on Lessee's behalf, and put the Premises in good order, condition and repair, and Lesses shall promptly reimburse Lessor for the cost thereof. (d) Replacement. Subject to Lessee's indemnification of Lessor as set forth in Paragraph 8,7 below, and without relieving Lessee of liability

resulting from Lessee's failure to exercise and perform good mathemance prectices, if an item described in Paragraph 7.1(b) cannot be repeired other than at a cost which is in excess of 50% of the cost of replacing such item, then such item shall be replaced by Lessor, and the cost thereof shall be pronted other than at a cost and Lessee shall only be obligated to pay, each month during the remainder of the term of this Lesse, on the date on which Base Rent is due, an amount equal to the product of multiplying the cost of such replacement by a fraction, the numerator of which is one, and the denominator of which is 144 (ie. 1/144th of the cost per month). Lessee shall pay interest on the unamortized balance at a rate that is commercially reasonable in the judgment of Lessor's accountants. Lessee may,

month). Lessee shall pay Interest on the unamortized balance at a rate that is commarcially reasonable in the judgment of Lessor's accountants. Lessee may, however, prepay its obligation at any time. 7.2 Lessor's Obligations, Subject to the provisions of Paragraphs 2.2 (Condition), 2.3 (Compliance), 4.2 (Common Area Operating Expenses), 6 (Use), 7.1 (Lessee's Obligations), 9 (Damage or Destruction) and 14 (Condemnation), Lessor, subject to reimbursement pursuant to Paragraph 4.2, shall keep in good order, condition and repair the foundations, exterior walls, structural conditions of Interior bearing wells, exterior root, fire oprinkler system, Common Area Paragraph 4.2, shall keep in good order, condition and repair the foundations, exterior walls, structural conditions of Interior bearing wells, exterior root, fire oprinkler system, Common Area fire eatern and/or smoke detection systems, fire hydrants, parking lots, well keys, parkways, kindecepting, fences, eigen and utility systems activity of any starter or which there is a Common Area Operating Expense pursuant to Paragraph 4.2. Lessor shall not be obligated to pairt the exterior or interior surfaces of exterior walls nor shall Lessor be obligated to maintain, repair or replace windows, doors or plate glass of the Premises. Lessee expressly valves the benefit of any statute now or hereafter in effect to the extent it is inconsister; with the terms of this Lesse. 7.3 Utility Installations; Trade Fixtures; Alterations. (a) Definitions. The term "Utility Installations" refers to all floors and window coverings, air lines, power panels, electrical distribution, security and fire ordection systems communication avairants. Interme starts and starts and function is or on the Premises. The term "Trade Fixtures"

(a) Definitions. The term "Utility installations" refers to all floor and window coverings, air lines, power panels, electrical distribution, escurity and fire protection systems, communication systems, lighting fixtures, HVAC equipment, plumbing, and fencing in or on the Premises. The term "Trade Fixtures" shall mean Lesses's machinery and equipment that can be removed without doing material damage to the Premises. The term "Atterations" etail mean any modification of the improvements, other than utility installations or Trade Fixtures, winter to y dolling on or deletion. "Lessee Comed Atterations and/or Utility installations or Utility installations to the Premises without sessor's prior written consent. Lessee may, however, make non-structural Utility installations interior of the Premises (excluding the roof) without such consent but upon notice to Lessor, as long as they are not visible from the outside, do not involve puncturing, relocating or memory the noof or any existing walls, and the cumulative cost thereof during this Lesse estall not make any end/or attent and/or install anything on the roof without the prior written opproval of Lessor, as long as they are not visible from the outside, do not involve puncturing, relocating or memory the noof or any existing walls, and the cumulative cost thereof during this Lesse as extended does not exceed a sum equal to 3 monitrie Base Rent in the eggregate or a sum equal to one month's Base Rent in any one year. Notwithstanding the foregoing, Lessee shall not make or permit any or of penetration cord relation exported by Lessor. Any Atterations or Utility installations that Lessee shall deelne to make and which require the consent of the Lessor shall be presented to Lessor. Any Atterations or Utility installations frances with all conditions of sald permits and other Applicable Requirements in a prompt and expecifications pranter.

Th-

441 Initiak

Page 4 of 12

(FAX)209 667 1251

Alterations of Utility Installations be performed in a workmanlike manner with good and sufficient materials or with as-built plans and specificenons. For work which costs an amount in excess of one month's Base Rent, . ae shall promptly upon completion furnish Any Alterations or Utility Installations. Lessor with as-built plans and specifications. For work which costs an amount in excess of one month's Base Rent, _____or may condition its consent upon Lessee providing a lien and completion bond in an amount equal to 150% of the estimated cost of such Alteration or Utility Installation and/or upon Lessee's posting an additional Security Deposit with Lessor.

(c) Indemnification. Lassee shall pay, when due, all claims for labor or materials furnished or alleged to have been furnished to or for Lassee (c) Indemnification. Leases shall pay, when due, all claims for labor or materials furnished or alleged to have been furnished to or for Lesses at or for use on the Premises, which claims are or may be secured by any mechanic's or materials furnished or alleged to have been furnished to or for Lesses at a for use on the Premises, which claims are or may be secured by any mechanic's or materials for the against the Premises, and Lessor shall have the right to post notices of non-responsibility. If Lesses shall context the validity of any such tien, claim or demand, then Lesses shall, at its sole expense defend and protect itself, Lessor and the Premises against the same and shall pay and satisfy any such adverse judgment that may be rendered thereon before the enforcement thereof. If Lessor shall furnish a surely bond in an amount equal to 150% of the amount of such contexte lien, claim or demand, indemnifying Lessor segainst lies for the same. If Lessor elects to participate in any such action, Lessee shall pay Lessor's attorneys' fees and costs.
7.4 Ownership: Removal; Surrender; and Restoration.
(a) Duranzella, Subject in Lessor is lessing to remove a statice context in the installed to the statice termination of the annount of such contexts.

To use same. If Lesson exercises periodpane in any suon action, Lesson summing y Lesson's excorneys rees and costs. 7.4 Downership; Removal; Surrender; and Restoration. (a) Ownership; Subject to Lesson's right to require removal or elect ownership as hereinafter provided, all Attentions and Utility installations made by Lessee shall be the property of Lessee, but considered is part of the Premises. Lesson may, at any time, elect in writing to be the owner of all or any specified part of the Lessee Owned Attentions and Utility installations. Unless otherwise instructed per paragraph 7.4(b) hereof, all Lessee Owned Attentions and Utility installations shall, at the expiration of termination of the Lesse, because the property of Lessor and be surrendered by Lessee with the Premises. (b) Removal. By delivery to Lessee of written notice from Lessor not earlier than 90 and not later than 30 days prior to the end of the term of this Lesse, Lesson may require that any or all Lessee Owned Attentions or Utility installations indee without the required consent. (c) Surrender; Restoration. Lessee Solved Attentions or Utility installations indee without the required consent. (c) Surrender; Restoration. Lessee shall surrender the Premises by the Expiration Date or any earlier termination date, with all of the Improvements, parts and aurfaces thereof broom clean and free of debris, and in good operating order, condition and state of repair, ordinary wear and tear "Ordinary wear and tear" shall not helude any damage or delation that would have been prevented by good maintenance practice. Notwithstanding the foregoing, if this Lesse is for 12 months or less, then Lessee shall surrender the Premises by or tor Lessee, or any third party except Hazardous Substances brook allow any damage or delation that would have been prevented by prove the casee. Lessee that also completely Attentions and/or Utility installations, furnishings, and equipment as well as the removal or any statistant on a sea is the removal of Trade F

Insurance; Indemnity. 8.

8.1 Payment of Premium Increases. (a) As used herein, the term "Insurance Cost Increase" is defined as any increase in the actual cost of the insurance applicable to the Sulking and/or the Project and required to be carried by Lessor, pursuant to Paragraphs 8.2(b), 8.3(a) and 8.3(b), ("Required Insurance"), over and above the Base Premium, as hereinafter defined, calculated on an annual basis. Insurance Cost increases that include, but not be limited to, requirements of the holder of a merugage or deed of trust covering the Premiues, Building and/or Project, increase valuation of the Premises, Building and/or Project, and/or a general premium rate increase. The term insurance Cost increase shall not. however, include any premium increases resulting from the nature of the occupancy of any other tenant of the Building. If the parties insert a dolar amount in Paragraph 1.9, such amount shall be considered the "Base Premium." The Base Premium shall be the annual premium applicable to the 12 month period immediately preceding the Start Date. If, however, the Project was not insurance for the entirety of such 12 month period, then the Base Premium shall be the lowest annual premium reasonably obtainable for the Required insurance as of the Start Date, assuming the most nominal use possible of the Building. In no event, however, shall Lessee be responsible for any portion of the premium cost attributable to liability insurance coverage in excess of \$2,2000,000 procured under Paragraph 8.2(b). Payment of Premium Increases.

bosicies of the boarding. In the order, investment, that are the provided and the second seco

to, or extending beyond, the term of this Lease shall be prorated to coincide with the corresponding Start Date or Expiration Date.
8.2 Liability insurance.
(a) Carried by Lessee. Lease shall obtain and keep in force a Commercial General Liability policy of insurance protecting Lessee and Leasor maintenance of the Premises and all areas appurtneent thereto. Such insurance shall be on an occurrence basis providing single limit coverage in an amount not less than \$1,000,000 per occurrence with an ennual aggregate of not less than \$2,000,000, an "Additional insured-Managero or Lessors of Premises Endorsement" for damage caused by heat, smole or furnes from a hostile fire. The policy shall not contain any intra-insured exclusions as between insured persons of organizations, but shall include coverage for fability assumed under this Lesses on a "insured contract" for the parformance of Lesser's informantly obligation hersunder. All insurance carried by Lesses. The limits of said insurance shall not, however, limit the liability of Lessee nor raiseve Lessee of any obligation hersunder. All insurance carried by Lesses that in exclusion insurance shall be considered excess insurance only.
(b) Carried by Lessor. Lessore shall maintain liability insurance es described in Paragraph 8.2(a), in addition to, and not in lieu of, the insurance shall be creatived to be maintained by Lessee. Lessee shall and keep in force a policy or policies of insurance in the name of Lessor, with loss reparation to the remines, and to any tender insuring loss or damage to the Premises, and to any tender the under the tothe or the another, but in no event more than the commercial be equal to the full reparative lass of and the provements. Lessor shall obtain and keep in force a policy or policies of insurance in the name of Lessor, with loss property insurance - Building, improvements and Rental Value.
8.3 Property insurance. Lessore shall be the time, or the annount required by any context, but in occurrence.

(b) Rental Value. Lessor shall also obtain and keep in force a policy or policies in the name of Lessor with loss payable to Lessor and any Lender, insuring the loss of the full Rent for one year with an extended period of Indemnity for an additional 180 days ("Rental Value Insurance"). Said insurance shall contain an agreed valuation provision in lieu of any coinsurance clause, and the amount of coverage shall be adjusted annually to reflect the projected Rent otherwise payable by Lesses, for the next 12 month period.

 otherwise payable by Lesses, for the next 12 month period.
 (c) Adjacent Premises. Lesses shall pay for any increase in the premiume for the property insurance of the Building and for the Common Areas or other buildings in the Project if said increase is caused by Lesses's acts, omissions, use or occupency of the Premises.

 Areas or other buildings in the Project if said increase is caused by Lesses's acts, omissions, use or occupency of the Premises.

 (d) Lesses's Improvements. Since Lessor is the insuring Party, Lessor shall not be required to insure Lessee Owned Alterations and Utility Installations unless the item in question has become the property of Lessor under the terms of this Lesse.

 8.4
 Lesses's Property; Business Interruption Insurance.

 (a) Property Burnage. Lessee shall obtain and maintain Insurance coverage on all of Lessee's personal property, Trade Fixtures, and Lessee (a) Property; Business Interruption Insurance shall be full replacement cost coverage with a deductible of not to exceed \$1,000 per occurrence. The proceeds from any euch insurance shall be full replacement of personal property. Trade Fixtures and Utility installations. Lessee for the replacement of personal property. Trade Fixtures and Utility installations. Lessee for the replacement of personal property. Trade Fixtures and Utility installations. Lessee that envidence that such insurance is information.

 (b) Business Interruption. Lessee shall chain and maintain loss of Income and extra expense insurance in amounts as will relimburable to prevention of access to the Premises as a result of such period.

access to the Premises as a result of such perils.

In create or intersections as a result of such perils. (c) No Representation of Adequate Coverage. Lessor makes no representation that the limits or forms of coverage of insurance specified herein are adequate to cover Lessee's property, business operations or obligations under this Lesse. S. Insurance Policies. Insurance required herein shall be by companies duy licensed or admitted to transact business in the state where the S. Insurance Policies. Insurance required herein shall be by companies duy licensed or admitted to transact business in the state where the S. Insurance Policies. Insurance required herein shall be by companies duy licensed or admitted to transact business in the state where the set located, and maintaining during the policy term a "General Policyholders Rating" of at teast 8+. V, as set forth in the most current issue of "Best's insurance Guide", or such other rating as may be required by a Lender. Lessee shall not do or permit to be done anything which invalidates the required insurance policies. Lessee shall, prior to the Stat Date, deliver to Lessor certified copies of policies of such insurance or certificate evidencing the existence and amounts of days prior to the expiration of such policies, furnish besor with evidence or renewale or "insurance binders" evidencing renewal thereof, or Lessor may order such days prior to the expiration of such policies, furnish beaser with evidence or renewale or "insurance binders" evidencing renewal thereof, or Lesser may order such line current length of the remaining term of this lesse, which amount shall be payable by Lessee to Lessor upon demand. Such policies shall be for a tarm of at least one set, or the length of the remaining term of this lessee. Michever is less. If either Party shall fail to procure and maintain the insurance required to be carried by it, the other Party may, but shall not be required to, procure and intaintain the same. 8.6 Waiver of Subrogation. Without affecting any other rights or remedies, Lessee and

UA

Initials

Page 5 of 12

79

herein. The effect of such reis. and walvers is not limited by the amount of insurance carried or required, o y deductibles applicable hereto. The Parties agree to have their respective property damage insurance carriers waive any right to subrogation that such comp., se may have against Lessor or Lessoe, as the case may be, so long as the insurance is not invalidated thereby.

case may be, so long as the insurance is not invalidated therapy. 8.7 indemnity. Except for Lessor's gross negligence or willful misconduct, Lessee shall indemnify, protect, defend and hold harmless the Premises, Lessor and its agents, Lessor's master or ground lessor, partners and Lenders, from and against any and all claims, loss of rents and/or damages, liens, judgments, penalties, attorneys and consultants' fees, expenses and/or lessor's institute artsing out of, involving, or in connection with, the use and/or occupancy of the Premises by Lessee. If any action or proceeding is brought against Lessor by reason of any of the foregoing matters, Lesser shall upon notice defend the same at Lesse's obparse by course! reasonably satisfactory to Lessor and Lessor shall cooperate with Lessee in such defense. Lessor need not have first paid any such table to be added or builted. claim in order to be defended or indemnified.

claim in order to be defended or indemnified, 8.8 Exemption of Lesson from Liability property of Lesson, Lesson's employees, contractors, invites, customers, or any other person in or about the Premises, whether such damage or injury is caused by or results from fire, steam, electricity, gas, water or rain, or from the breakage, leakage, obstruction or other defects of pipes, fire sprinklers, where, appliances, plumbing, HVAC or lighting fixtures, or from any other cause, whether the said injury or damage results from conditions arising upon the Premises or upon other portions of the Building, or from other sources or places. Lessor shall not be liable for any damages arising from any et or negligence or breach of this Lessor shall under no circumstances be liable for injury to Lessen's business or for any loss of income or profit therefrom.

Damage or Destruction. 9. Definitions.

(a) "Premises Partial Damage" shall mean damage or destruction to the improvements on the Premises, other than Lessee Owned Alterations and Utility Installations, which can reasonably be repaired in 3 months or less from the date of the damage or destruction, and the cost thereof does not exceed a sum equal to 6 months Base Rent. Lessor shall notify Lessee in writing within 30 days from the date of the damage or destruction as to whether or not the damage is Partial or Total.

(b) "Premises Total Destruction" shall mean damage or destruction to the improvements on the Premises, other than Lessee Owned Alterations and Utility installations and Trade Fixtures, which cannot reasonably be repaired in 3 months or less from the date of the damage or destruction and/or the cost thereof exceeds a sum equal to 6 month's Base Rent. Lessor shall notify Lessee in writing within 30 days from the date of the damage or destruction as to

(o) "insured Loss" shall mean damage or destruction to improvements on the Premises, other than Lessee Owned Alterations and Utility (o) "insured Loss" shall mean damage or destruction to improvements on the Premises, other than Lessee Owned Alterations and Utility installations and Trade Fidures, which was caused by an event required to be covered by the insurance described in Paragraph 8.3(a), traspective of any deductible amounts or coverage limits involved.

(d) "Replacement Cost" shall mean the cost to repair or rebuild the improvements owned by Lessor at the time of the occurrence to their condition existing immediately prior thereto, including demolition, debris removal and upgrading required by the operation of Applicable Requirements, and without deduction for depreciation.

deduction for depreciation.
(e) "Hazardous Substance Condition" shall mean the occurrence or discovery of a condition involving the presence of, or a contamination by,
(a) Hazardous Substance as defined in Peragraph 6.2(a), in, on, or under the Premises.
9.2 Partial Damage - Insured Loss. If a Premises Partial Damage that is an insured Loss occurs, then Lessor shall, at Lessor's expense, repair such damage (but not Lessee's Trade Fixtures or Lessee Owned Alterations and Utility Installations) as soon as reasonably possible and this Lesse shall continue in full force and effect; provided, however, that Lessor's expense, repair such damage (but not Lessee's Trade Fixtures or Lessee Owned Alterations and Utility Installations) as soon as reasonably possible and this Lesse shall continue in full force and effect; provided, however, that Lessor's expense, repair such damage (but not Lessee's Trade Fixtures or Lessee Owned Alterations and Utility Installations) as soon as reasonable possible and this Lesse shall continue in full force and effect; provided, however, that Lessor's expense, repair of any damage or destruction the total cost to repair of which is \$5,000 or less, and, in such event, Lessor shall make any applicable insurance proceeds aveilable to Lessee on a reasonable basis for that purpose. Notwithstanding the foregoing, if the required insurance was not in inforce or the insurance proceeds are not sufficient to effect such repair, the insuring Party shall promptly contribute the shortage in proceeds as and when required to complete said repairs. In the event, however, such shortage was due to the fact that, by reason of the unique nature of the improvements, full replacement cost insurance coverage was not commercially reasonable and available, Lessor shall have no obligation to pay for the shortage in Insurance proceeds or to fully restore the unique age and execute provides Lessor with the finds to cover obligation to pay for the shortage in Insurance proceeds or to the unique

obligation to pay for the shortage in insurance proceeds or to fully restore the unique aspects of the Premises unless Lesses provides Lessor with the funds to cover adequate assurance thereof, within 10 days following receipt of written notice of such shortage and request therefor. If Lessor receives table and this Lease shall remain in full force and effect. If such funds or assurance are not received, Lessor provides Lessor by possible and this Lease shall remain in full force and effect. If such funds or assurance are not received, Lessor maying any shortage in proceeds, in which case this Lease shall commercially reasonable with Lessor paying any shortage in proceeds, in which case this Lease shall remain in full force and effect, or (ii) have this Lesse terminate 30 days thereafter. Lesser shall not be entitled to reimbursement of any funds contributed by Lesser to repair any such damage or destruction. Premises Partial Damage due to flood or earthquike shall be aubject to Paregraph 33, notvithistanding that there may be some insurance coverage, but the net proceeds of any such insurance anell be made available for the repairs if made by either Party. 9.3 Partial Demage - Uninsured Loss. If a Premises Partial Damage that is not an insured Less cours, unless caused by a negligent or willful act of Lessee (in which event this Lesse shall ned the repairs at Lessee's expense), Lessor may either: (i) repair such damage as soon as reasonably possible after receipt by Lessor of incovietige of the occurrence of such damage. Such termination shall be effective 60 days following the date of such notice. In the event Lessee is advected, in which event this Lesse shall continue in full force and effect, or (ii) terminate this Lesse by giving written notice to Lessee within 30 days after receipt by Lessor of incovietige of the occurrence of such damage. Such termination shall be effective 60 days following the date of such notice. In the event Lessor is advected, in which event this Lesse shall continue in Lessor. Les

specified in the termination notice. 9.4 Total Destruction. Notwithstanding any other provision hereof, if a Premises Total Destruction occurs, this Lease shall terminate 60 days following such Destruction. If the damage or destruction was caused by the gross negligence or withtui misconduct of Leasee, Lessor shall have the right to recover Lessor's damages from Lessee, except as provided in Paragraph 8.6.

Lessor's damages from Lessee, except as provided in Paragraph 8.6. 9.5 Damage Near End of Term. If at any time during the last 6 months of this Lesse there is damage for which the cost to repair exceeds one month's Base Rent, whether or not an insured Less, Lessor may terminate this Lesse effective 60 days following the date of occurrence of such damage by giving a written termination notice to Lessee within 30 days after the date of occurrence of such damage. Notwithstanding the foregoing, if Lessee at that time has an exercisable option to extend this Lesse within 30 days after the date of occurrence of such damage. Notwithstanding the foregoing, if Lessee at that time has an exercisable option to extend this Lesse or to purchase the Premises, then Lessee may preserve this Lesse by, (a) exorcising such option and (b) providing Lessor with any shortage in Insurance proceeds (or adequate assurance thereof) needed to make the repairs on or before the earlier of (i) the date which is 10 days after Lessee's receipt of Lessor's written notice purporting to terminate this Lesse, or (ii) the date upon which such option applies. If Lessee duly exercises such option during such period and provides Lessor with funds (or adequate assurance thereof) to cover any shortage in insurance proceeds, Lessor's commercially reasonable expense, repair such damage as ocn as reasonably possible and this Lesse shall continue in full force and effect. If Lessee fails to exercise such option and provide such funds or assurance during such period, then this Lesse shall terminate on the date specified in the termination notice and Lessee's contion challs and provide such funds or assurance during such period, then this Lesse shall terminate on the date specified in the termination notice and Lessee's colon shall be admontished. Lessee's option shall be extinguished.

Abatement of Rent; Lessee's Rem

9.6 Abstament of Rent; Lessee's Remedies. (a) Abstament. In the event of Premises Partial Damage or Premises Total Destruction or a Hazardous Substance Condition for which Lessee Is not responsible under this Lesse, the Rent payable by Lessee for the partial required for the repair, remediation or restoration of such damage shall be abated in proportion to the degree to which Lessee's use of the Premises is Impaired, but not to exceed the proceeds received from the Rental Value insurance. All other obligations of Lessee hereunder shall be performed by Lessee, and Lessor shall have no liability for any such damage, destruction, remediation, repair or restoration except as provided herein.

(b) Remedies. If Lessor shall be obligated to repair or restore the Pramiess and does not commence, in a substantial and meaningful way, such repair or restoration within 90 days after such obligation shall accrue, Lessee may, at any time prior to the commencement of such repair or restoration, give written notice to Lessor and to any Lenders of which Lessee has actual notice, of Lessee's election to terminate this Lesse on a date not less than 60 days following the glving of such notice. If Lessee gives such notice and such repair or restoration is not commenced within 30 days thereafter, this Lesse shall terminate as of the date specified in said notice. If the repair or restoration is not commenced within 30 days thereafter, this Lesse shall terminate as of the mean either the unconditional authorization of the preparation of the required plans, or the beginning of the ectual work on the Premises, whichever first occurs. 9.7 Termination; Advance Payments. Upon termination of this Lesse bursent to Paragraph 6.2(g) or Peragraph 9.4 equivalent adjustment shall do concerning advance Base Rent and any other advance payments made by Lessee to Lessor. Lessor shall in addition, return to Lessee so much of Lesse's Security Deposit as has not been, or is not then required to ba, used by Lesser. (b) Remedies. If Lessor shall be obligated to repair or restore the Premiees and does not commence, in a substantial and meaningful way,

Real Property Taxes. 10,

Definitions. 10.1

10.1 Definitions. (a) "Real Property Taxes." As used herein, the term "Real Property Taxes." shall include any form of assessment; real estate, general, special, ordinary or extraordinary, or rental lawy or tax (other than inheritance, personal income or estate taxes); improvement bond; and/or license fee imposed upon or leviad egainst any legal or equilable interest of Lessor in the Project, Lessor's right to other income therefrom, and/or Lessor's business of leasing, by any sufficiently having the direct or indirect power to tax and where the funds are generated with reference to the Project address and where the proceeds so generated are to be applied by the city, county or other local taxing authority of a jurisdiction within which the Project is located. The term "Real Property Taxes."



Page 6 of 12

Initials

tax, ites, isvy, assessment or ca _____, or any increase therein, imposed by reason of events occurring during th ______, of this Lease, including but not limited to, a change in the ownership of the Project or any portion thereof or a change in the improvements thereon. (b) "these Real Property Taxes." As used herein, the term "Base Real Property Taxes" shall be the amount of Real Property Taxes, which are assessed against the Premises, Building, Project or Common Areas in the calendar year during which the Lease is executed. In calculating Real Property Taxes for any calendar year, the Real Property Taxes for any real estate tax year shall be included in the calculation of Real Property Taxes for such calendar year based

upon the number of days which such calendar year and tax year have in common. 10.2 Payment of Taxes. Lessor shall pay the Real Property Taxes applicable to the Project, and except as otherwise provided in Paragraph 10.3, any increases in such amounts over the Base Real Property Taxes shall be included in the calculation of Common Area Operating Expenses in accordance with the provisions of Paragraph 4.2.

provisions of Paragraph 4.2. 10.3 Additional improvements. Common Area Operating Expanses shall not include Real Property Taxes specified in the tax assessor's records and work sheets as being caused by additional improvements placed upon the Project by other teases or by Lessor for the exclusive enjoyment of such other teasees. Notwithstanding Paragraph 10.2 hereof, Lesses shall, however, pay to Lessor at the time Common Area Operating Expenses are payable under Paragraph 4.2, the entirety of any increase in Real Property Taxes if assessed solely by reason of Alterations, Trade Fidures or Utility installations placed upon the Premises by

4.2, the entroy of any increase in real Property factor in excessed svery by feeder to Acatabria, include to the Building and been equitable property factor in the second state of the Building and been equitable properties of the Building and the frame and improvements included within the tax parcel assessed, such properties to be determined by Lessor's reasonable determined by Lessor's to a set of the set of

roun, on an op concursive. 10.5 Personal Property Taxes. Lessee shall pay prior to delinquency all taxes assessed against and levied upon Lessee Owned Alterations and Utility installations, Trade Fodures, furnishings, equipment and all personal property of Lessee contained in the Premises. When possible, Lessee shall cause its Lessee Owned Alterations and Utility installations, Trade Fidures, furnishings, equipment and all other personal property to be assessed and billed separately from the real property of Lesser. If any of Lesser's said property shall be assessed with Lesser's real property, Lessee shall pay Lessor the bross attributable to Lesser's property within 10 days after receipt of a written statement setting forth the taxes applicable to Lesser's property.

11. Utilities. Lesses shall pay for all water, gas, heat, light, power, telephone, trash disposal and other utilities and services supplied to the Premises, together with any taxes thereon. Notwithetanding the provisions of Paragraph 4.2, if et any time in Lessor's sole judgment, Lessor determines that Lesse is using a disproportionate amount of water, electricity or other commonly metered utilities, or that Lesse is generating such a large volume of trash as to require an increase in the size of the dumpster and/or an increase in the number of times per month that the dumpster is amplied, then Lessor may increase Lesser's Base Rent by an amount equal to such increased posts.

12.

Assignment and Subletting. 12.1 Lessor's Consent Required. (a) Lessoe shall not voluntarily or by operation of law assign, transfer, mortgage or encumber (collectively, "assign or assignment") or sublet part of Lessoe's inferest in this Lessoe of in the Praintises without Lessor's prior written consent. (b) A change in the control of Lessee shall constitute an assignment requiring consent. The transfer, on a cumulative basis, of 25% or more of (c) A change in the control of Lessee shall constitute an assignment requiring consent. The transfer, on a cumulative basis, of 25% or more of all or any part of Le

(b) A change in the control of Lessee shall constitute an assignment requiring consent. The transfer, on a cumulative basis, of 25% or more of the voting control of Lessee shall constitute a change in control for this purpose.
 (c) The involvement of Lessee or its assignment or hypotheciation, or series of transactions (by way of merger, sale, acquisition, financing, transfer, leveraged buy-out or otherwise), whether or not a formal assignment or hypotheciation of this Lease or Lessee's assets occurs, which results or will result in a reduction of the Net Worth of Lessee by an amount greater than 25% of such Net Worth as it was represented at the time of the execution of this Lease or at the time of the execution of this Lesse or at the time of the execution of this Lessee is an yrise set of the second of this Lessee or assignment or sublating and the net worth of Lessee is an yrise sented, or as it exists immediately prior to add transaction or transactions constituting such reduction, which was or is greater, shall be considered an assignment of this Lesse to which Lesson may withhold its consent. "Net Worth of Lessee" shall mean the net worth of Lessee (woluding any guarantors) established under generally accepted accounting principles.
 (d) An assignment or sublating without consent eikal, at Lessor's option, be a Default curable after notice per Perggraph 13.1(c), or a noncurable Breach without the necessity of any notice and grace period. If Lessor elects to transaction to 10% of the Base Rent then in effect. Further, in the worth of such and remains adjustment; (i) the purchase price of any option to purchase the monthly Base Rent to 10% of the Base Rent then in effect. Further, in the solut of such Breach and remain adjustment; (i) the purchase price of any option to purchase the duiled during the remainder of the Lesse than thall be increased to 110% of the scheduled adjusted rent.
 (e) Lessee's remady for any breach of Paragraph 12.1 by Lesseor shell be

to 110% of the price previously in effect, and (ii) all fload and non-fixed rental adjustments acheduled during the remainder of the Lease term shall be increased to 110% of the scheduled adjusted rent. (a) Lessee's remedy for any breach of Paragraph 12.1 by Lessor shell be limited to compensatory damages and/or injunctive relief. 12.2 Terms and Conditions Applicable to Assignment or subletting. (a) Regardless of Lessee under this Lesse, (ii) release Lessee of any obligations hereunder, or (iii) after the primary liability of Lessee for the partement of sublessee of the obligations of Lessee under this Lesse, (ii) release Lessee of any obligations hereunder, or (iii) after the primary liability of Lessee for the partement of subjects to be performance of Lessee's obligations from any person other than Lessee pending approval or disapproval of an assignment nor the acceptance of Rent or for the performance of any assignment or sublessee. (b) Lessor may accept Rent or performance of Lessee, Lessor may person other than Lessee pending approval or disapproval of such assignment nor the acceptance of Rent or performance shall constitute a waiver or estopped of Lesseo's right to exercise its remedies for Lessee's Default or Breach. (c) Lessor's consent to any assignment or subletting shall not constitute a consent to any subsequent assignment or subletting. (d) In the event of any Default or Breach by Lessee, Lessor may proceed directly against Lesser's remedies against any other person or entity responsible therefore to Lessor, or any assignment or subletting shall be in writing, accompanied by information relevant to Lessor's determination as the financial and operational responsible with a fee of \$1,000 or 10% of the proposed essignment or entities achausting and processing aid request. Lessee agrees to introduce on the premises which is the subject of the proposed assignment or subletting and processing aid request. Lessee agrees to provide Lessor with a use of the operoted essignment or entering and proces

Leasof has specifically consented to in Wrang. 12.3 Additional Terms and Conditions Applicable to Subjetting. The following terms and conditions shall apply to any subjetting by Lessee of all or any part of the Premises and shall be deemed included in all subjesses under this Lesse whether or not expressly incorporated therein: (a) Lessee hereby assigns and transfers to Lessor all of Lessee's interest in all Rent payable on any subjects, and Lessor may collect such Rent and apply same foward Lessee's obligations under this Lesse; provided, however, that until a Breach shall occur in the performance of Lessee's obligations of Pert be determined in the collection of Pert be determined for the subjects of the collection of Pert be Term and approximate Lessers of a unitable to be a series of the series directs any such sublessee, upon receipt of a written notice from Lesser stating that a Breach exists in the performance of Lesser's obligations under this Lesse, to pay to Lessor all Rent due and to become due under the sublesse. Sublessee shall rely upon any such notice from Lessor and shall pay all Rents to Lessor without any obligation or right to inquire as to whether such Breach exists, notwithstanding any claim from Lesser to the contrary. (b) In the event of a Breach by Lesser, Lessor may, at its option, require sublesses to the contrary. (b) In the event of a Breach by Lesser, Lessor may, at its option, require sublesses to the contrary. (c) Any matter requiring the consent of the sublesser or dread or any performance of Lessors. (c) Any matter requiring the consent of the sublesser under a sublesse shall also require the consent of Lessor. (d) No sublesses shall further assign or sublet all or any performance of the explessers, who shall have the right to cure the Default of Lessers within the grace period, if any, specified in such nodes. The sublessee shall have a right of reimbursement and offset from and against Lessee for any such Defaults cured by the sublesses.

13. Default: Breach: Remedies.

 Joint present remotes.
 13.1 Default; Breach. A "Default" is defined as a failure by the Lessee to comply with or perform any of the terms, covenants, conditions or Rules and Regulations under this Lessee. A "Breach" is defined as the occurrence of one or more of the following Defaults, and the failure of Lessee to cure such Default within any applicable grace period:

(e) The abandonment of the Premises; or the vacating of the Premises without providing a commercially reasonable level of security, or where the covarage of the property insurance described in Paragraph 8.3 is jeoperdized as a result thereof, or without providing reasonable essurances to minimize potential vandalism.

(b) The failure of Lessee to make any payment of Rent or any Security Deposit required to be made by Lessee hersunder, whether to Lessor or to a third party, when due, to provide reasonable exidence of insurance or surely bond, or to fulfill any obligation under this Lesse which endangers or threatens life or property, where such failure continues for a period of 3 business days following written notice to Lessee.

1Af Initials

AL.

Page 7 of 12

hereof, other than those described in subparagraphs 13.1(a), (b) or (c), shore, where such bisfaut continues for a period of 30 days after written notice, provided, however, that if the nature of Lessee's Default is such that more than 30 days are reasonably required for its our, then it shell not be dearmed to be a Breach if Lessee, the written notice, provided, the such state of a paint of the fellowing swritts. (f) the making of any ganetal arrangement or susignment for the bareff of creditors; the dearmed be written notice, provided, where possession is not restored to Lessee's thirts therein (thanks, in the case of a patient) arrangement or susignment for the bareff of creditors; where possession is not restored to Lessee's thirts within 30 days, or (b) the statement, suscition or their provides of the vertility of the sussession is not restored to Lessee's thirts within 30 days, or (b) the statement, suscition or due to dealer provide, however, in the sussession is not restored to Lessee's thirts within a subparagraph (e) is contrary to way splicitable with a subparagraph (e) is contrary to way splicitable within a subparagraph (e) is contrary to way splicitable within a subparagraph (e) is contrary to way splicitable within a subparagraph (e) is contrary to way splicitable within the terms of such guarantic divent and and state the withing these. (i) If the parformation of a the subparagraph (e) is contrary to obligation on the subparagraph (e) is contrary to subparagraph (e) is contrary to obligation on the subparagraph (e) is contrary to event the asystem of the subparagraph (e) is contrary to obligation on the subparagraph (e) is contrary to event the asystem of a subparagraph (e) is contrary to obligation on the subparagraph (e) is contrary to event the subparagraph (e) is contrary to obligation on the s

subject only to reasonable limitations. Acts of maintenance, efforts to relet, and/or the appointment of a receiver to protect the Lesser's interests, shall not constitute a termination of the Lesser's interests, shall not constitute a termination of the Lesser's infinite possession. (c) Pursue any other remady now or hereafter available under the laws or judicial decisions of the state wherein the Premises are located. The expiration or termination of this Lesse and/or the termination of Lesser's right to possession shall not releve Lesser from liability under any indemnity provisions of this Lesse as to matters occurring or accruing during the term intereof or by reason of Lesser's occupancy of the Premises. 13.3 Inducement Recepture. Any agreement for these or abated rent or other changes, or for the gMing or paying by Lessor to or for Lesser's antering into this Lesse, all of which concessions are hereinater reterred to as "inducement Provisions", shall be deemed conditioned upon Lesser's full and faithful performance of all of the terms, covenants and conditions of this Lesse. Upon Breach of this Lesse by Lesses, any such inducement Provision shall extormaticely be deemed deleted from this Lesser and of no further force or effect, and any rent, other charge, bonus, inducement or consideration for tested, given or paid by Lesser under such an inducement Provision shall extormated by Lesser to Lesser of rent or other bonus and on the termidately due and payable by Lesser to Lesser of the pregram thall not be deemed which initiated the such acceptance by Lesser of rent or other bonus and other bonus and the such acceptance by Lesser and of no further force or effect, and any rent, other charge, bonus, inducement or consideration theretofore abated, given or paid by Lesser. The acceptance by Lessor of rent or the Breach which initiated the operation of this paragraph shall not be deemed wide by Lessor of rent or the cure of the Breach which initiated the such acceptance.

13.4 Late Charges. Lesses hereby acknowledges that late payment by Lesses of Rent will cause Lessor to incur costs not contemplated by this Lesse, the exact amount of which will be externely difficult to ascertain. Such costs include, but are not limited to, processing and accounting charges, and late charges within 5 days after such amount shall be due, then, without any requirement for notice to Lesses, Lesses chall pay to Lesser a on-stime late charge equal to 10% of each such overdue amount shall be due, then, without any requirement for notice to Lesses, Lesses chall pay to Lesser a on-stime late charge equal to 10% of each such overdue amount of \$100, whichever is greater. The parties hereby agree that such late charge represents a fair and reasonable estimate of the costs. Lessor will hour by reason of such late payment the available in a subject of such also charge agree that such late charge state such and reasonable estimate of Breach with respect to such overdue amount, nor prevent the exercise of any of the other rights and remedies granted hereunder. In the event that a late charge is payable hereunder, whether or not collected, for 3 consecutive insubinents of Base Rent, then notwithstanding any provision of the Lesse to the contrary. Base Rent shall, at Lessor's option, become due and payable quarterly in advance.

13.5 Interest. Any monetary payment due Lessor hereunder, other than late charges, not received by Lessor, when due as to scheduled payments (such as Base Rent) or within 30 days following the date on which it was due for non-scheduled payment, shall beer interest from the date when due, as to scheduled payments, or the 31st day after it was due as to non-scheduled payments. The interest ("interest") charged shall be equal to the prime rate reported in the Wall Street Journal as published closest prior to the date when due plus 4%, but shall not exceed the maximum rate allowed by law. Interest is payable in addition to the potential late charge provided for in Paragraph 13.4.

Breach by 13.6

13.6 Breach by Lessor. (a) Notice of Streach. Lessor shall not be deemed in breach of this Lesse unless Lessor falls within a reasonable time to perform an obligation required to be performed by Lessor. For purposes of this Paragraph, a reasonable time shell in no event be less than 30 days after receipt by Lessor, and any Lender whose name and address shall have been furnished Lesser. In writing for such purpose, of written notice specifying wherein such obligation of Lessor has not been performed; provided, however, that if the nature of Lessor's obligation is such that more than 30 days after receipt of use of the performance is commenced within such 30 day period and thereafter diligently pursued to completion. (b) Performance by Lessee on Sehalf of Lessor. In the event that more than 30 days after cures said breach within 30 days after receipt of said notice, or if having commenced said cure they do not diligently pursue it to completion, then Lessee may elect to cure said breach at Lesse's expenses and offset from Rent an amount equal to the greater of one month's Base Rent or the Security Deposit, and to pay an excess of such expense under protest, reserving Lessee's right to reimburgement from Lessor. Lessee. Lessee's near the court the case of such expense and offset from Rent an amount equal to the greater of one month's Base Rent or the Security Deposit, and to pay an excess of such expense under protest, reserving Lessee's right to reimburgement from Lessor.

14. Condemnation. If the Premises or any portion thereof are taken under the power of eminent domain or sold under the threat of the exercise of said power (collectively "Condemnation"), this Lease shall be triminate as to the part taken as of the date the condemning authority takes tills or possession, whichever first cocurs. If more than 10% of the floor area of the Unit, or more than 25% of Lesser's Reserved Parking Spaces, is taken by Condemnation, Lesses may, at Lessee's option, to be exercised in writing within 10 days after Leasor shall have given Lessee writien notice of such taking (or in the absence of such notice, within 10 days after Leasor shall have given Lessee writien notice of such taking (or in the absence of such notice, within 10 days after Leasor shall have given Lessee writien notice of such taking (or in the absence of such notice, within 10 days after the condemning authority takes such possession). If Lessee does not terminate this Lease in accordance with the foregoing, this lesse at all remain in full force and effect as to the portion of the Premises remaining, except that the property of Lessor, whether such award shall be made as compensation for Lesse's relocated, however, that Lessee shall be antitied to any compensation for Lesse's relocation, which ever and of the premises goodwill and/or trade Fixtures, damages; provided, however, that Lessee shall be antitied to any compensation of the Premises and Utility installations made to the Premises by lessee, for purposes of Condemnation only, shall be considered the property of the Lessee shall be antitied by reason of the Condemnation, Lessee shall be antitied to any and all compensation for Lessee shall be antitied to any and all compensation for Lessee shall be antitied to any and all compensation for Lessee shall be antitied to any and all compensation for Lessee shall be antitied to any compensation of the Same and Lessee shall be antitied to any compensation of the Same and Lessee shall be antitied to any compensation of the C

Brokerage Fees

Page 8 of 12

07/27/2016 15:13 ACME ELECTRIC

(FAX)209 667 1251

P 012/019

15.1 Additional Con. on. In addition to the payments owed pursuant to Paragraph 1.10 above, Views Lessor and the Brokers otherwise agree in writing, Lessor agrees that: (by , Lessor expression of the Premises are course from Lessor any right, Premises or other premises owned by Lessor and located within the Project, (c) if Lessee remains in possession of the Premises, with the consent of Lessor, after the expiration of this Lesse, or (d) if Base Rent is increased, whether by agreement or operation of an escalation clause herein, then, Lessor shall pay Brokers a fee in accordance with the schedule of the Brokers in effect at the time of the excution of this Lesse.

15.2 Assumption of Obligations. Any buyer or transferee of Lessor's interest in this Lease shall be deemed to have assumed Lessor's obligation hereunder. Brokers shall be third party beneficiaries of the provisions of Paragraphs 1.10, 15, 22 and 31. If Lessor fails to pay to Brokers any amounts due as and for brokerage fees pertaining to this Lesse when due, then such amounts shall acome interest. In addition, if Lessor fails to pay any amounts to Lesser's Broker when due, Lesser's Broker may send written notice to Lessor and Lesser of such failure and if Lessor fails to pay such amount within 10 days after said notice, Lessee shall pay said monies to its Broker and offset such amounts against Rant. In addition, Lesser's Broker shall be deemed to be a third party beneficiary of any commission agreement entered into by and/or between Lessor and Lessor's Broker for the limited purpose of collecting any brokerage fee owed.

Contraineston agreement entered into by anow resween Lesson and Lesson's crossen in the innited purpose of contacting any protecting in a provide graved. 15.3 Representations and indemnities of Broker Relationships. Lessoe and Lessor each represent and warrant to the other that it has had no dealings with any person, firm, broker or finder of the trokers, if any in connection with this Lessoe and their or one other than suid named Brokers is entitled to any commission or finder's fee in connection herewith. Lessee and Lessor do each hereby agree to indemnify, protect, defend and hold the other harmlesse from and against liability for compensation or charges which may be claimed by any such unnamed broker, finder or other similar party by reason of any dealings or actions of the Indemnifying Party, including any costs, experises, attorneys' fees reasonably incurred with respect thereto.

16. Estoppel Certificates

(a) Each Party (as "Responding Party") shall within 10 days after written notice from the other Party (the "Requesting Party") exacute, acknowledge and deliver to the Requesting Party a statement in writing in form similar to the then most current "Estoppel Certificate" form published by the American Industrial Real Estate Association, plus such additional information, confirmation and/or statements as may be reasonably requested by the Requesting

reary. (b) If the Responding Party shall fail to execute or deliver the Estoppel Cartificate within such 10 day period, the Requesting Party may execute an Estoppel Certificate stating that: (i) the Lasse is in full force and effect without modification except as may be represented by the Requesting Party, (ii) there are no uncured defaults in the Requesting Party's performance, and (iii) if Lessor is the Requesting Party, not more than one month's rent has been paid in advance. Prospective purchasers and encumbrancers may rely upon the Requesting Party's Estoppel Cartificate, and the Responding Party shall be estopped from denying the tot the funct exclusion of Cartificate (Cartificate) and (Cartifi the truth of the facts contained in said Certificate.

(c) If Lessor desired to finance, refinance, or sell the Premises, or any part thereof, Lessee and all Guarantors shall deliver to any potential (ender or purchaser designated by Lessor such financial statements as may be reasonably required by such lender or purchaser, including but not ilmited to Lessee's financial statements for the past 3 years. All such financial statements shall be received by Lessor and such lender or purchaser in confidence and shall be used only for the purposes herein set forth.

17. Definition of Lessor. The term "Lessor" as used herein shall mean the owner or owners at the time in question of the fee title to the Premises, or, if this is a sublease, of the Lesse's interest in the prior lesse. In the event of a transfer of Lessor's title or interest in the Premises or this Lesse, Lessor shall deliver to the transferse or easignee (in cash or by credit) any unused Security Deposit held by Lessor. Except as provided in Paragraph 15, upon such transfer or assignment and delivery of the Security Deposit, as aforesaid, the prior Lessor shall be relieved of all liability with respect to the obligations and/or covenants under this Lesse thereafter to be performed by the Lessor. Subject to the foregoing, the obligations and/or covenants under this Lesse thereafter to be performed by the Lessor shall be above, and subject to the provisions of Paragraph 20 below, the original Lessor under this Lesse, and all subject on the Lessor's interest in this Lesse shall remain liable and responsible with regard to the potential duties and liabilities of Lessor pertaining to Hazardous Substances as outsined in Paragraph 5.2 above.

16. Severability. The invalidity of any provision of this Lease, as determined by a court of competent jurisdiction, shall in no way affect the validity of any other provision hereof.

Days, Unless otherwise specifically indicated to the contrary, the word "days" as used in this Lease shall mean and refer to calendar days. 19.

20. Limitation on Liability. Subject to the provisions of Paragraph 17 above, the obligations of Lessor under this Lesses shall not constitute personal obligations of Lessor, the individual partners of Lessor or its or their individual partners, directors, officers or shareholders, and Lesses shall look to the Premises, and to no other essets of Lessor, for the satisfaction of any fability of Lessor with respect to this Lesses, and shall not seek recourse against the individual partners of Lessor, or its or their individual partners, directors, or its or their individual partners, directors, or its or their individual partners, directors, officers or shareholders, and shall not seek recourse against the individual partners of Lessor, or its or their individual partners, directors, officers or shareholders, or any of their personal assets for such satisfaction.

Time of Essence. Time is of the essence with respect to the performance of all obligations to be performed or observed by the Parties under this Lease. 21.

No Prior or Other Agreements; Broker Disclaimer. This Lease contains all agreements between the Parties with respect to any matter mentioned 22. No Prior or Unter Agreements; broker uschamer. This Lease contains all agreements between the Parties with respect to any matter methodied herein, and no other prior or contemporaneous agreement or understanding shall be affactive. Lessor and Lease each represents and warrants to the Brokers that it has made, and is relying solely upon, its own investigation as to the nature, quality, character and financial responsibility of the other Party to this Lease end as to the nature, quality, character and financial responsibility of the other Party to this Lease end as to the nature, quality, character and financial responsibility of the other Party to this Lease end as to the nature, quality, character and financial responsibility of the other Party to this Lease end as to the nature, quality, character and financial responsibility of the other Party to this Lease in the responsibility with respect to any default or breach hereof by either Party. The liability (including court costs and attorneys' fees), of any Broker with respect to negotiation, execution, delivery or performance by either Lessor or Lease under this Lease or any amendment or modification hereid shall be limited to an amount up to the fee received by such Broker Broker. Brokers that the use motion is further finality function and the table of the Brokers and the table of the Brokers and the second of the Brokers and the seco however, that the foregoing limitation on each Broker's liability shall not be applicable to any gross negligence or willful misconduct of such Broker.

23. Notices.

23. Notices. 23.1 Notice Requirements. All notices required or permitted by this Lease or applicable law shall be in writing and may be delivered in person (by hand or by couler) or may be earl by regular, certified or registered mail or U.S. Postal Service Express Mail, with postage prepaid, or by facalmile transmission, and shall be deemed sufficiently given if served in a manner specified in this Paragraph 23. The addresses noted adjacent to a Party's signature on this Lease shall be that Party's address for delivery or mailing of notices. Either Party may by written notice to the other specified in this Paragraph 23. The addresses noted adjacent to a Party's signature on this Lease shall be that Party's address for delivery or mailing of notices. Either Party may by written notice to the other specified in this paragraph 23. The addresses noted adjacent to a Party's signature on this Lease shall be that Party's address for delivery or mailing of notices. Either Party may by written notice to the other specified all notices to Leasor shall be concurrently transmitted to such addresses as Leases may from time to time hersafter designate in writing. 23.2 Date of Notice. Any notice sent by registered or certified mail, return receipt requested, shall be deemed given on the date of delivery shown on the receipt card, or if no delivery date is shown, the postmark thereon. If sent by regultar mail the notice shall be deemed given 46 hours after the same is the delivered by United States Express Mail or overlight courier that quarantee next devide weldivery date is shown, the postmark thereon. If sent by facalmine there there addressed as required herein and melled with postage prepaid. Notices delivered by United States Express Mail or overlight tourier that quarantee next devide delivery date devide weldivery or mail. The same is the there is an early is also delivered via delivery or mail. If notice is received on a Saturday, Sunday or legal holiday, it shell be deemed received on the next business day.

24. Waavers. No waiver by Lessor of the Default or Breach of any term, covenant or condition hereof by Lessee, shall be deemed a waiver of any other term, covenant or condition hereof, or of any subsequent Default or Breach by Lessee of the same or of any dither term, covenant or condition hereof, or of any subsequent Default or Breach by Lessee of the same or of any dither term, covenant or condition hereof. Lessor's consent to, or approval of, any subsequent to tender unnecessary the cotening of Lesser's consent to, or approval of, any subsequent or similar act by Lessee, or be construed as the basis of an estoppel to enforce the provision or provisions of this Lesse reguling such consent. The acceptance of Rent by Lessee, notwithatanding any qualifying statements or conditions made by Lesses in connection therewith, which such attaments and/or conditions shall be of no force or effect whatsoever unless specifically agreed to in writing by Lesser at or before the time of deposit of such payment. Walvers. No waiver by Lessor of the Default or Breach of any term, covenant or condition hereof by Lessee, shall be deemed a waiver of any other term,

25. Disclosures Regarding The Nature of a Real Estate Agency Relationship. (a) When entering into a discussion with a real estate egent regarding a real estate transaction, a Lessor or Lessee should from the outset understand what type of agency relationship or representation it has with the agent or agents in the transaction. Lessor and Lessee acknowledge being advised by the Brokers in this transaction, as follows:

the provers in this transaction, as recover: ()) Lessor's Agent. A Lessor's egent under a listing agreement with the Lessor acts as the agent for the Lessor only. A Lessor's agent or subagent has the following affirmative obligations: To the Lessor. A fiduciary duty of utmost care, integrity, honesty, and loyally in dealings with the Lessor. To the Lessoe and the Lessor: a. Diligent exercises of reasonable skills and care in performance of the agent's duties. b. A duty of honest and fair dealing and good faith, c. A duty to disclose all facts known to the agent materially affecting the value or desirability of the property that are not known to, or within the diligent attention and observation of, the Parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve the adjust of the parties. An agent is not obligated to reveal to either Party any confidential information obtained from the other Party which does not involve

and observation or, the Farces, or agents into comparison of the agent can agree to act as agent for the Lessee only. In these situations, the agent is not the Lessor's agent, (i) Lesse's Agent. An agent can agree to act as agent for the Lesse only. In these situations, the agent acting only for a Lesses has the following affirmative obligations. To the Lesses' A fiducity duty of utmance can agree to act as agent's duties. b. A duty of honest and fair dealing with the Lesses and the Lessor's a Diligent exercise of reasonable skills and care in performance of the agent's duties. b. A duty of honest and fair dealing and good faith. c. A duty to

EL

Pade 9 of 12

initials

duties set forth abov

P.013/019 (FAX)209 667 1251

> within the diligent attention and observation Party which does not involve the affirmative

Agent Representing Both Lessor and Lessee. A real estate egent, either acting directly or through one or more associate licenses, 680 (ii) <u>Agent Representing Join Lesson and the Lesson and Lesson that the Lesson that the Lesson and Lesson or Lesson or Lesson the transaction do the the the understanding of the transaction. A real estate agent is a person qualified to advise about real estate. If legal or tax advice is desired, the advice and the Lesson and Less</u> consult a competent professional.

consult a competent professional. (b) Brokers have no responsibility with names to any default or breach hereof by either Party. The liability (including court costs and attorneys' fees), of any Broker with respect to any breach of duty, error or ormission relating to this Lesse shall not exceed the fee received by such Broker pursuant to this Lesse; provided, however, that the foregoing limitation on each Broker's liability shall not be applicable to any gross negligence or willful misconduct of such Broker. (c) Buyer and Safer agree to identify to Brokers as "Confidential" any communication or information given Brokers that is considered by such Party (c) to be confidential.

25. No Right To Holdover. Lessee has no right to retain possession of the Premises or any part thereof beyond the expiration or termination of this Lesse. In the event that Lessee holds over, then the Base Rent shall be increased to 150% of the Base Rent applicable immediately preceding the expiration or termination. Nothing contained herein shall be construed as consent by Lessor to any holding over by Lessee.

Cumulative Remedies. No remedy or election hereunder shall be deemed exclusive but shall, wherever possible, be cumulative with all other remedies at law or in equity.

28. Covenants and Conditions; Construction of Agreement. All provisions of this Lease to be observed or performed by Lessee are both covenants and conditions. In construing this Lease, all headings and titles are for the convenience of the Parties only and shall not be considered a part of this Lease. Whenever required by the context, the singular shall include the plugal and vice versa. This Lease shall not be construed as if prepared by one of the Parties, but rather according to its fair meaning as a whole, as if both Parties had prepared it.

29. Binding Effect; Choice of Law. This Lease shall be binding upon the parties, their personal representatives, successors and assigns and be governed by the laws of the State in which the Premises are located. Any fitigation between the Parties hareto concerning this Lease shall be initiated in the county in which the Premises are located.

Subordination; Attomment; Non-Disturbance. 30.

30. Subordination; Attornment; Non-Disturbance. 30.1 Subordination: Attornment; Non-Disturbance. 30.1 Subordination. This Lease and any Option granted hereby shall be subject and subordinate to any ground lease, mortgage, deed of trust, or other hypothecation or security device (collectively, "Security Device"), now or hereafter placed upon the Premises, to any and all advances made on the security thereof, and to all renewals, modifications, and extensions thereof. Lesses agrees that the holders of any such Security Devices (in this Lease together referred to as "Lender") shall have no liability or obligation to perform any of the obligations of Lessor under this Lease. Any Lender may elect to have this Lease and/or any Option granted hereby superior to the lien of its Security Device by giving written notice thereof to Lessee, whereupon this Lease and such Options shall be deemed prior to such Security Device, notwithstanding the relative dates of the documentation or recordation thereof. 30.2 Attornment. Subject to the non-disturbance provisions of Paragraph 30.3, Lessee agrees to attorn to a Lender or any other party who acquires ownership of the Premises by reason of a foreclosure of a Security Device, and that in the event of such foreclosure, such new owner shall not: (a) be liable for any act or amission of any prior lessor, (c) be bound by prepayment of more than one month's rent, or (d) be liable for the ratum of any security deposit paid to env prior lessor.

lessor.

lessor. 30.3 Non-Disturbance. With respect to Security Devices entered into by Lessor after the execution of this Lesse, Lesser's subordination of this Lesse, here a shall be subject to receiving a commercially reasonable non-disturbance agreement (a "Non-Disturbance Agreement") from the Lender which Non-Disturbance Agreement provides that Lesser's possession of the Premises, and this Lesse, including any options to extend the term hereof, will not be disturbed so long as Lessee is not in Breach hereof and attorns to the record owner of the Premises. Further, within 60 days after the execution of this Lesser, because the execution of this Lesser, including any options to extend the term hereof, will not be disturbed so long as Lessee is not in Breach hereof and attorns to the record owner of the Premises. Further, within 60 days after the execution of this Lesser, Lessor shall use its commercially reasonable efforts to obtain a Non-Disturbance Agreement within said 60 days, then Lesser work that Lesser's option, directly contact Lender and attempt to hegotiate for the execution and delivery of a Non-Disturbance Agreement. 30.4 Self-Executing. The agreements contained in this Paragraph 30 shall be effective without the execution of any further documents; provided, and the remember and attempt to report the premise of a context part of an opported by a self effective without the securition of any further documents; provided, and the remember and lessor that account of a part and attempt to an opported by a self effective without the securition of any further documents; provided, and the remember and attempt to report the securition and delivery of a self effective without the the premise and Lesser and Lesser that account in a provided is a self discorting of the premise and Lesser and Lesser the approximation of any further documents; provided, and the premise account of a self effective without the previous of the premise and Lesser that account is a previous of the premise account of the premise account of

however, that, upon written request from Lessor or a Lender in connection with a sale, financing or refinancing of the Premitees, Lessee and Lessor shall execute such further writings as may be reasonably required to separately document any subordination, attornment and/or Non-Disturbance Agreement provided for herein.

31. Attorneys' Rees. If any Party or Broker brings an action or proceeding involving the Premises whether founded in tort, contract or equily, or to declare rights hereunder, the Prevailing Party (as hereafter defined) in any such proceeding, action, or appeal thereon, shall be entitled to reasonable attorneys' fees. Such fees may be awarded in the same suit or recovered in a separate suit, whether or not such action or proceeding is pursued to decision or judgment. The term, "Prevailing Party' shall include, without limitation, a Party or Broker who substantially obtains or defense. The attorneys' fees award shall not be computed in a computed in a computed in a secondance with any court fee schedule, but shall be such as to fully reimburee all attorneys' fees reasonably incurred. In addition, Lessor shall be entitled to attorneys' fees, costs and expenses incurred in the preparation and service of notices of Default and consultations the rewiting action in the attorneys' fees, costs and expenses incurred in the preparation and service of notices of Default and consultations in connection therewith, whether on to a legal action, party and the preparation and service of notices of Default and consultations in connection therewith, whether on the action is a terment in convecting the other Party or Broker of the schedule, but shall be entitled to attorneys' fees, costs and expenses incurred in the preparation and service of notices of Default and consultations in connection therewith, whether or not a legal action is a case nable minimum action in action in the preparation and service of notices of Default and consultations in connection therewith, whether or not a legal action is a case nable minimum action in the preparation and service of next the default and consultations in connection therewith, whether or not a legal action is a case nable minimum action in the preparation and service of next the default and consultations in the preparation and service of next the service action action action in the prepar action is subsequently commenced in connection with such Default or resulting Breach (\$200 is a reasonable minimum per occurrence for such services and consultation).

32. Lessor's Access; Showing Premises; Repairs. Lessor and Lessor's agents shall have the right to enter the Premises at any time, in the case of an emergency, and otherwise at reasonable times for the purpose of showing the same to prospective purchasers, lenders, or tunants, and making such alterationa, repairs, improvements or additiona to the Premises as Lessor may deem necessary. All such activities shall be without abatement of rent or liability to Lessee. Lessor may at any time place on the Premises any ordinary "For Sale" signs and Lessor may during the last 6 months of the term hereof place on the Premises any ordinary "For Lesse" signs.

33. Auctions. Lesses shall not conduct, nor permit to be conducted, any auction upon the Premises without Lessor's prior written consent. Lessor shall not be obligated to exercise any standard of reasonableness in determining whether to permit an auction.

Signs. Except for ordinary "For Sublesse" signs which may be placed only on the Premises, Lessee shall not place any sign upon the Project without r's prior written consent. All signs must comply with all Applicable Requirements.

35. Termination; Merger. Unless specifically started otherwise in writing by Lessor, the voluntary or other summer of this Lesse by Lesse, the mutual termination or cancellation hereof, or a termination hereof by Lessor for Breach by Lessee, shall automatically terminate any sublease or lesser estate in the Premises; provided, however, that Lessor may elect to continue any one or all existing subtenancies. Lessor's failure within 10 days following any such event to elect to the contrary by written notice to the holder of any auch lasser interest, shall constitute Lessor's election to have such event constitute the termination of auch

38. Consents. Except as otherwise provided herein, wherever in this Lease the consent of a Party is required to an act by or for the other Party, such consent shall not be unreasonably withheld or delayed. Lessor's actual reasonable costs and expenses (including but not limited to architects, attorneys, engineers' and other consultants' fees) incurred in the consideration of, or response to, a request by Lesser consent, including but not limited to architects, attorneys, engineers' and subgrantint, a subletting or the presence or use of a Hazardous Substance, shall be paid by Lesser consent, including but not limited to accepte and supporting documentation therefor. Lessor's consent to any act, assignment or subletting shall not constitute an acknowledgment that no Default or Breach by Lessee of this Lesse exists, nor shall such consent be deemed a waiver of any then edisting Default or Breach, except as may be otherwise specifically stated in writing by Lessor at the time of consent. The failure to specify herein any particular condition to Lessor's consent shall not preclude the Imposition by Lessor at the time of such active or dure conditione as are then reasonable with reference to the particular matter for which consent is being given. In the event that effore Party disagrees with any determination made by the other hereunder and reasonably requests the reasons for such determination, the determining party shall furnish its reasons in writing and in reasonable detail within 10 business days following such request.

Guarantor

F. L .

Page 10 of 12

84

(FAX)209 667 1251 P.014/019

37.1 Execution. Suarantors, if any, shall each execute a guaranty in the form most recently including the American Industrial Real Estate Association, and each such Guarantor shall have the same obligations as Lessee under this Lease. 37.2 Default. It shall constitute a Default of the Lessee if any Guarantor falls or refuses, upon request to provide: (a) evidence of the execution of the guaranty, including the authority of the party signing on Guarantor's behalf to obligate Guarantor, and in the case of a corporate Guarantor, a certified copy of a resolution of its board of directors authorizing the making of such guaranty, (b) current financial statements, (c) an Estoppei Certificate, or (d) written confirmation that the guaranty is still in effect.

38. Quiet Possession. Subject to payment by Lesses of the Rent and performance of all of the covenants, conditions and provisions on Lesses's part to be observed and performed under this Lesse, Lesses shall have quiet possession and quiet enjoyment of the Premises during the term hereof.

39

39. Options. If Lessee is granted an option, as defined below, then the following provisions shall apply. 39.1 Definition. "Option" shall mean: (a) the right to extend the term of or renew this Lesse or to extend or renew any lesse that Lessee has on other property of Lessor; (b) the right of first refusal or first offer to lesse either the Premisee or other property of Lessor; (c) the right to purchase or the right of first

other property of Lessor; (b) the right of miss relusal of the solar to lesse entrer the Premises of other property of Lessor; (c) the right of miss relusal or ther property of Lessor. 39.2 Options Personal To Original Lessee. Any Option granted to Lessee in this Lesse is personal to the original Lessee, and cannot be assigned or exercised by anyone other than said original Lessee and only while the original Lessee is in full possession of the Premises and, if requested by Lessor, with Lessee certifying that Lessee has no intention of thereafter assigning or subletting. 39.3 Multiple Options. In the event that Lessee has any multiple Options to extend or renew this Lesse, a later Option cannot be exercised unless

the prior Options have been validly exercised. 39.4 Effect of Default on Options.

(a) Lessee shall have no right to exercise an Option: (i) during the period commencing with the giving of any notice of Default and continuing until said Default is cured, (ii) during the period of time any Rent is unpaid (without regard to whether notice thereof is given Lessee), (ii) during the time Lessee is in Breach of this Lesse, or (iv) in the event that Lessee has been given 3 or more notices of separate Default, whether or not the Defaults are cured, during the 12 month period immediately preceding the exercise of the Option.

(b) The period of time within which an Option may be exercised shall not be extended or enlarged by reason of Lessee's inability to exercise an Option because of the provisions of Paragraph 39.4(a).

(c) An Option shall terminate and be of no further force or effect, notwithstanding Lassee's due and timely exercise of the Option, if, after such exercise and prior to the commencement of the extended term, (i) Lessee fails to pay Rent for a period of 30 days after such Rent becomes due (without any necessity of Lessor to give notice thereof), (ii) Lesser gives to Lassee 3 or more notices of separate Default during any 12 month period, whether or not the Defaults are cured, or (iii) if Lassee commits a Breach of this Lease.

Security Measures. Lesses hereby acknowledges that the Rent payable to Lessor hereunder does not include the cost of guard service or other security sures, and that Lessor shall have no obligation whatsoever to provide same. Lesses assumes all responsibility for the protection of the Fremises, Lesses, its 40. mea agents and invitees and their property from the acts of third parties

41. Reservations. Lessor reserves the right: (i) to grant, without the consent or joinder of Lessee, such easements, rights and dedications that Lessor deems necessary, (ii) to cause the recordation of parcel maps and restrictions, and (iii) to create and/or install new utility receways, so long as such easements, rights, dedications, maps, restrictions, and utility naceways do not unreasonably interfere with the use of the Premises by Lessee. Lessee agrees to sign any documents reasonably requested by Lesser to effectuate such rights.

42. Performance Under Protest. If at any time a dispute shall arise as to any amount or sum of money to be paid by one Party to the other under the provisions hareof, the Party against whom the obligation to pay the money is asserted shall have the right to make payment "under protest" and such payment shall not be regarded as a voluntary payment and there shall survive the right on the part of said Party to institute suit for recovery of such sum. If it shall be adjudged that there was no legal obligation on the part of said Party to pay such sum or any part thereof, said Party shall be entitled to recover such sum or so much thereof as it there are included to recover such sum or so much thereof as it was not legally required to pay.

43. Authority. If either Party hereto is a corporation, trust, ilmited liability company, partnership, or similar entity, each individual executing this Lease on behalf of such entity represents and warrants that he or she is duly authorized to execute and deliver this Lease on its behalf. Each party shall, within 30 days after request, deliver to the other party satisfactory evidence of such authority.

Conflict. Any conflict between the printed provisions of this Lease and the typewritten or handwritten provisions shall be controlled by the typewritten or handwritten provisions,

45. Offer. Preparation of this Lease by either party or their agent and submission of same to the other Party shall not be deemed an offer to lease to the other Party. This Lease is not Intended to be binding until executed and delivered by all Parties hereto.

46. Amendments. This Lesse may be modified only in writing, signed by the Parties in interest at the time of the modification. As long as they do not materially change Lesses's obligations hereunder, Lesse agrees to make such reasonable non-monetary modifications to this Lesse as may be reasonably required by a Lender in connection with the obtaining of normal financing or refinancing of the Premises.

47. Multiple Parties. If more than one person or entity is named herein as either Lessor or Lessee, such multiple Parties shall have joint and several responsibility to comply with the terms of this Lesse.

48 Waiver of Jury Trial, The Parties hereby waive their respective rights to trial by jury in any action or proceeding involving the Property or arising out of this Agreement.

Mediation and Arbitration of Disputes. An Addendum requiring the Mediation and/or the Arbitration of all disputes between the Parties and/or Brokers 49. arising out of this Lease D is D is not attached to this Lease.

LESSOR AND LESSEE HAVE CAREFULLY READ AND REVIEWED THIS LEASE AND EACH TERM AND PROVISION CONTAINED HEREIN, AND BY THE EXECUTION OF THIS LEASE SHOW THEIR INFORMED AND VOLUNTARY CONSENT THERETO. THE PARTIES HERBY AGREE THAT, AT THE THIS LEASE IS EXECUTED, THE TERMS OF THIS LEASE ARE COMMERCIALLY REASONABLE AND EFFECTUATE THE INTENT AND PURPOSE OF LESSOR AND LESSEE WITH RESPECT TO THE PREMISES.

ATTENTION; NO REPRESENTATION OR RECOMMENDATION IS MADE BY THE AMERICAN INDUSTRIAL REAL ESTATE ASSOCIATION OR BY ANY BROKER AS TO THE LEGAL SUFFICIENCY, LEGAL EFFECT, OR TAX CONSEQUENCES OF THIS LEASE OR THE TRANSACTION TO WHICH IT

BROKER AS TO THE LEGAL SUFFICIENCY, LEGAL EFFECT, OR TAX CONSEQUENCES OF THIS LEASE OR THE TRANSACTION TO WHICH TH RELATES. THE PARTIES ARE URGED TO: 1. SEEK ADVICE OF COUNSEL AS TO THE LEGAL AND TAX CONSEQUENCES OF THIS LEASE. 2. RETAIN APPROPRIATE CONSULTANTS TO REVIEW AND INVESTIGATE THE CONDITION OF THE PREMISES. SAID INVESTIGATION SHOULD INCLUDE BUT NOT BE LIMITED TO: THE POSSIBLE PRESENCE OF HAZARDOUS SUBSTANCES, THE ZONING OF THE PREMISES, THE STRUCTURAL INTEGRITY, THE CONDITION OF THE ROOF AND OPERATING SYSTEMS, COMPLIANCE WITH THE AMERICANS WITH DISABILITIES ACT AND THE SUITABILITY OF THE PREMISES FOR LESSEE'S INTENDED USE.

WARNING: IF THE PREMISES ARE LOCATED IN A STATE OTHER THAN CALIFORNIA, CERTAIN PROVISIONS OF THE LEASE MAY NEED TO BE REVISED TO COMPLY WITH THE LAWS OF THE STATE IN WHICH THE PREMISES ARE LOCATED. The parties hereto have executed this Lease at the place and on the dates specified above their respective signatures.

By LESSEE:

lock Executed at: February 2001 on: 26,

By LESSOR: L & M Industrial Park

CA Executed at: Harmand 2001 march on:

Construction Craft Training Center

Page 11 of 12

07/27/2016 15:14 ACME ELECTRIC	(FAX)209 667 1251 P.O	15/019
By:	By: Chris Floet	
Name Printed: Frank Lever	Name Printed:	
Title:	Title:	
By:	Ву:	
Name Printed:	Name Printed:	
Title:	Title:	
Address: P. O. Box 677	Addrese:	
Turlock, CA 95381		
Telephone: (209) 667-2851	Telephone: (510) 785-2282	
Faosimile: ()	Facelmile: (510) 785-9136	
Federal ID No.	Federal ID No.	

These forms are often modified to meet changing requirements of law and needs of the industry. Always write or call to make sure you are utilizing the most current form: American Industrial Real Estate Association, 700 South Flower Street, Suite 600, Los Angeles, CA 90017. (213) 687-8777.

> (c)Copyright 1998 By American Industrial Real Estate Association. All rights reserved. No part of these works may be reproduced in any form without permission in writing.



initials

P.018/019



STANDARD LEASE ADDENDUM

Dated

February 23, 2001

By and Between (Lessor) L & M Investments

(Lessee) Construction Craft Training Center

Address of Premises: 2480 Acme Ct., Turlock, CA

Paragraph 50

RENT ADJUSTMENTS:

The monthly rent for each month of the adjustment period(s) specified below shall be increased using the method(s) indicated below: (Check Method(s) to be Used and Fill In Appropriately)

I. Cost of Living Adjustment(s) (COLA)

e. On (Fill in COLA Dates): April 1 of each year of lease

the Base Rent shall be adjusted by the change, if any, from the Base Month specified below, in the Consumer Price Index of the Bureau of Labor Statistics of the U.S. Department of Labor for (select one): Ed CPI W (Urpan Wage Earners and Cierical Workers) or 🛄 CPI U (All Urban Consumers), for (Fill In Urban Area): _____ San Francisco - Oakland - San Jose . All Itema

(1982-1984 = 100), herein referred to as "CPI".

The monthly rent payable in accordance with paragraph A.I.a. of this Addendum shall be calculated as follows: the Base Rent set forth in paragraph 1.5 of the attached Lesse, shall be multiplied by a fraction the numerator of which shall be the CPI of the calendar month two months prior to the month(s) specified in paragraph A.La. above during which the adjustment is to take effect, and the denominator of which shall be the CPI of the calendar month which is two months prior to (select one): If the first month of the term of this Lease as set forth in paragraph 1.3 ("Base Month") or II (Fill in Other "Base Month"):

. The sum so calculated shall constitute the new monthly rent hereunder, but in no event, shall any such new monthly rank be less than the rent payable for the month immediately preceding the rent adjustment.

c. In the event the complication and/or publication of the CPI shall be transferred to any other governmental department or bureau or agency or shall be discontinued, then the index most nearly the same as the CPI shall be used to make such calculation. In the event that the Parties cannot agree on such alternative index, then the matter shall be submitted for decision to the American Arbitration Association in accordance with the then rules of said Association and the decision of the arbitrators shall be binding upon the parties. The cost of said Arbitration shall be paid equally by the Parties.

II. Market Rental Value Adjustment(s) (MRV)

a. On (Fill in MRV Adjustment Date(s):

the Base Rent shall be adjusted to the "Market Rental Value" of the property as follows:

1) Four months prior to each Market Rental Value Adjustment Date described above, the Parties shall attempt to agree upon what the new MRV will be on the adjustment date. If agreement cannot be reached within thirty days, then:

(a) Lessor and Lessor shall immediately appoint a mutually acceptable appraiser or broker to establish the new MRV within the next thirty days. Any associated costs will be splik equally between the Parties, or

(b) Both Lesser and Lesses shall each interediately make a reasonable determination of the MRV and submit such determination, in writing, to arbitration in accordance with the following provisions:

(i) Within fifteen days thereafter, Lessor and Lessee shall each select an 🗔 appraiser or 🗔 broker ("Consultant" - check one) of their choice to act as an arbitrator. The two arbitrators so appointed shall immediately select a third mutually acceptable Consultant to act as a third arbitrator.

(ii) The Three arbitrators shall within thirty days of the appointment of the third arbitrator reach a decision as to what the actual MRV for the Premises is, and whether Lessor's or Lessee's submitted MRV is the closest thereto. The decision of a majority of the arbitrators shall be binding on the Parties. The submitted MRV which is determined to be the closest to the actual MRV anal thereefter be used by the Parties.

(iii) if either of the Parties fails to appoint an arbitrator within the specified fifteen days, the arbitrator timely appointed by one of them shall reach a decision on his or her own, and said decision shall be binding on the Parties

(iv) The entire cost of such arbitration shall be paid by the party whose submitted MRV is not selected, ie. the one that is NOT the closest to the actual MRV.

2) Notwithstanding the foregoing, the naw MRV shall not be less than the rent payable for the month immediately preceding the rent adjustment.

b. Upon the establishment of each New Market Rental Value:
 1) the new MRV will become the new "Base Rent" for the purpose of calculating any further Adjustments, and
 2) the first month of each Market Rental Value term shall become the new "Base Month" for the purpose of calculating any further Adjustments.

initials: FL.

RENT ADJUSTMENTS Page 1 of 2

For this form, write: American Industrial Real Estate Association, 700 S. Flower Street, Suite 600, Los Angeles, Calif. 90917 ©1997 - American Industrial Real Estate Association REVISED POL PORM RA-2-3/97E

(FAX)209 667 1251

.

1. 1. P.019/019

i ш.	Fixed	Rental Ad	ustmen.
------	-------	-----------	---------

07/27/2016 15:15 ACME ELI	ECTRIC
	RA)
On (Fill in FRA Adjustment Date(s)):	lowing amounts on the dates set forth below: The New Bass Rent shall be:
	_ \$
	• • •
	- \$ <u></u>

B. NOTICE: Unless specified otherwise herein, notice of any such adjustments, other than Fixed Rental Adjustments, shell be made as specified in paragraph 23 of the Leese.

.

C. BROKER'S FEE: The Brokers specified in paragraph 1.10 shall be paid a Brokerage Fee for each adjustment specified above in accordance with paragraph 16 of the Lease.

initiale: AA

RENT ADJUSTMENTS Page 2 of 2

For this form, write: American industrial Real Estate Association, 700 S. Flower Street, Suite 600, Los Angeles, Calif. 90617 ©1997 - American industrial Real Estate Association REVISED FORM RA-2-3/87E

07/27/2016 15:14 ACME ELECTRIC

(FAX)209 667 1251

P.016/019



Dated

VE FI

February 23, 2001

By and Between (Lessor) L & M Investments

(Lessee) Construction Craft Training Center

Address of Premises:2480 Acme Ct., Turlock, CA 93581

Peregraph 51

A. OPTION(6) TO EXTEND:

Lessor hereby grants to Lessee the option to extend the term of this Lesse for _____1 additional _____24 ____ month period(s) commencing when the prior term expires upon each and ell of the following terms and conditions:

(i) In order to exercise an option to extend, Lessee must give written notice of such election to Lessor and Lessor must receive the same at least <u>3</u> but not more than <u>6</u> months prior to the date that the option period would commance, time being of the essence. If proper notification of the

exercise of an option is not given and/or received, such option shall automatically expire. Options (if there are more than one) may only be exercised consecutively.

(ii) The provisions of persograph 39, including those relating to Lessee's Default set forth in paragraph 39.4 of this Lesse, are conditions of this Option.

(iii) Except for the provisione of this Lease graining an option or options to extend the term, all of the terms and conditions of this Lease except where specifically modified by this option shall apply.

(iv) This Option is personal to the original Lessee, and cannot be assigned or exercised by anyone other than said original Lessee and only while the original Lessee is in full possession of the Premises and without the Intention of thereafter assigning or subletting.

(v) The monthly rent for each month of the option period shall be calculated as follows, using the method(s) indicated below: (Check Method(s) to be Used and Fill in Appropriately)

I. Cost of Living Adjustment(s) (COLA)

e. On (Fill in COLA Dates): April 1 of each year

the Base Rent shall be adjusted by the change, if any, from the Base Month specified below, in the Consumer Price Index of the Bureau of Labor Statistics of the U.S. Department of Labor for (select one); 🖸 CPI W (Urban Wage Esners and Clarical Workers) or 🗋 CPI U (All Urban Consumers), for (Fill in Urban Area); San Francisco – Oakland – San Jose All Items (1982-1984 = 100), herein referred to as "CPI".

b. The monthly rent payable in accordance with paragraph A.I.a. of this Addendum shall be calculated as follows: the Base Rent set forth in paragraph 1.5 of the attached Lease, shall be multiplied by a fraction the numerator of which shall be the CPI of the calendar month two months prior to the month(s) spacified in paragraph A.I.a. ebove during which the adjustment is to take effect, and the denominator of which shall be the CPI of the calendar month which is two months prior to (select one): If the first month of the term of this Lease as set forth in paragraph 1.3 ("Base Month") or D (Fill in Other "Base Month"):

. The sum so calculated shall constitute the new monthly rent hereunder, but in no event, shall any such new monthly rent be less than the rant psyable for the month immediately praceding the rent adjustment.

c. In the event the compliation and/or publication of the CPI shall be transferred to any other governmental department or bureau or agency or shall be discontinued, then the index most nearly the same as the CPI shall be used to make such calculation. In the event that the Parties cannot agree on such alternative index, then the matter shall be submitted for decision to the American Arbitration Association in accordance with the then rules of said Association and the decision of the arbitrators shall be binding upon the parties. The cost of said Arbitration shall be paid equally by the Parties.

II. Market Rental Value Adjustment(s) (MRV)

a. On (Fill in MRV Adjustment Date(s))

the Base Rent shall be edjusted to the "Market Rental Value" of the property as follows:

1) Four months prior to each Market Rental Value Adjustment Date described above, the Parties shall attempt to agree upon what the new MRV will be on the adjustment date. If agreement cannot be reached, within thirty days, then:

(a) Lessor and Lessee shall immediately appoint a mutually acceptable appraiser or broker to establish the new MRV within the next thirty days. Any associated costs will be split equally between the Parties, or

(b) Both Lessor and Lessoe shell each immediately make a reasonable determination of the MRV and submit such determination, in writing, to arbitration in accordance with the following provisions:

(i) Within fifteen days thereafter, Lesson and Lessee shall each select an 🗔 appraiser or 🗆 broker ("Consustant" - check one) of their choice to act as an arbitrator. The two arbitrators so appointed shall immediately select a third mutually acceptable Consultant to act as a third arbitrator.

initials: Fb

initiels/

©1997 - American Industrial Real Estate Association

Page 1 of 2 REVISED

FORM OE-2-3/97E

(FAX)209 667 1251 P.017/019

(iii) If either of the Parties fails to appoint an arbitrator within the specified fifteen days, the arbitrator timely appointed by one of them shall reach a decision on his or her own, and seld decision shall be binding on the Parties.

(N) The entire cost of such arbitration shall be paid by the party whose submitted MRV is not selected, ie. the one that is NOT the closest to the

actual MRV.

-.

2) Notwithstanding the foregoing, the new MRV shall not be less than the rent payable for the month immediately preceding the rent adjustment.

b. Upon the establishment of each New Market Rental Value:

the new MRV will become the new "Base Rent" for the purpose of calculating any further Adjustments, and
 the first month of each Market Rental Value term shall become the new "Base Month" for the purpose of calculating any further Adjustments.

U. Fixed Rental Adjustment(s) (FRA)
The Base Rent shall be increased to the following amounts on the dates set forth below:

On (Fili in FRA Adjustment Date(s));	The N	lew Base Rent shall be:
	\$	\$
	5	
	5	
	\$,

B. NOTICE:

Unloss specified otherwise herein, notice of any rental adjustments, other than Fixed Rental Adjustments, shell be made as specified in paragraph 23 of the Lease.

C. BROKER'S FEE:

The Brokers specified in paragraph 1.10 shall be paid a Brokerage Fee for each adjustment specified above in accordance with paragraph 15 of the Lease.

Initials:

FL

@1997 - American Industrial Real Estate Association

Page 2 of 2 **REVISED**



FORM OE-2-3/97E

APPENDIX VI TO DONATION AGREEMENT

Assumed Liabilities

- 1. Liabilities under the 26200 Lease
- 2. Liabilities under the 26206 Lease
- 3. Liabilities under the Turlock Lease
- 4. [Accounts Payable as listed on attached list]

Construction Craft Training Center Balance Sheet June 30,2016

June 30,2016				
	Assets			
Current Assets: Cash Accounts Receivable Less: Reserve for Bad Debts Merchandise Inventory Prepaid Expenses Notes Receivable Total Current Assets	22580 \$51,875 2,594	\$22,580 49,281 4,300 11,512 0	\$87,673	
<i>Fixed Assets:</i> Vehicles Less: Accumulated Depreciation	0 0	0		
Furniture and Fixtures Less: Accumulated Depreciation	22,500 2,250	20,250		
Equipment Less: Accumulated Depreciation	135,000 13,500	121,500		
Buildings Less: Accumulated Depreciation	0	0		
Land Total Fixed Assets		0	141,750	
<i>Other Assets:</i> Goodwill <i>Total Other Assets</i>		0	0	
Total Assets		=	\$229,423	

Liabilities and Capital

Current Liabilities:		
Accounts Payable	\$8,300	
2410 - Credit Card #8885	\$368	
2420- Credit Card #0339	\$11,401	
Payroll Liabilities	5,212	
Payroll wages	34,829	
Unearned Revenues	0	
Short-Term Notes Payable	0	
Short-Term Bank Loan Payable	0	
Total Current Liabilities		\$60,110

ZEdenAreaROP

DATE:	October 6, 2016
TO:	ROP Governing Board
FROM:	Linda Granger, Superintendent
PREPARED BY:	Marites Fermin, Business Manager
SUBJECT:	Request the Governing Board to approve the Lease
	Agreement with L&M Investments for the Electrical Trainee
	Program–Turlock

BACKGROUND

The Eden Area ROP's Governing Board approved the assumption of the Construction Craft Training Center (CCTC) operations on September 1, 2016. As stated in the assumption agreement converting the existing lease agreements with CCTC for instructional space needs to transition to a lease agreement with the Eden Area ROP.

CURRENT SITUATION

In order to continue providing services, the Eden Area ROP recommends that we continue to lease the Turlock facility to provide electrical training to adults. The attached contract is the lease agreement between L& M Investments and the Eden Area ROP for the property located at:

2430 Acme Court, Turlock, CA 95380

The lease agreement will commence from October 1, 2016 to September 30, 2018.

CONSENT CALENDAR



COMMERCIAL LEASE AGREEMENT

(C.A.R. Form CL, Revised 12/15)

Da	te (For reference only): September 26, 2016
	L & M Investments ("Landlord") and Eden Area ROP ("Tenant") agree as follows:
1.	PROPERTY: Landlord rents to Tenant and Tenant rents from Landlord, the real property and improvements described as: 2480 Acme Court, Turlock, CA 95380 ("Premises"), which
	comprise approximately <u>50.000</u> % of the total square footage of rentable space in the entire property. See exhibit for a further description of the Premises.
2.	TERM: The term begins on (date) ("Commencement Date"),
	 (Check A or B): A. Lease: and shall terminate on (date) <u>September 30, 2018</u> at <u>11:59</u> AM X PM. Any holding over after the term of this agreement expires, with Landlord's consent, shall create a month-to-month tenancy that either party may terminate as specified in paragraph 2B. Rent shall be at a rate equal to the rent for the immediately preceding month, payable in advance. All other terms and conditions of this agreement shall remain in full force and effect. B. Month-to-month: and continues as a month-to-month tenancy. Either party may terminate the tenancy by giving written notice to the other at least 30 days prior to the intended termination date, subject to any applicable laws. Such notice may be given on any date.
	C. RENEWAL OR EXTENSION TERMS: See attached addendum
3.	 BASE RENT: A. Tenant agrees to pay Base Rent at the rate of (CHECK ONE ONLY:) (1) \$ per month, for the term of the agreement. (2) \$ per month, for the first 12 months of the agreement. Commencing with the 13th month, and upon expiration of each 12 months thereafter, rent shall be adjusted according to any increase in the U.S. Consumer Price Index of the Bureau of Labor Statistics of the Department of Labor for All Urban Consumers ("CPI") for
	(the city nearest the location of the Premises), based on the following formula: Base Rent will be multiplied by the most current CPI preceding the first calendar month during which the adjustment is to take effect, and divided by the most recent CPI preceding the Commencement Date. In no event shall any adjusted Base Rent be less than the Base Rent for the month immediately preceding the adjustment. If the CPI is no longer published, then the adjustment to Base Rent shall be based on an alternate index that most closely reflects the CPI.
	X (3) <u>1,475.00</u> per month for the period commencing October 1, 2016 and ending September 30, 2017 and \$ 1,550.00 per month for the period commencing October 1, 2017 and ending September 30, 2018 and \$ per month for the period commencing October 1, 2017 and ending September 30, 2018 and
	 (4) In accordance with the attached rent schedule. (5) Other: B. Base Rent is payable in advance on the 1st (or) (adv of each calendar month, and is delinquent on the next day. C. If the Commencement Date falls on any day other than the first day of the month, Base Rent for the first calendar month shall be prorated based on a 30-day period. If Tenant has paid one full month's Base Rent in advance of Commencement Date, Base Rent for the second calendar month
	shall be prorated based on a 30-day period.
4.	RENT:
	A. Definition: ("Rent") shall mean all monetary obligations of Tenant to Landlord under the terms of this agreement, except security deposit. B. Payment: Rent shall be paid to (Name)
	location specified by Landlord in writing to Tenant. C. Timing: Base Rent shall be paid as specified in paragraph 3. All other Rent shall be paid within 30 days after Tenant is billed by Landlord.
5.	EARLY POSSESSION: Tenant is entitled to possession of the Premises on in possession
	If Tenant is in possession prior to the Commencement Date, during this time (i) Tenant is not obligated to pay Base Rent, and (ii) Tenant [] is not obligated to pay Rent other than Base Rent. Whether or not Tenant is obligated to pay Rent prior to Commencement Date, Tenant is obligated to comply with all other terms of this agreement.
6.	SECURITY DEPOSIT:
	A. Tenant agrees to pay Landlord \$as a security deposit. Tenant agrees not to hold Broker responsible for its return (IF CHECKED:) [] If Base Rent increases during the term of this agreement, Tenant agrees to increase security deposit by the same proportion
	 as the increase in Base Rent. B. All or any portion of the security deposit may be used, as reasonably necessary, to: (i) cure Tenant's default in payment of Rent, late charges non-sufficient funds ("NSF") fees, or other sums due; (ii) repair damage, excluding ordinary wear and tear, caused by Tenant or by a guest or licensee of Tenant; (iii) broom clean the Premises, if necessary, upon termination of tenancy; and (iv) cover any other unfulfilled obligation or Tenant. SECURITY DEPOSIT SHALL NOT BE USED BY TENANT IN LIEU OF PAYMENT OF LAST MONTH'S RENT. If all or any portion of the
	security deposit is used during tenancy, Tenant agrees to reinstate the total security deposit within 5 days after written notice is delivered to Tenant. Within 30 days after Landlord receives possession of the Premises, Landlord shall: (i) furnish Tenant an itemized statement indicating the amount of any security deposit received and the basis for its disposition, and (ii) return any remaining portion of security deposit to Tenant However, if the Landlord's only claim upon the security deposit is for unpaid Rent, then the remaining portion of the security deposit, afte deduction of unpaid Rent, shall be returned within 14 days after the Landlord receives possession.
	C. No interest will be paid on security deposit, unless required by local ordinance.
La	andlord's Initials () () Tenant's Initials () ()

© 2015, California Association of REALTORS®, Inc. CL REVISED 12/15 (PAGE 1 of 6)



Date September 26, 2016

	×
7. PAYMENTS	52

		TOTAL DUE	PAYMENT RECEIVED	BALANCE DUE	DUE DATE
A.	Rent: From <u>10/01/2016</u> To <u>10/31/2016</u> Date Date	\$1,475.00	\$	\$1,475.00	10/01/2016
Β.	Security Deposit	\$	\$	\$	
C.	Other:	\$	\$	\$	
D.	Category Other:	\$	\$	\$	
E.	Category Total:	\$1,475.00	\$	\$1,475.00	

8. PARKING: Tenant is entitled to <u>common area</u> unreserved and <u>reserved vehicle parking spaces</u>. The right to parking <u>x</u> is <u>is</u> is not included in the Base Rent charged pursuant to paragraph 3. If not included in the Base Rent, the parking rental fee shall be an additional <u>per month</u>. Parking space(s) are to be used for parking operable motor vehicles, except for trailers, boats, campers, buses or trucks (other than pick-up trucks). Tenant shall park in assigned space(s) only. Parking space(s) are to be kept clean. Vehicles leaking oil, gas or other motor vehicle fluids shall not be parked in parking spaces or on the Premises. Mechanical work or storage of inoperable vehicles is not allowed in parking space(s) or elsewhere on the Premises. No overnight parking is permitted.

- 9. ADDITIONAL STORAGE: Storage is permitted as follows: <u>within warehouse and fenced yard</u> The right to additional storage space is is is not included in the Base Rent charged pursuant to paragraph 3. If not included in Base Rent, storage space shall be an additional \$ _______ per month. Tenant shall store only personal property that Tenant owns, and shall not store property that is claimed by another, or in which another has any right, title, or interest. Tenant shall not store any improperly packaged food or perishable goods, flammable materials, explosives, or other dangerous or hazardous material. Tenant shall pay for, and be responsible for, the clean-up of any contamination caused by Tenant's use of the storage area.
- 10. LATE CHARGE; INTEREST; NSF CHECKS: Tenant acknowledges that either late payment of Rent or issuance of a NSF check may cause Landlord to incur costs and expenses, the exact amount of which are extremely difficult and impractical to determine. These costs may include, but are not limited to, processing, enforcement and accounting expenses, and late charges imposed on Landlord. If any installment of Rent due from Tenant is not received by Landlord within 5 calendar days after date due, or if a check is returned NSF, Tenant shall pay to Landlord, respectively, \$ <u>150.00</u> as late charge, plus 10% interest per annum on the delinquent amount and \$25.00 as a NSF fee, any of which shall be deemed additional Rent. Landlord and Tenant agree that these charges represent a fair and reasonable estimate of the costs Landlord may incur by reason of Tenant's late or NSF fee shall not constitute a waiver as to any default of Tenant. Landlord's right to collect a Late Charge or NSF fee shall not be deemed an extension of the date Rent is due under paragraph 4, or prevent Landlord from exercising any other rights and remedies under this agreement, and as provided by law.
- 11. CONDITION OF PREMISES: Tenant has examined the Premises and acknowledges that Premise is clean and in operative condition, with the following exceptions: <u>none</u>.

Items listed as exceptions shall be dealt with in the following manner:

- 12. ZONING AND LAND USE: Tenant accepts the Premises subject to all local, state and federal laws, regulations and ordinances ("Laws"). Landlord makes no representation or warranty that Premises are now or in the future will be suitable for Tenant's use. Tenant has made its own investigation regarding all applicable Laws.
- 13. TENANT OPERATING EXPENSES: Tenant agrees to pay for all utilities and services directly billed to Tenant <u>and reimburse Landlord for water</u>, <u>sewer</u>, garbage and security, if not billed directly.

14. PROPERTY OPERATING EXPENSES:

A. Tenant agrees to pay its proportionate share of Landlord's estimated monthly property operating expenses, including but not limited to, common area maintenance, consolidated utility and service bills, insurance, and real property taxes, based on the ratio of the square footage of the Premises to the total square footage of the rentable space in the entire property.

OR B. X (If checked) Paragraph 14 does not apply.

15. USE: The Premises are for the sole use as warehouse

No other use is permitted without Landlord's prior written consent. If any use by Tenant causes an increase in the premium on Landlord's existing property insurance, Tenant shall pay for the increased cost. Tenant will comply with all Laws affecting its use of the Premises.

16. RULES/REGULATIONS: Tenant agrees to comply with all rules and regulations of Landlord (and, if applicable, Owner's Association) that are at any time posted on the Premises or delivered to Tenant. Tenant shall not, and shall ensure that guests and licensees of Tenant do not, disturb, annoy, endanger, or interfere with other tenants of the building or neighbors, or use the Premises for any unlawful purposes, including, but not limited to, using, manufacturing, selling, storing, or transporting illicit drugs or other contraband, or violate any law or ordinance, or committing a waste or nuisance on or about the Premises.

17. MAINTENANCE:

- A. Tenant OR [(If checked, Landlord) shall professionally maintain the Premises including heating, air conditioning, electrical, plumbing and water systems, if any, and keep glass, windows and doors in operable and safe condition. Unless Landlord is checked, if Tenant fails to maintain the Premises, Landlord may contract for or perform such maintenance, and charge Tenant for Landlord's cost.
- B. Landlord OR [(If checked, Tenant) shall maintain the roof, foundation, exterior walls, common areas and

<u>pl_)(___)</u> Landlord's Initials (

Tenant's Initials ()()



CL REVISED 12/15 (PAGE 2 of 6)

Date September 26, 2016

- 18. ALTERATIONS: Tenant shall not make any alterations in or about the Premises, including installation of trade fixtures and signs, without Landlord's prior written consent, which shall not be unreasonably withheld. Any alterations to the Premises shall be done according to Law and with required permits. Tenant shall give Landlord advance notice of the commencement date of any planned alteration, so that Landlord, at its option, may post a Notice of Non-Responsibility to prevent potential liens against Landlord's interest in the Premises. Landlord may also require Tenant to provide Landlord with lien releases from any contractor performing work on the Premises.
- 19. GOVERNMENT IMPOSED ALTERATIONS: Any alterations required by Law as a result of Tenant's use shall be Tenant's responsibility. Landlord shall be responsible for any other alterations required by Law.
- 20. ENTRY: Tenant shall make Premises available to Landlord or Landlord's agent for the purpose of entering to make inspections, necessary or agreed repairs, alterations, or improvements, or to supply necessary or agreed services, or to show Premises to prospective or actual purchasers, tenants, mortgagees, lenders, appraisers, or contractors. Landlord and Tenant agree that 24 hours notice (oral or written) shall be reasonable and sufficient notice. In an emergency, Landlord or Landlord's representative may enter Premises at any time without prior notice.
- 21. SIGNS: Tenant authorizes Landlord to place a FOR SALE sign on the Premises at any time, and a FOR LEASE sign on the Premises within the 90 (or ______) day period preceding the termination of the agreement.
- 22. SUBLETTING/ASSIGNMENT: Tenant shall not sublet or encumber all or any part of Premises, or assign or transfer this agreement or any interest in it, without the prior written consent of Landlord, which shall not be unreasonably withheld. Unless such consent is obtained, any subletting, assignment, transfer, or encumbrance of the Premises, agreement, or tenancy, by voluntary act of Tenant, operation of law, or otherwise, shall be null and void, and, at the option of Landlord, terminate this agreement. Any proposed sublessee, assignee, or transferee shall submit to Landlord an application and credit information for Landlord's approval, and, if approved, sign a separate written agreement with Landlord and Tenant. Landlord's consent to any one sublease, assignment, or transfer, shall not be construed as consent to any subsequent sublease, assignment, or transfer, and does not release Tenant of Tenant's obligation under this agreement.
- 23. POSSESSION: If Landlord is unable to deliver possession of Premises on Commencement Date, such date shall be extended to the date on which possession is made available to Tenant. However, the expiration date shall remain the same as specified in paragraph 2. If Landlord is unable to deliver possession within 60 (or ______) calendar days after the agreed Commencement Date, Tenant may terminate this agreement by giving written notice to Landlord, and shall be refunded all Rent and security deposit paid.
- 24. TENANT'S OBLIGATIONS UPON VACATING PREMISES: Upon termination of agreement, Tenant shall: (i) give Landlord all copies of all keys or opening devices to Premises, including any common areas; (ii) vacate Premises and surrender it to Landlord empty of all persons and personal property; (iii) vacate all parking and storage spaces; (iv) deliver Premises to Landlord in the same condition as referenced in paragraph 11; (v) clean Premises; (vi) give written notice to Landlord of Tenant's forwarding address; and (vii)

All improvements installed by Tenant, with or without Landlord's consent, become the property of Landlord upon termination. Landlord may nevertheless require Tenant to remove any such improvement that did not exist at the time possession was made available to Tenant.

- 25. BREACH OF CONTRACT/EARLY TERMINATION: In event Tenant, prior to expiration of this agreement, breaches any obligation in this agreement, abandons the premises, or gives notice of tenant's intent to terminate this tenancy prior to its expiration, in addition to any obligations established by paragraph 24, Tenant shall also be responsible for lost rent, rental commissions, advertising expenses, and painting costs necessary to ready Premises for re-rental. Landlord may also recover from Tenant: (i) the worth, at the time of award, of the unpaid Rent that had been earned at the time of termination; (ii) the worth, at the time of award, of the amount by which the unpaid Rent that would have been earned after expiration until the time of award exceeds the amount of such rental loss the Tenant proves could have been reasonably avoided; and (iii) the worth, at the time of award, of the amount by which the unpaid Rent the time of such rental loss that Tenant proves could be reasonably avoided. Landlord may elect to continue the tenancy in effect for so long as Landlord does not terminate Tenant's right to possession, by either written notice of termination of possession or by releting the Premises to another who takes possession, and Landlord may enforce all Landlord's rights and remedies under this agreement, including the right to recover the Rent as it becomes due.
- 26. DAMAGE TO PREMISES: If, by no fault of Tenant, Premises are totally or partially damaged or destroyed by fire, earthquake, accident or other casualty, Landlord shall have the right to restore the Premises by repair or rebuilding. If Landlord elects to repair or rebuild, and is able to complete such restoration within 90 days from the date of damage, subject to the terms of this paragraph, this agreement shall remain in full force and effect. If Landlord is unable to restore the Premises within this time, or if Landlord elects not to restore, then either Landlord or Tenant may terminate this agreement by giving the other written notice. Rent shall be abated as of the date of damage. The abated amount shall be the current monthly Base Rent prorated on a 30-day basis. If this agreement is not terminated, and the damage is not repaired, then Rent shall be reduced based on the extent to which the damage interferes with Tenant's reasonable use of the Premises. If total or partial destruction or damage occurs as a result of an act of Tenant or Tenant's guests, (i) only Landlord shall have the right, at Landlord's sole discretion, within 30 days after such total or partial destruction or damage to treat the lease as terminated by Tenant, and (ii) Landlord shall have the right to recover damages from Tenant.
- 27. HAZARDOUS MATERIALS: Tenant shall not use, store, generate, release or dispose of any hazardous material on the Premises or the property of which the Premises are part. However, Tenant is permitted to make use of such materials that are required to be used in the normal course of Tenant's business provided that Tenant complies with all applicable Laws related to the hazardous materials. Tenant is responsible for the cost of removal and remediation, or any clean-up of any contamination caused by Tenant.
- 28. CONDEMNATION: If all or part of the Premises is condemned for public use, either party may terminate this agreement as of the date possession is given to the condemner. All condemnation proceeds, exclusive of those allocated by the condemner to Tenant's relocation costs and trade fixtures, belong to Landlord.
- 29. INSURANCE: Tenant's personal property, fixtures, equipment, inventory and vehicles are not insured by Landlord against loss or damage due to fire, theft, vandalism, rain, water, criminal or negligent acts of others, or any other cause. Tenant is to carry Tenant's own property insurance to protect Tenant from any such loss. In addition, Tenant shall carry (i) liability insurance in an amount of not less than \$ 1,000.00 and (ii) property insurance in an amount sufficient to cover the replacement cost of the property if Tenant is responsible for maintenance under paragraph 17B. Tenant's insurance shall name Landlord and Landlord's agent as additional insured. Tenant, upon Landlord's request, shall provide Landlord with a certificate of insurance establishing Tenant's compliance. Landlord shall maintain liability insurance insuring Landlord, but not Tenant, in an amount of at least \$ 1,000.00 , plus property insurance in an amount sufficient to cover the property insurance in an amount at least sufficient to cover Tenant's complete rental obligation to Landlord. Landlord is advised to obtain a policy of rental loss insurance. Both Landlord and Tenant release each other, and waive their respective rights to subrogation against each other, for loss or damage covered by insurance.

Landlord's Initials (() (

CL REVISED 12/15 (PAGE 3 of 6)

Tenant's Initials (_____) (_____)



COMMERCIAL LEASE AGREEMENT (CL PAGE 3 OF 6)

)

Date September 26, 2016

- 30. TENANCY STATEMENT (ESTOPPEL CERTIFICATE): Tenant shall execute and return a tenancy statement (estoppel certificate), delivered to Tenant by Landlord or Landlord's agent, within 3 days after its receipt. The tenancy statement shall acknowledge that this agreement is unmodified and in full force, or in full force as modified, and state the modifications. Failure to comply with this requirement: (i) shall be deemed Tenant's acknowledgment that the tenancy statement is true and correct, and may be relied upon by a prospective lender or purchaser, and (ii) may be treated by Landlord as a material breach of this agreement. Tenant shall also prepare, execute, and deliver to Landlord any financial statement (which will be held in confidence) reasonably requested by a prospective lender or buyer.
- 31, LANDLORD'S TRANSFER: Tenant agrees that the transferee of Landlord's interest shall be substituted as Landlord under this agreement, Landlord will be released of any further obligation to Tenant regarding the security deposit, only if the security deposit is returned to Tenant upon such transfer, or if the security deposit is actually transferred to the transferee. For all other obligations under this agreement, Landlord is released of any further liability to Tenant, upon Landlord's transfer.
- 32. SUBORDINATION: This agreement shall be subordinate to all existing liens and, at Landlord's option, the lien of any first deed of trust or first mortgage subsequently placed upon the real property of which the Premises are a part, and to any advances made on the security of the Premises, and to all renewals, modifications, consolidations, replacements, and extensions. However, as to the lien of any deed of trust or mortgage entered into after execution of this agreement, Tenant's right to quiet possession of the Premises shall not be disturbed if Tenant is not in default and so long as Tenant pays the Rent and observes and performs all of the provisions of this agreement, unless this agreement is otherwise terminated pursuant to its terms. If any mortgagee, trustee, or ground lessor elects to have this agreement placed in a security position prior to the lien of a mortgage, deed of trust, or ground lease, and gives written notice to Tenant, this agreement shall be deemed prior to that mortgage, deed of trust, or ground lease, or the date of recording.
- 33. TENANT REPRESENTATIONS; CREDIT: Tenant warrants that all statements in Tenant's financial documents and rental application are accurate. Tenant authorizes Landlord and Broker(s) to obtain Tenant's credit report at time of application and periodically during tenancy in connection with approval, modification, or enforcement of this agreement. Landlord may cancel this agreement: (i) before occupancy begins, upon disapproval of the credit report(s); or (ii) at any time, upon discovering that information in Tenant's application is false. A negative credit report reflecting on Tenant's record may be submitted to a credit reporting agency, if Tenant fails to pay Rent or comply with any other obligation under this agreement.
- 34. CONSTRUCTION-RELATED ACCESSIBILITY STANDARDS: Landlord states that the Premises has, or X has not been inspected by a Certified Access Specialist. If so, Landlord states that the Premises has, or has not been determined to meet all applicable construction-related accessibility standards pursuant to Civil Code Section 55.53.

35. DISPUTE RESOLUTION:

- A. MEDIATION: Tenant and Landlord agree to mediate any dispute or claim arising between them out of this agreement, or any resulting transaction, before resorting to arbitration or court action, subject to paragraph 35B(2) below. Paragraphs 35B(2) and (3) apply whether or not the arbitration provision is initialed. Mediation fees, if any, shall be divided equally among the parties involved. If for any dispute or claim to which this paragraph applies, any party commences an action without first attempting to resolve the matter through mediation, or refuses to mediate after a request has been made, then that party shall not be entitled to recover attorney fees, even if they would otherwise be available to that party in any such action. THIS MEDIATION PROVISION APPLIES WHETHER OR NOT THE ARBITRATION PROVISION IS INITIALED.
- B. ARBITRATION OF DISPUTES: (1) Tenant and Landlord agree that any dispute or claim in Law or equity arising between them out of this agreement or any resulting transaction, which is not settled through mediation, shall be decided by neutral, binding arbitration, including and subject to paragraphs 35B(2) and (3) below. The arbitrator shall be a retired judge or justice, or an attorney with at least 5 years of real estate transactional law experience, unless the parties mutually agree to a different arbitrator, who shall render an award in accordance with substantive California Law. In all other respects, the arbitration shall be conducted in accordance with Part III, Title 9 of the California Code of Civil Procedure. Judgment upon the award of the arbitrator(s) may be entered in any court having jurisdiction. The parties shall have the right to discovery in accordance with Code of Civil Procedure §1283.05.

(2) EXCLUSIONS FROM MEDIATION AND ARBITRATION: The following matters are excluded from Mediation and Arbitration hereunder: (i) a judicial or non-judicial foreclosure or other action or proceeding to enforce a deed of trust, mortgage, or installment land sale contract as defined in Civil Code §2985; (ii) an unlawful detainer action; (iii) the filing or enforcement of a mechanic's lien; (iv) any matter that is within the jurisdiction of a probate, small claims, or bankruptcy court; and (v) an action for bodily injury or wrongful death, or for latent or patent defects to which Code of Civil Procedure §337.1 or §337.15 applies. The filing of a court action to enable the recording of a notice of pending action, for order of attachment, receivership, injunction, or other provisional remedies, shall not constitute a violation of the mediation and arbitration provisions.

(3) BROKERS: Tenant and Landlord agree to mediate and arbitrate disputes or claims involving either or both Brokers, provided either or both Brokers shall have agreed to such mediation or arbitration, prior to, or within a reasonable time after the dispute or claim is presented to Brokers. Any election by either or both Brokers to participate in mediation or arbitration shall not result in Brokers being deemed parties to the agreement.

"NOTICE: BY INITIALING IN THE SPACE BELOW YOU ARE AGREEING TO HAVE ANY DISPUTE ARISING OUT OF THE MATTERS INCLUDED IN THE 'ARBITRATION OF DISPUTES' PROVISION DECIDED BY NEUTRAL ARBITRATION AS PROVIDED BY CALIFORNIA LAW AND YOU ARE GIVING UP ANY RIGHTS YOU MIGHT POSSESS TO HAVE THE DISPUTE LITIGATED IN A COURT OR JURY TRIAL. BY INITIALING IN THE SPACE BELOW YOU ARE GIVING UP YOUR JUDICIAL RIGHTS TO DISCOVERY AND APPEAL, UNLESS THOSE RIGHTS ARE SPECIFICALLY INCLUDED IN THE 'ARBITRATION OF DISPUTES' PROVISION. IF YOU REFUSE TO SUBMIT TO ARBITRATION AFTER AGREEING TO THIS PROVISION, YOU MAY BE COMPELLED TO ARBITRATE UNDER THE AUTHORITY OF THE CALIFORNIA CODE OF CIVIL PROCEDURE. YOUR AGREEMENT TO THIS ARBITRATION PROVISION IS VOLUNTARY."

"WE HAVE READ AND UNDERSTAND THE FOREGOING AND AGREE TO SUBMIT DISPUTES ARISING OUT OF THE MATTERS INCLUDED IN THE 'ARBITRATION OF DISPUTES' PROVISION TO NEUTRAL **ARBITRATION."**

	Landlord's Initials/	Tenant's Initials	/	
Landlord's Initials ()	Tenant's Initials ()()		
CL REVISED 12/15 (PAGE 4 of 6)				LEJ
COMMERC	IAL LEASE AGREEMENT (CL PAGE	E 4 OF 6)		EQUAL HOUSING OPPORTUNITY
	terrene	* *	in the s	~ 7

Date September 26, 2016

36. JOINT AND INDIVIDUAL OBLIGATIONS: If there is more than one Tenant, each one shall be individually and completely responsible for the performance of all obligations of Tenant under this agreement, jointly with every other Tenant, and individually, whether or not in possession.

37. NOTICE: Notices may be served by mail, facsimile, or courier at the following address or location, or at any other location subsequently designated:

Landlord: L & M Investments	
P.O. Box 766	
Turlock, CA 95381	
209-667-2851	

Tenant: Eden Area ROP Linda Granger, Superintendent

Notice is deemed effective upon the earliest of the following: (i) personal receipt by either party or their agent; (ii) written acknowledgement of notice; or (iii) 5 days after mailing notice to such location by first class mail, postage pre-paid.

38. WAIVER: The waiver of any breach shall not be construed as a continuing waiver of the same breach or a waiver of any subsequent breach.

39. INDEMNIFICATION: Tenant shall indemnify, defend and hold Landlord harmless from all claims, disputes, litigation, judgments and attorney fees arising out of Tenant's use of the Premises.

40. OTHER TERMS AND CONDITIONS/SUPPLEMENTS:

The following ATTACHED supplements/exhibits are incorporated in this agreement: Option Agreement (C.A.R. Form OA)

- 41. ATTORNEY FEES: In any action or proceeding arising out of this agreement, the prevailing party between Landlord and Tenant shall be entitled to reasonable attorney fees and costs from the non-prevailing Landlord or Tenant, except as provided in paragraph 35A.
- 42. ENTIRE CONTRACT: Time is of the essence. All prior agreements between Landlord and Tenant are incorporated in this agreement, which constitutes the entire contract. It is intended as a final expression of the parties' agreement, and may not be contradicted by evidence of any prior agreement or contemporaneous oral agreement. The parties further intend that this agreement constitutes the complete and exclusive statement of its terms, and that no extrinsic evidence whatsoever may be introduced in any judicial or other proceeding, if any, involving this agreement. Any provision of this agreement that is held to be invalid shall not affect the validity or enforceability of any other provision in this agreement. This agreement shall be binding upon, and inure to the benefit of, the heirs, assignees and successors to the parties.
- 43. BROKERAGE: Landlord and Tenant shall each pay to Broker(s) the fee agreed to, if any, in a separate written agreement. Neither Tenant nor Landlord has utilized the services of, or for any other reason owes compensation to, a licensed real estate broker (individual or corporate), agent, finder, or other entity, other than as named in this agreement, in connection with any act relating to the Premises, including, but not limited to, inquiries, introductions, consultations, and negotiations leading to this agreement. Tenant and Landlord each agree to indemnify, defend and hold harmless the other, and the Brokers specified herein, and their agents, from and against any costs, expenses, or liability for compensation claimed inconsistent with the warranty and representation in this paragraph 43.

44. AGENCY CONFIRMATION: The following agency relationships are hereby confirmed for this transaction: Listing Agent:

(Print Firm Name) is the agent of (check one):

the Landlord exclusively; or both the Tenant and Landlord.

Selling Agent: (Print Firm Name) (if not same as Listing Agent) is the agent of (check one): the Tenant exclusively; or the Landlord exclusively; or both the Tenant and Landlord. Real Estate Brokers are not parties to the agreement between Tenant and Landlord.

bly) Landlord's Initials (

CL REVISED 12/15 (PAGE 5 of 6)

Tenant's Initials ()()



COMMERCIAL LEASE AGREEMENT (CL PAGE 5 OF 6)

Braducad with TinEarm® by Tinl ariv 10070 Eiftaan Mila Dood Eracar Michigan 40026 yaway Tinl ariv aam

Landlord and Tenant acknowledge and agree that Brokers: (i) do not guarantee the condition of the Premises; (ii) cannot verify representations made by others; (iii) will not verify zoning and land use restrictions; (iv) cannot provide legal or tax advice; (v) will not provide other advice or information that exceeds the knowledge, education or experience required to obtain a real estate license. Furthermore, if Brokers are not also acting as Landlord in this agreement, Brokers: (vi) do not decide what rental rate a Tenant should pay or Landlord should accept; and (vii) do not decide upon the length or other terms of tenancy. Landlord and Tenant agree that they will seek legal, tax, insurance, and other desired assistance from appropriate professionals.

Tenant		Date	
Eden Area ROP (Print name)	······		
Address	City	State Zip	
Tenant		Date	
(Print name)			
Address	City	State Zip	

GUARANTEE: In consideration of the execution of this Agreement by and between Landlord and Tenant and for valuable consideration, receipt of which is hereby acknowledged, the undersigned ("Guarantor") does hereby: (i) guarantee unconditionally to Landlord and Landlord's agents, successors and assigns, the prompt payment of Rent or other sums that become due pursuant to this Agreement, including any and all court costs and attorney fees included in enforcing the Agreement; (ii) consent to any changes, modifications or alterations of any term in this Agreement agreed to by Landlord and Tenant; and (iii) waive any right to require Landlord and/or Landlord's agents to proceed against Tenant for any default occurring under this Agreement before seeking to enforce this Guarantee.

Guarantor (Print Na	me)			
Guarantor			Date	
		City	State	Zip
Telephone	Fax	E-mail		
Landlord agrees to rent th	e Premises on the above terms	and conditions.		
Landlord (owner or agent	with authority to enter into this age	reement) L & M Investments	Date	-22-16
Address P.O. Box 766		CityCity	State CA	Zip <u>95381</u>
Landlord			Date	
	t with authority to enter into this ag			
Address		City	State	Zip
Agency relationships are con Landlord and Tenant.	nfirmed as above. Real estate brok	ers who are not also Landlord in this agree	ement are not a party to the	ne agreement betwee
Real Estate Broker (Leasing	J Firm)		CalBRE Lic. #	
			Date	
Address		City	State	Zip
Telephone	Fax	E-mail		

Real Estate Broker (Listin	g Firm)		CalBRE Lic. #	
By (Agent)		CalBRE Lic. #	Date	
Address		City	State Zip	
Telephone	Fax	E-mail		

© 2015, California Association of REALTORS®, Inc. United States copyright law (Title 17 U.S. Code) forbids the unauthorized distribution, display and reproduction of this form, or any portion thereof, by photocopy machine or any other means, including facsimile or computerized formats. THIS FORM HAS BEEN APPROVED BY THE CALIFORNIA ASSOCIATION OF REALTORS® (C.A.R.). NO REPRESENTATION IS MADE AS TO THE LEGAL VALIDITY OR ACCUMATION OF REALTORS® (C.A.R.). NO REPRESENTATION IS MADE AS TO THE LEGAL VALIDITY OR TARGET AND ADDITION OF REALTORS® (C.A.R.). NO REPRESENTATION IS MADE AS TO THE LEGAL VALIDITY OR ACCUMATION OF REALTORS® (C.A.R.). NO REPRESENTATION IS MADE AS TO THE LEGAL VALIDITY OR ACCUMATION OF REALTORS® (C.A.R.). NO REPRESENTATION IS MADE AS TO THE LEGAL VALIDITY OR ACCUMATION OF REALTORS® (C.A.R.). NO REPRESENTATION IS MADE AS TO THE LEGAL VALIDITY OR ACCUMATION OF REALTORS® (C.A.R.). NO REPRESENTATION IS MADE AS TO THE LEGAL VALIDITY OR ACCUMATION OF REALTORS® (C.A.R.). NO REPRESENTATION IS MADE AS TO THE LEGAL VALIDITY OR ACCUMATION OF REALTORS® (C.A.R.). NO REPRESENTATION IS MADE AS TO THE LEGAL VALIDITY OR ACCUMATION OF REALTORS® (C.A.R.). NO REPRESENTATION IS MADE AS TO THE LEGAL VALIDITY OR ACCUMATION OF REALTORS® (C.A.R.). NO REPRESENTATION IS MADE AS TO THE LEGAL VALIDITY OR ACCUMATION OF REALTORS® (C.A.R.). NO REPRESENTATION IS MADE AS TO THE LEGAL VALIDITY OR ACCUMATION OF REALTORS® (C.A.R.). NO REPRESENTATION IS MADE AS TO THE LEGAL VALIDITY OR ACCUMATION OF REALTORS® (C.A.R.). NO REPRESENTATION IS MADE AS TO THE LEGAL VALIDITY OR ACCUMATION OF REALTORS® (C.A.R.). NO REPRESENTATION IS MADE AS TO THE LEGAL VALIDITY OR ACUMATION OF REALTORS® (C.A.R.). NO REPRESENTATION IS MADE AS TO THE ACUMATION OF REALTORS® (C.A.R.). NO REPRESENTATION IS MADE AS TO THE LEGAL VALIDITY OR ACUMATION OF REALTORS® (C.A.R.). NO REPRESENTATION OF REALTORS® (C.A.R.). NO REPRESENTATION IS MADE AS TO THE ACUMATION OF REALTORS® (C.A.R.). NO REPRESENTATION OF REALTORS® (C.A.R.). NO REPRESENTATION OF REALTORS® (C.A.R.). NO

ACCURACY OF ANY PROVISION IN ANY SPECIFIC TRANSACTION. A REAL ESTATE BROKER IS THE PERSON QUALIFIED TO ADVISE ON REAL ESTATE TRANSACTIONS. IF YOU DESIRE LEGAL OR TAX ADVICE, CONSULT AN APPROPRIATE PROFESSIONAL.

Published and Distributed by:



■ a subsidiary of the California Association of REALTORS®

525 South Virgil Avenue, Los Angeles, California 90020

CL REVISED 12/15 (PAGE 6 of 6)

Reviewed by _____ Date ____



COMMERCIAL LEASE AGREEMENT (CL PAGE 6 OF 6)

ZEdenAreaROP

DATE:	October 6, 2016
TO:	ROP Governing Board
FROM:	Linda Granger, Superintendent
PREPARED BY:	Marites Fermin, Business Manager
SUBJECT:	Request the Governing Board to approve the Sublease
	Agreement with American Home Inspectors Training (AHIT)

BACKGROUND

The Eden Area ROP's Governing Board approved the Eden Area ROP's assumption of Construction Craft Training Center (CCTC) operations on September 1, 2016. CCTC annually subleases the campus located in 26200 Industrial Blvd., Hayward, CA.

CURRENT SITUATION

Previously, CCTC subleased the property on 26200 Industrial Boulevard, Hayward, CA 94545 to American Home Inspectors Training (AHIT) three times a year. They are interested in continuing this practice. The sublease agreement enables them to provide training on the following dates at the Hayward facility this year:

- February 10-12, 2017
- July 7-9, 2017
- October 27-29, 2017

Attached is the sublease contract with AHIT.

Fiscal Impact:

\$2,250 annual revenue received by the ROP.

CONSENT CALENDAR



REFERRAL AND CLASSROOM AGREEMENT

This strategic referral and classroom agreement between OnCourse Real Estate, Inc dba American Home Inspectors Training ("AHIT") and Eden Area ROP ("EAROP") outlines the responsibilities of each party and the payments by AHIT to EAROP.

In 2017, AHIT will teach classes at the location designated below on:

2.10-2.12.17; 7.7-7.9.17; 10.27-10.29.17.

I. AHIT Rights and Obligations

- A. AHIT will provide the following:
 - 1. instructors to teach the classroom course;
 - 2. course materials;
 - 3. confirmation letters to students;
 - 4. marketing and advertising through national and local channels;
 - 5. indemnity for AHIT students and instructors during class;
 - 6. payment to homeowners for use of local homes during practice inspections (\$150 per home per day);
 - 7. signs for display on classroom door(s) and walls as well as directional signs;
 - 8. student and classroom fees paid to EAROP according to the fee schedule outlined below;
 - 9. contact information for each student referred by EAROP to AHIT who AHIT enrolls in the home inspection course, including name, address, phone number, and email address.
- B. AHIT shall retain all right, title, and interest in and to all of AHIT's brands, logos, trademarks, course materials, and any and all copy or material that represents the creative effort of AHIT and/or the utilization of creativity, illustrations, labor, composition or material furnished by it. EAROP understands and agrees that it cannot authorize photographic or other reproduction, in whole or in part, of any such copy or material for use in any other medium without AHIT's prior written consent.
- C. In the event marketing efforts do not yield the minimum number (6) of students, AHIT shall have the right to cancel a scheduled class.

II. EAROP Rights and Obligations

- A. EAROP will:
 - provide an appropriate classroom at EAROP's campus located at 26200 Industrial Blvd. Hayward, CA 94545, that can accommodate up to 25 students, for the 3-day class, with Internet access, a whiteboard, and the ability to project video and PowerPoint presentations;

- 2. provide lockable storage for AHIT training materials in the building that houses the designated classroom;
- 3. permit access to the building and classroom from 7am to 7pm for the 3-day period for each scheduled class; the classroom will be opened each morning no later than 7am and locked each evening no earlier than 7pm;
- 4. permission for AHIT to post directional signage during class training sessions.
- B. EAROP shall retain all right, title, and interest in and to all of EAROP's brands, logos, and trademarks, course materials, and any and all copy or material that represents the creative effort of EAROP and/or the utilization of creativity, illustrations, labor, composition or material furnished by it. AHIT understands and agrees that it cannot authorize photographic or other reproduction, in whole or in part, of any such copy or material for use in any other medium without EAROP's prior written consent.

III. Fee Structure

- A. AHIT agrees to pay EAROP \$250 per day for use of the classroom facility for the 3 days of the classroom course. EAROP will invoice AHIT for this payment.
- IV. **Term.** This agreement shall become effective on the date it has been signed by both parties and shall remain in effect until December 31, 2017.

DATE

Katie Nitzsche Manager, School Administration AHIT

Linda Granger Superintendent Eden Area ROP DATE

Information Items



ZEdenAreaROP

DATE:	October 6, 2016
TO:	ROP Governing Board
FROM:	Linda Granger, Superintendent
PREPARED BY:	Sheila Lawrence, Assistant Director of Offsite Programs
SUBJECT:	CDE Course Review – Photography

BACKGROUND

Courses for regional occupational centers and programs (ROCPs) shall be developed with the cooperation of industry representatives from the occupation for which the courses are designed. The course outline will serve as the regional occupational center or program (ROCP) teacher's road map for providing learning experiences and opportunities for students to achieve career technical objectives effectively and efficiently. The outline helps the ROCP teacher ensure competency achievement.

CURRENT SITUATION

Attached is the Course Review for the following programs: Photography, the statistics provided are derived from the 2015-2016 C101 follow-up study completed June 2016.

RECOMMENDATION

Information only

EDEN AREA REGIONAL OCCUPATIONAL PROGRAM Criteria for Course Approval and Expansion Annual Review For Class Offerings 2015-2016 School Year

SCHOO	OL DIS	TRICT:	EDEN	AREA ROP		LOCATION:		CV, SLZ, AF	R, MTE	, HAY
PROG	RAM:	Photog	hotography			INSTRUCTO				Baughman ,Gurley,
		ourse		Enrollment as o				of 14/15	E	Enrollment as of 13/14
		ame		Year to Dat	te	Ye	ar to D	late		Year to Date
		ography		787			796			860
	Retent] YES	🗌 NO	nt: Class enrollme	/ear-to-dat		ed m			and staffing.
Text Bo	ok: N/A		•			Author: N/A		Edition	: N/A	
NO.	YES	NO								
1.	×			<u>MENT – Course r</u> LASS SCHEDUL Block		ent or future labor SECTIONS PER YEAR:		et news. MINS PER SECTION:		XPECTED MINIMUM DENTS PER SECTION:
				Varies		1		60 min.	310	25+
						TRUCTOR – Quali	fied		ruotor too	
2.	×		AVAILA		IFIED INS		meu/			ching course.
3.	×		work wit	h their counterpart	s in other p	orograms.				system improvements and
4.	×		aspects	of industry.		·				in and understanding of all
5.	×		of course	SCHOOL-TO-CAREER AND CAREER PATHWAY DEVELOPMENT – Course is designed as part of a sequence of courses, career pathways, etc.						
6.	X		 ADVISORY COMMITTEE – The course has been reviewed and recommended by a pre-established committee. Yes, instructor was present at advisory meeting and minutes are on file at ROP No, instructor was not present at advisory. Program was represented, at a joint industrial, by ROP personnel. Instructor MUST attend next advisory for program to meet compliance. Comments: 							
7.	x			LABOR MARKET NEEDS – Course meets current or future labor market needs.						
8.	×		WORK BASED LEARNING – Course incorporates work based learning opportunities (i.e. guest speakers, field trips, mock interviews, or student organizations)							
9.	×		COMMUNITY CLASSROOM AND COOPERATIVE VOCATIONAL EDUCATION – Course incorporates community classroom and cooperative vocational education (i.e., job training, internships, or job shadowing)							
10.	×		JOB PLACEMENT/FURTHER EDUCATION OPTIONS – Course has potential for student job placement in entry- level positions or course prepares students for further training opportunities within the designed career pathway.							
Image: Markowski bit Image: Second Seco										
OTHER	CONSID	ERATION	S:							
A-G Credit for UC - "G" " F"										
Community College Articulation – Chabot & Ohlone College					Strong Business or Industry Partnership					
Dual Enrollment Emerging Technologies -										
COMPLIANCE CATEGORIES										
R – <u>Program</u> meets al	<u>ı</u> : Progra		All criteria	Vatch Program: not met. See need to be with.	Criteria i	Probation: s not being met. i in danger of ion.		_ R – <u>Reduce</u> rogram∷ Downsiz ogram.	ing	S/T - Suspend/ Terminate program.

ZEdenAreaROP

DATE:	October 6, 2016
то:	ROP Governing Board
FROM:	Linda Granger, Superintendent
PREPARED BY:	Craig Lang, Director
SUBJECT:	Back to School Night

BACKGROUND INFORMATION

The Eden Area ROP has held an annual Back to School night for parents, family and friends to explore our programs at the Hayward Center Campus. Teachers have students demonstrate and explain what they are learning in their program.

CURRENT SITUATION

Back to School night was held on September 21, 2016.

RECOMMENDATION

Information only

ZEdenAreaROP

DATE:	October 6, 2016
TO:	Governing Board
FROM:	Linda Granger, Superintendent
PREPARED BY:	Sheila Lawrence, Assistant Director of Offsite Programs
SUBJECT:	Principals' Breakfast

BACKGROUND

It is the goal of the Eden Area Regional Occupational Program to continue to develop and implement career technical education course offerings and career pathways that encourage all students to enroll and stay enrolled in a rigorous course of study.

ROP administrators, faculty and staff work with students, parents, counselors, district and school site administrators to expand and improve practices and processes that will further build our partnership, as well as, build courses of study to meet career, academic and employability standards and competencies.

CURRENT SITUATION

On an ongoing basis, school site and ROP administrators meet to discuss goals, challenges, and strategies for student success. On September 29, 2016 high school principals from the 12 high schools were invited to meet, share and discuss college and career readiness.

RECOMMENDATION

Information only

Action Items



ZEdenAreaROP

DATE: October 6, 2016
TO: ROP Governing Board
FROM: Linda Granger, Superintendent
SUBJECT: Request the Governing Board to approve the MOU with Leadership Public School (LPS)

BACKGROUND

Occasionally charter schools come to the Eden Area ROP wanting to partner with us to provide services to their students. Leadership Public School (LPS) is interested in providing their students with career technical education opportunities that are not possible given the size of their school.

CURRENT SITUATION

Leadership Public School in Hayward is interested in piloting a program in which no more than 2 students from their charter school attend classes at the center. They would be responsible for transporting the students to and from the classes at the center and we would provide staffing for the programs in which the students enroll. This pilot program would be for one year and reevaluated for effectiveness prior to committing to any future years.

Fiscal Impact

No additional cost to the ROP, fees would cover the cost of services.

RECOMMENDATION

It is recommended that the Governing Board approve the MOU with Leadership Public School (LPS).



MEMORANDUM OF UNDERSTANDING

The Eden Area Regional Occupation Program (EAROP) and Leadership Public School (LPS), a public charter school in Alameda County, agree to the following terms for a pilot program:

I. TERMS OF MEMORANDUM OF UNDERSTANDING (MOU):

This agreement shall commence on July 1, 2016 and shall extend through June 30, 2017.

II. CONTRACT AMOUNT:

The Leadership Public School will pay the Eden Area Regional Occupational Program \$2,933 per student up to 2 students who enroll in the programs offered at the EAROP center during the 2016-2017 school year.

III. PURPOSE:

This pilot program will provide up to 2 students enrolled in Leadership Public School the opportunity to take Career Technical Education (CTE) classes at the Eden Area ROP center.

IV. ROLE AND RESPONSIBILLITES OF EDEN AREA ROP:

- Provide up to 13 classroom spaces in programs offered at the EAROP center for students in grades 11 and 12 of LPS.
- Invoice LPS for services two times during the school year, (1) October 1, 2016 and (2) May 1, 2017.
- Maintain accurate student attendance and grading information for each student.
- Provide LPS with grades and attendance in accordance with the EAROP calendar.
- Provide one main point of contact as a liaison with LPS.

V. ROLES AND RESPONSIBILITIES OF LEADERSHIP PUBLIC SCHOOL:

- Provide accurate and timely student demographic information, including information for students possessing an IEP or Sect. 504 Plan.
- Be responsible for transporting students to and from the EAROP center programs.
- Provide one main point of contact as a liaison with EAROP.
- Remit payment for services within 30 days of receipt of invoice billing.

VI. TERMS OF AGREEMENT:

An effort will be directed by both parties to maintain the terms of the agreement as defined. However, if urgent circumstances beyond either parties control occur, making one party unable to fulfill its agreement, this agreement can be renegotiated or terminated with 30 days notice.

The Eden Area ROP shall indemnify, defend and hold harmless the Leadership Public School and its employees from and against any and all loss, liability, expense, claims, costs, suites and damages of every kind, nature and description directly or indirectly arising from the performance of the work. This paragraph shall not be construed to exempt Leadership Public School and its employees from its own fraud, willful injury or violation of law whether willful or negligent.

The Leadership Public School shall indemnify, defend and hold harmless the Eden Area ROP and its employees from and against any and all loss, liability, expense, claims, costs, suites and damages of every kind, nature and description directly or indirectly arising from the performance of the work. This paragraph shall not be construed to exempt Eden Area ROP and its employees from its own fraud, willful injury or violation of law whether willful or negligent.

Under penalty of perjury I agree to the statements above and I am designated to sign this agreement on behalf of my agency.

Date:

NAME

NAME

CBO

Leadership Public School

TITLE Eden Area ROP